FINAL

PRELIMINARY ASSESSMENT REPORT FOR PERFLUORINATED COMPOUNDS

AT

PATRICK AIR FORCE BASE BREVARD COUNTY, FLORIDA

Prepared for:



Note: Throughout this report, there is a recurring erroneous statement indicating that wastewater from various sources at Patrick is sent "to the City of Cocoa's Dyal Water Treatment Plant". This statement is incorrect but was not caught prior to document finalization. All wastewater from Patrick AFB flows to the City of Cocoa Beach Wastewater Treatment Facility. (The Dyal Water Treatment Plant is actually the drinking water treatment plant on the mainland. No waste is directed there).

Air Force Civil Engineer Center 2261 Hughes Avenue, Suite 155 Lackland AFB, Texas 78236-9853

Contract No. FA8903-08-D-8772 Task Order 0065 CDRL A001A

September 2015

FINAL PRELIMINARY ASSESSMENT REPORT FOR PERFLUORINATED COMPOUNDS AT

PATRICK AIR FORCE BASE BREVARD COUNTY, FLORIDA

Prepared for:



Air Force Civil Engineer Center 2261 Hughes Avenue, Suite 155 Lackland AFB, Texas 78236-9853

Contract No. FA8903-08-D-8772 Task Order 0065 CDRL A001A

Prepared by:

HydroGeoLogic, Inc. 404 East Ramsey Road, Suite 210 San Antonio, Texas 78216

September 2015

			Form A	pproved	
REPORT DOCUME	OMB N	OMB No. 0704-0188			
Public reporting for this collection	response, ir	cluding the time for reviewing			
instruction, searching existing of collection of information. Send including suggestions for reduci and Reports, 1215 Jefferson Da Budget, Paperwork Reduction Ph	data sources, gathering and comments regarding this bu ng this burden, to Washing wis Highway, Suite 1024, A roject (0704B0188), Washin	maintaining the data ne rden estimate or any othe ton Headquarters Service Arlington, VA 22202-130 pton, DC 20503.	eded, and er aspect of s, Director 2, and to 1	completing and reviewing the this collection of information, ate for information Operations he Office of Management and	
1. AGENCY USE ONLY (Leav	e blank) 2. REPORT DA	TE	3. REPC	RT TYPE AND DATES	
			COVI	ERED	
	September 2015			Final	
4. TITLE AND SUBTITLE			4. FUNI	DING NUMBERS	
Preliminary Assessment Report f Patrick Air Force Base, Brevard	for Perfluorinated Compoun County, Florida	ls at	Contract Task Oro	No. FA8903-08-D-8772 ler No. 0065	
6. AUTHOR(S)					
HydroGeoLogic, Inc.					
7. PERFORMANCE ORGANIZATION NAMES(S) AND ADDRESS(S) HydroGeoLogic, Inc.				8. PERFORMANCE ORGANIZATION REPORT NUMBER	
404 East Ramsey Road, San Antonio, Texas 782	AF5065				
9. SPONSORING/MONITORING AGENCY NAME(S) AND ADDRESS(S) 10. SPONSORING/MONITORING AGENCY NAME(S) AND ADDRESS(S)					
AFCEC/EXEW					
2261 Hughes Avenue, S	A001A				
Lackiand APD, Texas 7					
11. SUPPLEMENTARY NOTE	S				
12a. DISTRIBUTION/AVAILABILITY STATEMENT Unlimited 12b. DISTRIBUTION CODE					
13. ABSTRACT (Maximum 200	words)		•		
This is a Preliminary Assessmer compounds may have been releas	at Report of locations at Patiest sed to the environment through	ick Air Force Base, Brev gh the use or discharge of	ard County f aqueous fi	y, Florida where perfluorinated ilm forming foam.	
14. SUBJECT TERMS			15. NUN	15. NUMBER OF PAGES	
16. PRICE CODE				CE CODE	
17. SECURITY 18. SECURITY 19. SECURITY 20. LIMITATION OF					
CLASSIFICATION OF CLASSIFICATION OF CLASSIFICA				ABSTRACT.	
KLI UKI					
Unclassified Unclassified Unclassified		d	Unlimited		

TABLE OF CONTENTS

						Page
1.0	INT	RODU	CTION.			1-1
	1.1	BACK	GROUN	D		1-1
	1.2	PURP	OSE ANI	D OBJECT	IVES	1-2
	1.3	BASE	WIDE EN	VIRONM	ENTAL SETTING	1-2
		1.3.1	Geology			1-2
		1.3.2	Hydroge	ologic Setti	ing	1-3
		1.3.3	Hydrolo	gic Setting	-	1-4
		1.3.4	Ecologic	al Receptor	rs	1-4
	1.4	PREL	IMINAR	Y ASSESSN	MENT METHODS	1-5
	1.5	REPO	ORT ORG	ANIZATIO	DN	1-5
2.0	FIR	E TRA	INING A	REAS		2-1
	2.1	FORM	AER FIRE	E TRAININ	IG AREA 1	2-1
		2.1.1	Descript	ion and Op	erational History	2-1
		2.1.2	Waste C	haracteristi	CS	2-1
		2.1.3	Pathway	and Enviro	onmental Hazard Assessment	2-1
			2.1.3.1	Groundwa	ter Pathway	2-1
			2.1.3.2	Surface W	ater Pathway	2-1
			2.1.3.3	Soil and A	ir Exposure Pathways	2-1
	2.2	FORM	AER FIRE	E TRAININ	IG AREA 2	2-2
		2.2.1	Descript	ion and Op	erational History	2-2
		2.2.2	Waste C	haracteristi	cs	2-2
		2.2.3	Pathway	and Enviro	onmental Hazard Assessment	2-2
			2.2.3.1	Groundwa	ter Pathway	2-2
			2.2.3.2	Surface W	ater Pathway	2-3
			2.2.3.3	Soil and A	ir Exposure Pathways	2-3
3.0	NO	N-FIRF	E TRAIN	ING AREA	AS	3-1
	3.1	HAN	GARS	•••••		3-1
		3.1.1	Hangar	630		3-1
			3.1.1.1	Description	n and Operational History	3-1
			3.1.1.2	Waste Cha	aracteristics	3-1
			3.1.1.3	Pathway a	nd Environmental Hazard Assessment	3-3
				3.1.1.3.1	Groundwater Pathway	3-3
				3.1.1.3.2	Surface Water Pathway	3-3
				3.1.1.3.3	Soil and Air Exposure Pathways	3-4
		3.1.2	Hangar	647		3-4
			3.1.2.1	Description	n and Operational History	3-4
			3.1.2.2	Waste Cha	aracteristics	3-5
			3.1.2.3	Pathway a	nd Environmental Hazard Assessment	3-6
				3.1.2.3.1	Groundwater Pathway	3-6
				3.1.2.3.2	Surface Water Pathway	3-6
				3.1.2.3.3	Soil and Air Exposure Pathways	3-7

TABLE OF CONTENTS (Continued)

	3.1.3	Hangar	750		3-7
		3.1.3.1	Description	n and Operational History	3-7
		3.1.3.2	Waste Cha	racteristics	3-8
		3.1.3.3	Pathway and	nd Environmental Hazard Assessment	3-9
			3.1.3.3.1	Groundwater Pathway	3-9
			3.1.3.3.2	Surface Water Pathway	3-9
			3.1.3.3.3	Soil and Air Exposure Pathways	3-10
	3.1.4	Hangar	751		3-10
		3.1.4.1	Description	n and Operational History	3-10
		3.1.4.2	Waste Cha	racteristics	3-11
		3.1.4.3	Pathway and	nd Environmental Hazard Assessment	3-11
			3.1.4.3.1	Groundwater Pathway	3-11
			3.1.4.3.2	Surface Water Pathway	3-11
			3.1.4.3.3	Soil and Air Exposure Pathways	3-11
	3.1.5	Hangar	985	· · ·	3-11
		3.1.5.1	Description	n and Operational History	3-11
		3.1.5.2	Waste Cha	racteristics	3-12
		3.1.5.3	Pathway and	nd Environmental Hazard Assessment	3-12
			3.1.5.3.1	Groundwater Pathway	3-12
			3.1.5.3.2	Surface Water Pathway	3-12
			3.1.5.3.3	Soil and Air Exposure Pathways	3-12
	3.1.6	Hangar	986	· · ·	3-12
		3.1.6.1	Description	n and Operational History	3-12
		3.1.6.2	Waste Cha	racteristics	3-13
		3.1.6.3	Pathway and	nd Environmental Hazard Assessment	3-13
			3.1.6.3.1	Groundwater Pathway	3-13
			3.1.6.3.2	Surface Water Pathway	3-13
			3.1.6.3.3	Soil and Air Exposure Pathways	3-13
3.2	FIRE	STATIO	NS		3-13
	3.2.1	Fire Star	tion (Buildii	ng 810)	3-13
		3.2.1.1	Description	n and Operational History	3-13
		3.2.1.2	Waste Cha	racteristics	3-14
		3.2.1.3	Pathway and	nd Environmental Hazard Assessment	3-15
			3.2.1.3.1	Groundwater Pathway	3-15
			3.2.1.3.2	Surface Water Pathway	3-15
			3.2.1.3.3	Soil and Air Exposure Pathways	3-16
3.3	EMER	GENCY	RESPONS	Е	3-16
3.4	OTHE	R SPILL	S AND RE	LEASES	3-16
	3.4.1	Fire Tru	ick Rollovei	r Area	3-16
		3.4.1.1	Description	n and Operational History	3-16
		3.4.1.2	Waste Cha	racteristics	3-16
		3.4.1.3	Pathway and	nd Environmental Hazard Assessment	3-17

TABLE OF CONTENTS (Continued)

			3.4.1.3.1	Groundwater Pathway	3-17
			3.4.1.3.2	Surface Water Pathway	3-17
			3.4.1.3.3	Soil and Air Exposure Pathways	3-18
	3.4.2	Outfall 2	21 to Banan	a River	3-18
		3.4.2.1	Descriptio	n and Operational History	3-18
		3.4.2.2	Waste Cha	aracteristics	3-18
		3.4.2.3	Pathway a	nd Environmental Hazard Assessment	3-19
			3.4.2.3.1	Groundwater Pathway	3-19
			3.4.2.3.2	Surface Water Pathway	3-19
			3.4.2.3.3	Soil and Air Exposure Pathway	3-20
	3.4.3	Northern	n Sewage T	reatment Plant	3-20
		3.4.3.1	Descriptio	n and Operational History	3-20
		3.4.3.2	Waste Cha	aracteristics	3-21
		3.4.3.3	Pathway a	nd Environmental Hazard Assessment	3-21
			3.4.3.3.1	Groundwater Pathway	3-21
			3.4.3.3.2	Surface Water Pathway	3-22
			3.4.3.3.3	Soil and Air Exposure Pathway	3-22
	3.4.4	Building	; 705 Pump	House	3-23
		3.4.4.1	Descriptio	n and Operational History	3-23
		3.4.4.2	Waste Cha	aracteristics	3-23
		3.4.4.3	Pathway a	nd Environmental Hazard Assessment	3-24
			3.4.4.3.1	Groundwater Pathway	3-24
			3.4.4.3.2	Surface Water Pathway	3-24
			3.4.4.3.3	Soil and Air Exposure Pathway	3-25
	3.4.5	Building	<i>,</i> 984		3-25
		3.4.5.1	Descriptio	n and Operational History	3-25
		3.4.5.2	Waste Cha	aracteristics	3-25
		3.4.5.3	Pathway a	nd Environmental Hazard Assessment	3-26
			3.4.5.3.1	Groundwater Pathway	3-26
			3.4.5.3.2	Surface Water Pathway	3-26
		D '1 1'	3.4.5.3.3	Soil and Air Exposure Pathways	3-26
	3.4.6	Building	; 6/6 – Fue	I Truck Maintenance Facility	3-26
		3.4.6.1	Descriptio	n and Operational History	3-26
		3.4.6.2	Waste Cha	aracteristics	3-26
		3.4.6.3	Pathway a	nd Environmental Hazard Assessment	3-26
			3.4.6.3.1	Groundwater Pathway	3-26
			3.4.6.3.2	Surface Water Pathway	3-26
			5.4.0.5.3	Son and Air Exposure Pathways	3-27
SUM	IMARY	Y AND C	CONCLUS	[ONS	4-1
4.1	SUMN	/IARY			
	4.1.1	Fire Tra	ining Areas	3	

4.0

TABLE OF CONTENTS (Continued)

			4.1.2.2 4.1.2.3	Emergency Response	
			4.1.2.4	Other Spills or Releases	
	4.2	CONC	CLUSION	S	
5.0	REF	EREN	CES		5-1

Table 1.1	FTAs and Non-FTAs Identified for Potential AFFF Releases	1-2
Table 1.2	Endangered Species	1-4
Table 4.1	Preliminary Assessment Report Summary and Findings	4-4

LIST OF FIGURES

- Figure 1.1 Patrick Air Force Base
- Figure 2.1 Former Fire Training Area 1
- Figure 2.2 Former Fire Training Area 2
- Figure 3.1 Hangars 630 and 647
- Figure 3.2 Hangar 750, Hangar 751, and Building 705-
- Figure 3.3 Hangar 985
- Figure 3.4 Fire Station (Building 810)
- Figure 3.5 Fire Truck Rollover Area
- Figure 3.6 Outfall 21 to Banana River
- Figure 3.7 Northern Sewage Treatment Plant
- Figure 3.8 Building 984
- Figure 3.9 Building 676 –

LIST OF APPENDICES

- Appendix A Photo Documentation
- Appendix B Field Documentation
- Appendix C Records of Communication

AFB	Air Force Base
AFCEC	Air Force Civil Engineer Center
AFFF	aqueous film forming foam
Air Force	U.S. Air Force
ASR	aquifer storage and recovery
AST	aboveground storage tank
bgs	below ground surface
CERCLA	Comprehensive Environmental Response, Compensation, and Liability Act
EPA	Environmental Protection Agency
FDEP	Florida Department of Environmental Protection
FTA	Fire Training Area
HEF	high expansion foam
HGL	HydroGeoLogic, Inc.
μg/kg	microgram per kilogram
μg/L	microgram per liter
PA	Preliminary Assessment
PFOA	perfluorooctanoic acid
PFOS	pefluorooctane sulfonate
PFC	Perfluorinated Compounds
PHA	provisional health advisory
POC	point of contact
PWS	public water supply
RSSLs	risk-based soil screening levels
SES	SES Construction and Fuel Services
STP	Sewage Treatment Plant
USCT	underground storage collection tank

LIST OF ACRONYMS AND ABBREVIATIONS

FINAL PRELIMINARY ASSESSMENT REPORT FOR PERFLUORINATED COMPOUNDS PATRICK AIR FORCE BASE BREVARD COUNTY, FLORIDA

1.0 INTRODUCTION

HydroGeoLogic, Inc. (HGL) has been contracted by the Air Force Civil Engineer Center (AFCEC) to perform preliminary assessment (PA) activities at multiple U.S. Air Force (Air Force) and Air National Guard Fire Training Areas (FTAs) and Non-FTAs to determine locations of potential environmental release of perfluorinated compounds (PFCs). Specifically, the HGL Team is to complete PA activities to determine potential releases of PFCs at 82 Air Force and Air National Guard installations from FTAs and other known and suspected releases of PFCs from Aqueous Film Forming Foam (AFFF) usage or storage areas. This work is being performed by HGL under the existing 4P Architecture and Engineering contract, Contract No. FA8903-08-D-8772, Task Order 0065.

HGL conducted activities associated with this PA at Patrick Air Force Base (AFB) during the week of May 26, 2015, in accordance with the Comprehensive Environmental Response, Compensation, and Liability Act of 1980 (CERCLA) and the Superfund Amendments and Reauthorization Act of 1986 Preliminary Assessment processes. Patrick AFB is an Air Force Space Command installation located between Satellite Beach and Cocoa Beach in Brevard County, Florida (Figure 1.1). The installation is home to the 45th Space Wing which controls and operates Cape Canaveral Air Force Station.

1.1 BACKGROUND

PFCs are compounds used in the formulation of AFFF, which the Air Force has used in fire training exercises, suppressing aircraft and other vehicle fires, and in aircraft hangar fire suppression systems. Although PFCs are not regulated under CERCLA or the Resource Conservation and Recovery Act, there is evidence that pefluorooctane sulfonate (PFOS) and perfluorooctanoic acid (PFOA), which can be found in the environment following an AFFF release, may present potential, non-carcinogenic risks to human health and the environment (Chang et al., 2014; Porter, 2011; Rak et al., 2009).

Several federal government documents confirm the initial use of AFFF by the Air Force beginning in 1970:

- MILSpec for AFFF (MIL-F-24385) formally issued in 1969;
- General Accounting Office determination on sole source award protest to provide AFFF to the Navy in December 1969; and
- A History of Fire Protection Training at Chanute AFB, 1964-1976 (Coates, 1977).

Based on Air Force performance testing results on AFFF, the Air Force Director of Civil Engineering, M.G. Goddard, in 1970 issued authorization for the Air Force to procure AFFF. No usage within the Air Force is documented or suspected prior to 1970.

1.2 PURPOSE AND OBJECTIVES

The purpose and objective of this PA is to identify locations at Patrick AFB where PFCs may have been released to the environment and to conduct an initial assessment of possible migration pathways and receptors of potential contamination.

This PA report documents the known FTAs, as well as additional locations (non-FTAs), where AFFF may have been released into the environment at Patrick AFB (Table 1.1). Locations that are considered non-FTAs include, but are not limited to, hangars, fire stations, emergency response locations and any other locations where the potential exists for AFFF to have been released into the environment. This PA report also differentiates between locations that pose little or no potential threat to human health and the environment from locations that warrant further investigation.

1 1115 and 1000 1 1115 Identified for 1 otential 11111 Releases
Fire Training Areas
Former Fire Training Area 1
Former Fire Training Area 2
Non-Fire Training Areas
Hangars
Hangar 630
Hangar 647
Hangar 750
Hangar 751
Hangar 985
Hangar 985
Fire Stations
Fire Station (Building 810)
Other Spills and Releases
Fire Truck Rollover Area
Outfall 21 to Banana River
Building 705
Building 984
Building 676 –

Table 1.1
FTAs and Non-FTAs Identified for Potential AFFF Releases

1.3 BASEWIDE ENVIRONMENTAL SETTING

1.3.1 Geology

Patrick AFB is situated on undifferentiated marine sands overlying the Pleistocene-age Anastasia Formation and Caloosahatchee Marl Formation; these three units comprise the surficial unconsolidated deposits in the area. Above the Caloosahatchee Marl Formation, the surficial deposits form a shallow unconfined aquifer. The Anastasia Formation is a discontinuous layer of undifferentiated sands with silt and shells that may not be present in this

Air Force Civil Engineer Center
1-2

area. The Caloosahatchee Marl Formation consists primarily of calcareous sandy clay deposits. Underlying the Caloosahatchee Marl Formation is the Tamiami Formation, which is made up of limestones, marls, silty sands, and clay. The Tamiami Formation forms a shallow bedrock aquifer. The marine sands, clays, and limestones of the Hawthorn Formation underlie the Tamiami Formation. Interspersed limestone layers form localized aquifers within the Hawthorn Formation. Beneath the Hawthorn Formation is the Floridan Aquifer, which is comprised of Ocala Formation limestone and extends to a depth of over 1,500 feet below mean sea level (Parsons Engineering Science, Inc., 1995).

1.3.2 Hydrogeologic Setting

The aquifer systems present in Brevard County include the surficial aquifer and the Floridan Aquifer. The surficial aquifer system is contained in undifferentiated Late Miocene, Pliocene, and Recent Pleistocene deposits. These deposits are composed primarily of medium to coarse quartz sand with coquina and shell occurring more frequently with increasing depths. The surficial aquifer is hydraulically separated from the underlying Floridan aquifer by sediments of the Hawthorn Formation of Miocene Age. The low permeability clays, silts, and marls of the Hawthorn Formation act as an aquitard between the surficial and the Floridan aquifer systems. The Floridan aquifer system consists of a series of highly permeable limestone formations, including the Ocala Formation and the Avon Park Limestone, both of Eocene age (SES, 2014).

The surficial aquifer contains groundwater under nonartesian conditions and is approximately 4 feet below ground surface (bgs). Water enters the aquifer through direct infiltration from percolation and rainwater. Groundwater in the zone of saturation of the surficial aquifer moves laterally toward canals or rivers at Patrick AFB. The groundwater in the surficial aquifer is typically classified by the Florida Department of Environmental Protection (FDEP) as Class G-II (less than 10,000 milligrams per liter total dissolved solids). Class G-II is defined as being able to supply water treatable for human consumption. However, the surficial aquifer is not used to supply potable water at Patrick AFB (SES, 2014).

Shallow groundwater flow at Patrick AFB generally follows surface topography and flows from the dune ridges east of the Base toward the Banana River along the west side of the Base. Groundwater in the Floridan aquifer is under artesian conditions and flows northeast in the vicinity of Patrick AFB. The water enters the aquifer near the center of the Florida peninsula and moves laterally toward the coasts. The Floridan aquifer is also classified by FDEP as Class G-II (SES, 2014).

Patrick AFB receives potable water from the City of Cocoa, whose water supply is from a combination of groundwater wells, aquifer storage and recovery (ASR) wells, and surface water. The City of Cocoa is located approximately 13 miles northwest from Patrick AFB. The groundwater is acquired from the intermediate and Floridan Aquifer from 48 groundwater wells located in east Orange County. The city also stores approximately 1 billion gallons of treated water 300 feet underground in ASR wells. An additional source of water is surface water from the Taylor Creek Reservoir. The City of Cocoa treats ground and surface water at the Dyal Water Treatment Plant. The groundwater, ASR water, and surface water are blended

for distribution and provided to Patrick AFB via underground piping. The water is further treated by Patrick AFB with the addition of chlorine and is distributed throughout Patrick AFB (Patrick AFB, 2013).

1.3.3 Hydrologic Setting

There are no natural drainage features at Patrick AFB. Surface water runoff at Patrick AFB either infiltrates the ground surface or is controlled by a series of drainage channels, manmade ditches, culverts, and canals. These structures form a drainage system at Patrick AFB that moves surface water westward to discharge points along the Banana River (SES, 2014).

The Banana River is a 31-mile long lagoon system that lies between Cape Canaveral and Merritt Island. Merritt Island is located west of the southern portion of the river and Cape Canaveral is located east of the northern portion of the river. Patrick AFB is situated along the eastern banks of the southern portion of the river. The river is part of the Indian River Lagoon system and has only one outlet to the Atlantic Ocean (Banana River, Florida, n.d.).

1.3.4 Ecological Receptors

Ecological receptors include any living organisms other than humans, the habitat that supports such organisms or natural resources that could be adversely affected by environmental contaminations resulting by a release at or migration from an identified location.

Patrick AFB is located in Brevard County, Florida, and is surrounded by multiple wetlands and a Florida Habitat Conservation Area. These sensitive environments and the diversity of plants and animal species that inhabit them are considered primary ecological receptors for Patrick AFB (EDR, 2015c). Table 1.2 provides a list of endangered species for Brevard County that have the potential to inhabit the aforementioned sensitive environments.

Table 1 3

Endangered Species
BIRDS
Red-Cockaded Woodpecker
Piping Plover
Bald Eagle
Wood Stork
Florida Scrub Jay
PLANTS
Johnson's Seagrass
MAMMALS
West Indian (Florida) Manatee
Southeastern Beach Mouse
REPTILES
Loggerhead Sea Turtle
Leatherback Sea Turtle
Green Sea Turtle
Eastern Indigo Snake
Atlantic Salt Marsh Snake

1.4 PRELIMINARY ASSESSMENT METHODS

The performance of this PA included:

- Reviewing information and reports in the available Administrative Record.
- Reviewing documents related to Air Force use of AFFF.
- Conducting a PA visit at Patrick AFB.
- Conducting interviews with base environmental management personnel, Patrick AFB personnel, and aircraft hangar maintenance and operations personnel.
- Photographing locations where AFFF has been used.
- Performing an environmental data records search to document nearby populations, water supply well information, and wetlands.

If the operational history of an identified location indicates that AFFF was not used, then no exposure pathway could exist and the pathway and environmental hazard assessments within the PA will not be applicable.

1.5 REPORT ORGANIZATION

This PA report is organized as follows:

- Section 1.0, Introduction, includes a project overview, provides a basewide environmental setting, and describes the methods used to conduct the PA.
- Section 2.0, Fire Training Areas, describes the FTAs identified during the PA visit.
- Section 3.0, Non-Fire Training Areas, describes the non-FTAs identified during the PA visit.
- Section 4.0, Summary and Conclusions, summarizes conclusions for both FTAs and non-FTAs.
- Section 5.0, References, provides references consulted during the preparation of this PA report.
- Appendix A, Photo Documentation, provides photos taken during the PA visit.
- Appendix B, Field Documentation, provides the Potential Hazardous Waste Site Preliminary Assessment Forms.
- Appendix C, Records of Communications, provides records of all interviews conducted during the PA visit.

FIGURE



HGL—Preliminary Assessment Report Patrick Air Force Base, Brevard County, Florida

Figure 1.1 Patrick Air Force Base Brevard County, Florida

Legend

Street

Surface Water

.....

Inferred Location Boundary

Installation Boundary

\\gst-srv-01\HGLGIS\PA_Sites\Patrick_AFB\PA_Report\ (1-01)Patrick_AFB.mxd 7/31/2015 SS Source: HGL, Patrick AFB ArcGIS Online Imagery



2.0 FIRE TRAINING AREAS

2.1 FORMER FIRE TRAINING AREA 1

2.1.1 Description and Operational History

Former FTA 1 (Site FT-21 [Solid Waste Management Unit #032]) was located along the eastern central portion of Patrick AFB, in the general location of Building 820 (Figure 1.1 and Figure 2.1). Former FTA 1 operated from 1950 to 1963 and was used for burning waste fuels (aviation gasoline, motor gasoline, and diesel), waste oils, halogenated and non-halogenated solvents during firefighting training exercises. The FTA consisted of approximately 0.25 acres with a shallow unlined depression in sandy soils into which combustible materials were placed and ignited during fire training activities (O'Brien & Gere Engineers, Inc., 1993). There was no evidence of previous fire training activities at the location of Former FTA 1 during the PA visit. The geographic coordinates of the former FTA are

The Assistant Fire Chief was not aware of any AFFF releases at Former FTA 1 as the operational timeframe of the FTA was prior to his tenure as acting Assistant Fire Chief (Appendix C, Records of Communication). In addition, the operational timeframe of Former FTA 1 predates the use of AFFF by the Air Force; therefore, impact to the environmental media surrounding Former FTA 1 is unlikely.

Photographic documentation is provided in Appendix A.

2.1.2 Waste Characteristics

Not Applicable.

2.1.3 Pathway and Environmental Hazard Assessment

Not Applicable.

2.1.3.1 Groundwater Pathway

Not Applicable.

2.1.3.2 Surface Water Pathway

Not Applicable.

2.1.3.3 Soil and Air Exposure Pathways

Not Applicable.

2.2 FORMER FIRE TRAINING AREA 2

2.2.1 Description and Operational History

Former FTA 2 (Site FT-22) was located on the northwestern portion of Patrick AFB (Figure 1.1) and operated from 1963 to 1985. Former FTA 2 was used as a firefighting training area and consisted of a burn pit to which petroleum wastes and waste products from industrial solvents/degreasing operations were applied and ignited (URS, 2005). The pit was an unlined circular burn pit approximately 2 feet in depth and 150 feet in diameter that was used for igniting combustible wastes (Environmental Science and Engineering, Inc., 1988). The pit is bordered to the north, east, and south by lightly vegetated undeveloped areas and to the west by a concrete pathway/sidewalk followed by the Banana River (Figure 2.2). There was no evidence of previous fire training activities at the general area of burn pit 1 during the PA visit. The geographic coordinates of the central portion of Former FTA 2 are

Photographic documentation is provided in Appendix A.

2.2.2 Waste Characteristics

The Assistant Fire Chief was not aware of fire training activities conducted at Former FTA 2 as the operational timeframe of the FTA was prior to his tenure as acting Assistant Fire Chief (Appendix C, Records of Communication). Based on the use of AFFF by the Air Force beginning in 1970, the potential exists for AFFF having been used to extinguish fires at Former FTA 2 during the operational period from 1970 through 1985.

2.2.3 Pathway and Environmental Hazard Assessment

A complete exposure pathway typically includes the following components: a source of contamination (an environmental medium contaminated at the source or a release mechanism by which chemicals are released from a source medium and transported), an exposure medium by which a receptor comes into contact, and a route of intake for the contaminant into the receptor's body at the exposure point. If any of these elements are missing, the pathway is incomplete. Other release mechanisms resulting in exposure media for receptors may include the uptake of soil contaminants by plants and animals and the emission of soil contaminants into the air in association with dust particles (EPA, 1989).

2.2.3.1 Groundwater Pathway

The basewide geologic and hydrogeologic settings are provided in Section 1.3. Groundwater in the general area of Former FTA 2 is assumed to follow the basewide shallow groundwater flow westward toward the Banana River. The Banana River is the closest surface water body and is located approximately 100 feet west of burn pit 1 and 374 feet west of burn pit 2. The potential presence of PFCs in groundwater at Former FTA 2 exists based on the operational time frames of the FTA.

Patrick AFB and surrounding off-base communities receive drinking water from the City of Cocoa, which acquires water from groundwater in the Floridan aquifer, ASR wells, and surface water as discussed in Section 1.3.2. The closest public water supply (PWS) well is located 4.56 miles southwest of Former FTA 2. The PWS well is part of the FL3050985 Palm Shores RV Park well system, a non-community public water system that serves a population of 25 residents (EDR, 2015a).

The combined on- and off-base population within a 4-mile radius from Former FTA 2 is approximately 20,379 residents (EDR, 2015b). The closest residential area is an on-base residential campground located approximately 2,604 feet south of Former FTA 2 (EDR, 2015b).

2.2.3.2 Surface Water Pathway

The topography in the general area of Former FTA 2 is relatively flat. As discussed in Section 1.3.3 there are no drainage features at Patrick AFB. However, based on the close proximity of the Banana River, surface water would likely flow west overland surface and discharge to the Banana River. The Banana River merges with the Indian River approximately 8.5 miles downstream (south) from the discharge point. The Indian River continues to flow in a northern direction over 15 miles (Geofin, 2015).

The Banana River is identified as a wetland and is classified as E1UBL: (E)-Esturaine, (1)-Subtidal, (UB)-Unconsolidated Bottom, (L)-Subtidal. Multiple wetlands are also identified 15 miles downstream of Former FTA 2 along the banks of the Banana River and Indian River (EDR, 2015c). Ingestion of surface water by wildlife at these wetlands is a potential pathway for ecological receptors. These wetlands are identified as an ecologically sensitive environment adjacent to the surface water migration pathway. The bodies of water encountered downstream of Former FTA 2 are known to be used for recreational fishing by residents or nearby communities, and could provide an exposure to humans through dermal contact and ingestion of fish (Banana River, Florida, n.d.).

Former FTA 2 is located within the 100-year flood zone (EDR, 2015c). There are no surface water intakes or downstream fisheries adjacent to the surface water migration path 15 miles downstream of the Former FTA 2 (EDR, 2015c; Geofin, 2015).

2.2.3.3 Soil and Air Exposure Pathways

The Former FTA 2 is an inactive FTA currently covered with light vegetation and no remnants of a burn pit. Former FTA 2 is not located within a restricted area and is accessible by all civilians and Patrick AFB personnel. There are no residents or workers on site. The closest building with workers is Building 9693, located approximately 294 feet northeast of burn pit 2. The potential exists for soil exposure to the workers at this building; however, direct contact by workers with soil associated with Former FTA 2 is not anticipated. Landscape workers who perform mowing at the Former FTA could potentially be exposed to soil through the emission of soil contaminants into the air as dust particles while mowing. The potential exists for soil exposure to burrowing animals. The closest residential area is an on-

base residential campground located approximately 2,604 feet south of Former FTA 2 (EDR, 2015b). Population details within a 4-mile radius are discussed in Section 2.2.3.1.

There are no daycare facilities or schools within a 200-foot radius of Former FTA 2. The closest school is Sea Park Elementary School, located approximately 2.9 miles south/southeast. The closest daycare is the Child Development Center located on base, approximately 1.5 miles southeast (EDR, 2015b).

FIGURES



HGL—Preliminary Assessment Report Patrick Air Force Base, Brevard County, Florida

Figure 2.1 Former Fire Training Area 1 Patrick Air Force Base Brevard County, Florida

Legend

- Storm Sewer Inlet
- Storm Sewer Pipeline
- Storm Sewer Culvert
- Storm Sewer Open Drainage
- Surface Water
- 699 Building Number
- Inferred Location Boundary
- Installation Boundary

\\Gst-srv-01\HGLGIS\PA_Sites\Patrick_AFB\PA_Report\ (2-01)Former_FTA_1.mxd 9/17/2015 JP Source: HGL, Patrick AFB ArcGIS Online Imagery





HGL—Preliminary Assessment Report Patrick Air Force Base, Brevard County, Florida

Figure 2.2 Former Fire Training Area 2 Patrick Air Force Base Brevard County, Florida

Legend

- Storm Sewer Inlet
- ► Storm Sewer Pipeline
- Storm Sewer Open Drainage
- Surface Water
- 699 Building Number
- Inferred Location Boundary
- Installation Boundary

\\Gst-srv-01\HGLGIS\PA_Sites\Patrick_AFB\PA_Report\ (2-02)Former_FTA_2.mxd 9/17/2015 SS Source: HGL, Patrick AFB ArcGIS Online Imagery



3.0 NON-FIRE TRAINING AREAS

3.1 HANGARS

Hangars at Patrick AFB either have deluge (water) fire suppression systems, or fixed foam fire suppression systems that are either AFFF or high expansion foam (HEF). The fire department staff and suppression system installers (or operation and maintenance staff) were interviewed for the types of systems that have been used historically and currently at each hangar. According to the Assistant Fire Chief, hangars are tested every two years and the foam is sampled annually. A flow test is performed during the testing and requires a system dump, but only with water. AFFF is not released during the testing process.

3.1.1 Hangar 630

3.1.1.1 Description and Operational History

Hangar 630 is located in the northwestern portion of Patrick AFB. The hangar is bordered to the north and west by restricted airfield ramps, to the east by grassy areas followed by Hangar 647, and to the south by grassy areas followed by Building 632 and Taxiway J (Figure 3.1). The geographic coordinates for Hangar 630 are

Hangar 630 was constructed in 1964 and is currently equipped with an AFFF fire suppression system. The AFFF fire suppression system was installed in 1999 and was upgraded to a new AFFF fire suppression system in 2014. The hangar AFFF fire suppression system has always been supplied with 3% AFFF from an 800-gallon aboveground storage tank (AST) located in the hangar mechanical room (Appendix C, Records of Communication).

Photographic documentation is provided in Appendix A.

3.1.1.2 Waste Characteristics

In the event of a discharge from the fire suppression system, foam is discharged to the hangar floor and flows into the hangar floor drains. Prior to 1995, discharge into the hangar floor drains was directed to the Northern Sewage Treatment Plant (STP) and the effluent was released to the Banana River. Subsequent to 1995, discharge to the hangar floor drains flowed to the City of Cocoa's Dyal Water Treatment Plant. According to interviews with base personnel, a 30,000-gallon underground storage collection tank (USCT) was installed in June/July 2014 on the west/northwest side of Hangar 630. The USCT was installed to contain drainage from the floor drains of Hangars 630 and 647. The contents of the USCT are pumped out by a subcontractor for off-base disposal. The frequency at which the USCT is pumped out and disposed of is unknown (Appendix C, Records of Communication).

According to the Assistant Fire Chief, the fire suppression system was tripped in 1999 and AFFF was discharged inside the hangar. Approximately three feet of foam filled the inside of the hangar. The AFFF drained down the hangar floor drains where it would have been

pumped to the City of Cocoa's Dyal Water Treatment Plant. The Assistant Fire Chief could not recall the amount of AFFF discharged (Appendix C, Records of Communication).

The Fire System point of contact (POC) recalled another AFFF discharge at Hangar 630 that occurred in 2004. The AFFF fire suppression system was tripped due to a hurricane in the area and resulted in the discharge of AFFF into the hangar floor drains and outside the hangar doors. The AFFF that drained into the hangar floor drains would have been pumped to the City of Cocoa's Dyal Water Treatment Plant. The AFFF that flowed out the hangar doors infiltrated the ground surface at the grassy areas north of the hangar and drained into the storm sewer inlets. Drainage into the storm sewer inlets flows west via underground storm sewer pipelines and releases to a drainage canal through Outfall 21. The drainage canal is located approximately 470 feet west/northwest of the hangar. Outfall 21 is further discussed in Section 3.4.2. The Fire System POC was not aware of the amount of AFFF discharged in the hangar or outside of the hangar doors (Appendix C, Records of Communication).

In October 2013, SES Construction and Fuel Services (SES) performed an investigation to identify potential PFC usage areas and to select locations for further evaluation. Hangar 630 was identified as a location for further investigation based on several accidental AFFF discharges that occurred from 1998 to 2005. The discharges resulted in AFFF running out the hangar doors to the grassy areas north and south of the hangar. Some AFFF discharges reportedly flowed across the grass to storm sewer inlets leading to the drainage canal through Outfall 21 (SES, 2014). The SES investigation did not identify the discharges of AFFF that occurred from 1998 to 2005 and they were not confirmed during this PA visit, with the exception of those listed.

On April 30, 2014, four subsurface soil samples and three groundwater samples were collected from the grassy area north of the hangar and two surface soil samples, four subsurface soil samples, and four groundwater samples were collected from the grassy area south of the hangar. Additionally, two surface water and three sediment samples were collected from the drainage canal as part of the limited investigation (SES, 2014). The samples collected from the drainage canal are further discussed in Section 3.4.2.

Groundwater analytical results indicated that PFOA and PFOS were detected in all the groundwater samples collected in the grassy area north of the hangar. The PFOA and PFOS detections in all three groundwater samples were reported above the corresponding Environmental Protection Agency (EPA) provisional health advisory (PHA) values of 0.4 micrograms per liter (μ g/L) and 0.2 μ g/L, respectively. The subsurface soil analytical results indicated that PFOA was detected in two of the four samples and PFOS was detected in three of the four samples collected at the grassy area north of the hangar. The PFOA and PFOS detections were reported below their respective EPA risk-based soil screening levels (RSSLs) of 16,000 micrograms per kilogram (μ g/kg) and 6,000 μ g/kg (SES, 2014).

Analytical results for the soil samples (2 surface and 4 subsurface) collected in the grassy area south of the hangar indicated that PFOA was detected in two of the six soil samples collected and PFOS was detected in all six samples. All detected concentrations of PFOA and PFOS in soils were reported below the respective EPA RSSLs of 16,000 μ g/kg and 6,000 μ g/kg. The

groundwater analytical results indicated that PFOA and PFOS were detected in all four groundwater samples collected in the grassy area south of the hangar. None of the PFOA detections were reported above the EPA PHA value of 0.4 μ g/L. However, the PFOS detections were all reported above the EPA PHA value of 0.2 μ g/L (SES, 2014).

The SES investigation confirmed the presence of PFCs in the environmental media surrounding Hangar 630, specifically the grassy areas north and south of the hangar (SES, 2014).

3.1.1.3 Pathway and Environmental Hazard Assessment

A complete exposure pathway typically includes the following components: a source of contamination (an environmental medium contaminated at the source or a release mechanism by which chemicals are released from a source medium and transported), an exposure medium by which a receptor comes into contact, and a route of intake for the contaminant into the receptor's body at the exposure point. If any of these elements are missing, the pathway is incomplete. Other release mechanisms resulting in exposure media for receptors may include the uptake of soil contaminants by plants and animals and the emission of soil contaminants into the air in association with dust particles (EPA, 1989).

3.1.1.3.1 Groundwater Pathway

The basewide geologic and hydrogeologic settings are provided in Section 1.3. According to the SES investigation, groundwater in the general area of the hangar was gauged at depths ranging from 3.5 to 5 feet bgs. Groundwater flow is assumed to follow the basewide shallow groundwater flow westward toward the Banana River, approximately 1,042 feet west of the hangar. According to the SES investigation, the presence of PFCs was confirmed in the shallow groundwater underlying the area surrounding Hangar 630.

Patrick AFB and surrounding off-base communities receive drinking water from the City of Cocoa, which acquires water from groundwater in the Floridan aquifer, ASR wells, and surface water as discussed in Section 1.3.2. The closest PWS well is located 4.8 miles southwest of the hangar. The PWS well is part of the FL3050985 Palm Shores RV Park well system, a non-community public water system that serves a population of 25 residents (EDR, 2015a).

The combined on- and off-base population within a 4-mile radius from Hangar 630 is approximately 16,416 (EDR, 2015b). The closest residential area is an on-base residential campground located approximately 3,573 feet south of the hangar.

3.1.1.3.2 Surface Water Pathway

There are no natural drainage features at Patrick AFB as discussed in Section 1.3.3. Surface water runoff at Patrick AFB either infiltrates the ground surface or is controlled by a series of drainage channels, manmade ditches, culverts, and canals. Drainage from Hangar 630 follows the surface topography from east to west toward a drainage canal located approximately 440

feet west of the hangar. The drainage canal discharges at Outfall 21 to the Banana River which merges with the Indian River approximately 8.7 miles downstream (south) from the discharge point. The Indian River flows in a northern direction over 15 miles (Geofin, 2015).

The Banana River is identified as a wetland and is classified as E1UBL: (E)-Esturaine, (1)-Subtidal, (UB)-Unconsolidated Bottom, (L)-Subtidal. Multiple wetlands are also identified 15 miles downstream of the hangar along the banks of the Banana River and Indian River (EDR, 2015c). Ingestion of surface water by wildlife at these wetlands is a potential pathway for ecological receptors. These wetlands are identified as an ecologically sensitive environment potentially adjacent to the surface water migration pathway. Additionally, the Banana River and Indian River are known to be used for recreational activities including fishing and boating by residents or nearby communities, providing an exposure to humans through dermal contact and ingestion of fish (Banana River, Florida, n.d.).

Hangar 630 is not located within a flood plain and there are no surface water intakes or downstream fisheries adjacent to the surface water migration path 15 miles downstream of the hangar (EDR, 2015c; Geofin, 2015).

3.1.1.3.3 Soil and Air Exposure Pathways

According to the SES investigation, the presence of PFCs in the soil at Hangar 630 has been confirmed (SES, 2014). Hangar 630 is an active hangar located in a restricted area of the airfield and is accessible to authorized military personnel, the base fire department, and escorted guests. The number of workers at the hangar varies depending on the type of aircraft maintenance being performed. Although the potential exists for soil exposure to workers at the hangar, direct contact by workers with soil is not anticipated. There are no residents at the hangar. The closest residential area is 3,573 feet south of the hangar. Population details of the residential areas within a 4-mile radius are discussed in Section 3.1.1.3.1.

There are no daycare facilities or schools within a 200-foot radius of the hangar. The closest school is Sea Park Elementary School, located approximately 3.1 miles south/southeast of Hangar 630. The closest daycare is the Child Development Center located on base, approximately 1.5 miles south/southeast (EDR, 2015b).

3.1.2 Hangar 647

3.1.2.1 Description and Operational History

Hangar 647 is located in the northwestern portion of Patrick AFB. The hangar is bordered to the north by grassy areas and a restricted airfield ramp, to the east by grassy areas followed by Building 610, to the south by grassy areas and a parking area followed by Building 651, and to the west by grassy areas followed by Hangar 630 (Figure 3.1). The geographic coordinates for Hangar 647 are

Hangar 647 is a fuel cell maintenance hangar that was constructed in 1970 and is currently equipped with an AFFF fire suppression system. The hangar is supplied with 3% AFFF from

a 2,000-gallon AST located in the hangar mechanical room (Appendix C, Records of Communication).

Photographic documentation is provided in Appendix A.

3.1.2.2 Waste Characteristics

In the event of a discharge from the fire suppression system, foam is discharged to the hangar floor and flows into the hangar floor drains. Prior to 1995, discharge into the hangar floor drains was directed to the Northern STP and the effluent was released to the Banana River. Subsequent to 1995, discharge to the hangar floor drains flowed to the City of Cocoa's Dyal Water Treatment Plant. According to interviews with base personnel, a 30,000-gallon USCT was installed in June/July 2014 on the west/northwest side of Hangar 630. The USCT was installed to collect drainage from the floor drains of Hangars 630 and 647. The contents of the USCT are pumped out by a subcontractor for off-base disposal. The frequency at which the USCT is pumped out and disposed of is unknown (Appendix C, Records of Communication).

The Fire System POC recalled an AFFF discharge at Hangar 647 that occurred in 2004. The AFFF fire suppression system was tripped due to a hurricane in the area and AFFF discharged into the hangar floor drains and outside the hangar doors. The AFFF that drained into the hangar floor drains would have flowed to the City of Cocoa's Dyal Water Treatment Plant. The AFFF that flowed out of the hangar doors infiltrated the ground surface at the grassy areas north of the hangar and drained into the storm sewer inlets. Drainage into the storm sewer inlets flows west via underground storm sewer pipelines and releases to a drainage canal through Outfall 21. The drainage canal is located approximately 720 feet west/northwest. Outfall 21 is further discussed in Section 3.4.2. The Fire System POC was not aware of the amount of AFFF discharged in the hangar or outside of the hangar doors.

In October 2013, SES performed an investigation to identify potential PFC usage areas and to select locations for further evaluation. Hangar 647 was identified as a location for further investigation based on several accidental discharges of AFFF that occurred from 1998 to 2005 (SES, 2014). The discharges resulted in AFFF flowing out the hangar doors to the grassy area north and south of the hangar. Some releases reportedly flowed across the grass to storm sewer inlets leading to the drainage canal through Outfall 21 (SES, 2014). The SES investigation did not identify the discharges of AFFF that occurred from 1998 to 2005 and they were not confirmed during this PA visit, with the exception of those listed.

On April 30, 2014, four subsurface soil samples and three groundwater samples were collected from the grassy area north of the hangar. Additionally, two surface water and two sediment samples were collected from the drainage canal as part of the investigation. The samples collected form the drainage canal are further discussed in Section 3.4.2, and the samples collected from the grassy area south of the hangar are discussed in Section 3.1.5.

Groundwater analytical results indicated that PFOA and PFOS were detected in all three collected groundwater samples. The PFOA and PFOS detections in all three groundwater samples were reported above the corresponding EPA PHA value of $0.4 \ \mu g/L$ and $0.2 \ \mu g/L$,

respectively. The subsurface soil analytical results indicated that PFOA was detected in two of the four samples and PFOS was detected in three of the four samples. The PFOA and PFOS detections were reported below their respective EPA RSSLs of 16,000 μ g/kg and 6,000 μ g/kg (SES, 2014).

The SES investigation confirmed the presence of PFCs in the environmental media surrounding Hangar 647, specifically the grassy area north of the hangar (SES, 2014).

3.1.2.3 Pathway and Environmental Hazard Assessment

A complete exposure pathway typically includes the following components: a source of contamination (an environmental medium contaminated at the source or a release mechanism by which chemicals are released from a source medium and transported), an exposure medium by which a receptor comes into contact, and a route of intake for the contaminant into the receptor's body at the exposure point. If any of these elements are missing, the pathway is incomplete. Other release mechanisms resulting in exposure media for receptors may include the uptake of soil contaminants by plants and animals and the emission of soil contaminants into the air in association with dust particles (EPA, 1989).

3.1.2.3.1 Groundwater Pathway

The basewide geologic and hydrogeologic settings are provided in Section 1.3. According to the SES investigation, groundwater in the general area of the hangar was gauged at depths ranging from 4 to 5 feet bgs. Groundwater flow is assumed to follow the basewide shallow groundwater flow westward toward the Banana River, approximately 1,277 feet west of the hangar. According to the SES investigation, the shallow groundwater underlying the area surrounding Hangar 647 has been impacted by PFCs.

Patrick AFB and surrounding off-base communities receive drinking water from the City of Cocoa, which acquires water from groundwater in the Floridan aquifer, ASR wells, and surface water as discussed in Section 1.3.2. The closest PWS well is located 4.8 miles southwest of the hangar. The PWS well is part of the FL3050985 Palm Shores RV Park well system, a non-community public water system that serves a population of 25 residents (EDR, 2015a).

The combined on and off-base population within a 4-mile radius from Hangar 647 is approximately 16,416 (EDR, 2015b). The closest residential area is an on-base residential campground located approximately 3,569 feet south of the hangar.

3.1.2.3.2 Surface Water Pathway

There are no natural drainage features at Patrick AFB as discussed in Section 1.3.3. Surface water runoff at Patrick AFB either infiltrates the ground surface or is controlled by a series of drainage channels, manmade ditches, culverts, and canals. Drainage from Hangar 647 follows the surface topography from east to west toward a drainage canal at Outfall 21 located approximately 685 feet west of the hangar. The drainage canal discharges to the Banana River
which merges with the Indian River approximately 9.46 miles downstream (south) from the discharge point. The Indian River flows in a northern direction over 15 miles (Geofin, 2015).

The Banana River is identified as a wetland and is classified as E1UBL: (E)-Esturaine, (1)-Subtidal, (UB)-Unconsolidated Bottom, (L)-Subtidal. Multiple wetlands are also identified 15 miles downstream of the hangar along the banks of the Banana River and Indian River (EDR, 2015c). Ingestion of surface water by wildlife at these wetlands is a potential pathway for ecological receptors. These wetlands are identified as an ecologically sensitive environment potentially adjacent to the surface water migration pathway. Additionally, the Banana River and Indian River are known to be used for recreational activities including fishing and boating by residents or nearby communities, providing an exposure to humans through dermal contact and ingestion of fish (Banana River, Florida, n.d.).

Hangar 647 is not located within a flood plain and there are no surface water intakes or downstream fisheries adjacent to the surface water migration path 15 miles downstream of the hangar (EDR, 2015c; Geofin, 2015).

3.1.2.3.3 Soil and Air Exposure Pathways

The SES investigation confirmed the presence of PFCs in the soil at Hangar 647 (SES, 2014). Hangar 647 is an active hangar located in a restricted area of the airfield and is accessible to authorized military personnel, the base fire department, and escorted guests. The number of workers at the hangar varies depending on the type of aircraft maintenance being performed. Although the potential exists for soil exposure to workers at the hangar, direct contact by workers with soil is not anticipated. There are no residents at the hangar. The closest residential area is 3,569 feet south of the hangar. Population details of the residential areas within a 4-mile radius are discussed in Section 3.1.2.3.1.

There are no daycare facilities or schools within a 200-foot radius of Hangar 647. The closest school is Sea Park Elementary School, located approximately 3.0 miles south/southeast. The closest daycare is the Child Development Center located on base, approximately 1.5 miles south/southeast of the hangar (EDR, 2015b).

3.1.3 Hangar 750

3.1.3.1 Description and Operational History

Hangar 750 is located in the northern portion of Patrick AFB, southeast of the intersection of Rescue Road and Redstone Road. The hangar is bordered to the north by a parking lot, to the east and west by restricted airfield ramps, and to the south by grassy areas followed by the restricted airfield (Figure 3.2). The geographic coordinates for Hangar 750 are

Hangar 750 was constructed in 1943 and was initially equipped with a wet fire sprinkler system. In 1999, Hangar 750 was retrofitted with an AFFF fire suppression system that operated until 2006. In 2006, the fire suppression system was retrofitted to an HEF system.

Preliminary Assessment Report

When the hangar was equipped with an AFFF fire suppression system, a 1,200-gallon AST charged with 3% AFFF supplied the hangar's fire suppression system via underground piping. The AST is stored at the Building 705 Pump House, which is located approximately 50 feet northeast of the hangar and also supplies Hangar 751 (Appendix C, Records of Communication). The Building 705

The underground piping that supplied AFFF to Hangar 750 was capped off in 2006 prior to switching to the HEF fire suppression system. Hangar 750 is currently operating with an HEF fire suppression system that is supplied by an 800-gallon HEF AST located in the hangar mechanical room. Hangar 750 was not accessible during the PA visit.

Photographic documentation is provided in Appendix A.

3.1.3.2 <u>Waste Characteristics</u>

While the hangar included an AFFF fire suppression system, AFFF discharged from the system drained into the hangar floor drains, which flow via underground pipelines to a 30,000-gallon USCT. The USCT is located in the grassy area directly south of the hangar. According to interviews with personnel familiar with the long-term history of the hangar, the contents of the 30,000-gallon USCT have historically been pumped out to the surrounding ground surface or occasionally pumped out by a subcontractor using a vacuum truck and disposed of off base (Appendix C, Records of Communication).

In October 2013, SES performed an investigation to identify potential PFC usage areas and to select locations for further evaluation. Hangar 750 was identified as a location for further investigation based on an AFFF discharge that occurred in 2001. According to the investigation, approximately 1,200 gallons of AFFF concentrate was discharged from a supply tank at Hangar 750. The majority of the release was contained by the USCT; however, surface flow of AFFF over the tank area was observed (SES, 2014). However, according to interviews conducted for this PA, a 1,200-gallon AST located in Building 705 pump house supplied AFFF concentrate to Hangar 750. It appears that the discharge may have occurred from the supply tank located in the pump house and not in Hangar 750.

On April 30, 2014, SES collected five groundwater and eight soil samples (3 surface and 5 subsurface) from the grassy area surrounding the general location of the USCT. Groundwater analytical results indicated that PFOA and PFOS were detected in all the collected groundwater samples. The detected concentrations of PFOS in all five groundwater samples exceeded the PFOS EPA PHA value of 0.2 μ g/L. The detected concentrations of PFOA were reported below the PFOA EPA PHA value of 0.4 μ g/L (SES, 2014).

Surface soil analytical results indicated that PFOA and PFOS were detected in all three surface soil samples. The detected concentrations of PFOA and PFOS were reported below their respective EPA RSSLs of 16,000 μ g/kg and 6,000 μ g/kg. Subsurface analytical results indicated that PFOA was detected in only two of five subsurface soil samples and PFOS was detected in all five subsurface soil samples. The detections of PFOA and PFOS in subsurface

soil were reported below their respective EPA RSSLs of 16,000 μ g/kg and 6,000 μ g/kg (SES, 2014).

The SES investigation confirmed the presence of PFCs in the environmental media surrounding Hangar 750, specifically in the grassy area surrounding the USCT (SES, 2014).

3.1.3.3 Pathway and Environmental Hazard Assessment

A complete exposure pathway typically includes the following components: a source of contamination (an environmental medium contaminated at the source or a release mechanism by which chemicals are released from a source medium and transported), an exposure medium by which a receptor comes into contact, and a route of intake for the contaminant into the receptor's body at the exposure point. If any of these elements are missing, the pathway is incomplete. Other release mechanisms resulting in exposure media for receptors may include the uptake of soil contaminants by plants and animals and the emission of soil contaminants into the air in association with dust particles (EPA, 1989).

3.1.3.3.1 Groundwater Pathway

The basewide geologic and hydrogeologic settings are provided in Section 1.3. Groundwater in the general area of the hangar was detected at depths ranging from 4 to 5 feet bgs. Groundwater flow is assumed to follow the basewide shallow groundwater flow westward toward the Banana River, approximately 1,042 feet west. The SES investigation confirmed the presence of PFCs in the shallow groundwater underlying Hangar 750 (SES, 2014).

Patrick AFB and surrounding off-base communities receive drinking water from the City of Cocoa, which acquires water from groundwater in the Floridan aquifer, ASR wells, and surface water as discussed in Section 1.3.2. The closest PWS well is located 5.06 miles southwest of the hangar. The PWS well is part of the FL3050985 Palm Shores RV Park well system, a non-community public water system that serves a population of 25 residents (EDR, 2015a).

The combined on- and off-base population within a 4-mile radius from Hangar 750 is approximately 17,124 (EDR, 2015b). The closest residential area is located approximately 3,178 feet north of the hangar.

3.1.3.3.2 Surface Water Pathway

There are no natural drainage features at Patrick AFB as discussed in Section 1.3.3. Surface water runoff at Patrick AFB either infiltrates the ground surface or is controlled by a series of drainage channels, manmade ditches, culverts, and canals. Drainage from Hangar 750 will flow into the surrounding grassy areas and infiltrate the ground surface or possibly flow into the surrounding storm sewer inlets that discharge to the Banana River.

The Banana River merges with the Indian River approximately 9.2 miles downstream (south) from the discharge point. The Indian River flows in a northern direction over 15 miles (Geofin, 2015).

The Banana River is identified as a wetland and is classified as E1UBL: (E)-Esturaine, (1)-Subtidal, (UB)-Unconsolidated Bottom, (L)-Subtidal. Multiple wetlands are also identified 15 miles downstream of the hangar along the banks of the Banana River and Indian River (EDR, 2015c). Ingestion of surface water by wildlife at these wetlands is a potential pathway for ecological receptors. These wetlands are identified as an ecologically sensitive environment potentially adjacent to the surface water migration pathway. Additionally, the Banana River and Indian River are known to be used for recreational activities including fishing and boating by residents or nearby communities, providing an exposure to humans through dermal contact and ingestion of fish (Banana River, Florida, n.d.).

Hangar 750 is not located within a flood plain and there are no surface water intakes or downstream fisheries adjacent to the surface water migration path downstream of the hangar (EDR, 2015c).

3.1.3.3.3 Soil and Air Exposure Pathways

The SES investigation confirmed the presence of PFCs in soil at Hangar 750. Hangar 750 is an active hangar located in a restricted area of the airfield and is accessible to authorized military personnel, the base fire department, and escorted guests. The number of workers at the hangar varies depending on the type of aircraft maintenance being performed. Although the potential exists for soil exposure to workers at the hangar, direct contact by workers with soil is not anticipated. There are no residents at the hangar. The closest residential area is 3,178 feet north of the hangar. Population details of the residential areas within a 4-mile radius are discussed in Section 3.1.3.3.1.

There are no daycare facilities or schools within a 200-foot radius of Hangar 750. The closest school is Sea Park Elementary School, located approximately 3.3 miles south/southeast. The closest daycare is the Child Development Center located on base, approximately 1.7 miles south/southeast of the hangar (EDR, 2015b).

3.1.4 Hangar 751

3.1.4.1 Description and Operational History

Hangar 751 is located in the northern portion of Patrick AFB, south of the intersection of Falcon Avenue and Redstone Road. The hangar is bordered to the north by a parking lot followed by Redstone Road, to the east and west by restricted airfield ramps, and to the south by grassy areas followed by the restricted airfield (Figure 3.2). The geographic coordinates for Hangar 751 are

Hangar 751 was constructed in 1945 and is currently equipped with an AFFF fire suppression system. The hangar is supplied 3% AFFF from a 1,200-gallon AST located in the Building

705 Pump House. The pump house is located approximately 430 feet west/northwest of the hangar (Appendix C, Records of Communication). The Building 705 Pump House is further discussed in Section 3.4.4.

AFFF discharged from the fire suppression system drains into the hangar floor drains, which flow via underground piping to a 30,000-gallon USCT where it is contained. The USCT is located in the grassy area directly south of Hangar 750 (Figure 3.2). According to interviews with personnel familiar with the long-term history of the hangar, the contents of the 30,000-gallon USCT have historically been pumped out to the surrounding ground surface above the USCT or occasionally pumped out by a subcontractor using a vacuum truck and disposed of off base.

Available documents do not indicate a release of AFFF to the environment surrounding Hangar 751. Releases at the 30,000-gallon USCT were discussed in Section 3.1.3. According to the Assistant Fire Chief, there have been no releases of AFFF associated with Hangar 751 (Appendix C, Records of Communication). Therefore, the environmental media surrounding Hangar 751 does not appear to have been impacted by PFCs.

Photographic documentation is provided in Appendix A.

3.1.4.2 Waste Characteristics

Not Applicable.

3.1.4.3 Pathway and Environmental Hazard Assessment

Not Applicable.

3.1.4.3.1 Groundwater Pathway

Not Applicable.

3.1.4.3.2 Surface Water Pathway

Not Applicable.

3.1.4.3.3 Soil and Air Exposure Pathways

Not Applicable.

3.1.5 Hangar 985

3.1.5.1 Description and Operational History

Hangar 985 is located in the east central portion of Patrick AFB. The hangar is bordered to the north by a restricted airfield ramp, to the east by Building 988 followed by South Patrick Drive, to the south by a parking lot, and to the west by Hangar 986 (Figure 3.3). The

geographic coordinates for Hangar 985 are

Hangar 985 was constructed in 1953 and is currently equipped with an AFFF fire suppression system and four low level turrets charged with AFFF. The hangar is supplied 3% AFFF from an 800-gallon AST located in the hangar mechanical room. There is no containment system associated with Hangar 985. According to interviews with personnel familiar with the long-term history of the hangar, there have been no reported or documented releases of AFFF at Hangar 985 (Appendix C, Records of Communication).

There was no available documentation or evidence of an AFFF release to the environment from Hangar 985. Therefore, the environmental media surrounding Hangar 985 is not likely to be impacted by PFCs.

Photographic documentation is provided in Appendix A.

3.1.5.2 Waste Characteristics

Not Applicable.

3.1.5.3 Pathway and Environmental Hazard Assessment

Not Applicable.

3.1.5.3.1 Groundwater Pathway

Not Applicable.

3.1.5.3.2 Surface Water Pathway

Not Applicable.

3.1.5.3.3 Soil and Air Exposure Pathways

Not Applicable.

3.1.6 Hangar 986

3.1.6.1 Description and Operational History

Hangar 986 is located in the east central portion of Patrick AFB (Figure 1.1). The hangar is bordered to the north by a restricted airfield ramp, to the east by Hangar 985, to the south by a parking lot, and to the west by grassy areas and concrete areas with radar systems (Figure 3.3). The geographic coordinates for Hangar 986 are

Hangar 986 was constructed in 1953 and is currently equipped with a deluge fire suppression system. According to the Assistant Fire Chief, the hangar has always been equipped with a deluge fire suppression system and has never been equipped with an AFFF fire suppression system. Interviews with personnel familiar with the long-term history of the hangar indicated that there have been no reported or documented releases of AFFF at Hangar 986 (Appendix C, Records of Communication).

Hangar 986 has never been equipped with an AFFF fire suppression system and there are no reported or documented releases of AFFF. Therefore, the environmental media surrounding Hangar 986 is not likely to have been impacted by PFCs.

Photographic documentation is provided in Appendix A.

3.1.6.2 <u>Waste Characteristics</u>

Not Applicable.

3.1.6.3 Pathway and Environmental Hazard Assessment

Not Applicable.

3.1.6.3.1 Groundwater Pathway

Not Applicable.

3.1.6.3.2 Surface Water Pathway

Not Applicable.

3.1.6.3.3 Soil and Air Exposure Pathways

Not Applicable.

3.2 FIRE STATIONS

3.2.1 Fire Station (Building 810)

3.2.1.1 Description and Operational History

The Patrick AFB Fire Station is identified as Building 810 and is located in the northwestern portion of Patrick AFB. The fire station is bordered to the north by a grassy area followed by a parking lot and the 800 airfield ramp, to the east by a parking lot followed by South Patrick Drive, to the south by a restricted airfield ramp followed by a grassy area with storm sewer open drainage systems, and to the west by restricted airfield and Taxiway A (Figure 3.4). The geographic coordinates for the fire station are

Photographic documentation is provided in Appendix A.

3.2.1.2 Waste Characteristics

The fire station houses two fire engines (Fire Engine 9 and Fire Engine 4), two crash trucks (Crash Truck 6 and Crash Truck 5), and a foam trailer. The two fire engines have a foam capacity of 55 gallons each, Crash Truck 6 and Crash Truck 5 have foam capacities of 210 gallons and 500 gallons respectively, and the foam trailer has a foam capacity of 2,000 gallons. The crash trucks and fire engines are refilled at the fire station with AFFF that is stored at Building 984. According to the Assistant Fire Chief, there have been no releases of AFFF from the vehicles housed at the fire station including during refilling operations.

Time and distance testing is more commonly known as operational checks by the Patrick AFB Fire Department. Operational checks and flushing out of hoses with residual AFFF has been historically performed with the use of AFFF at the 800 airfield ramp and the grassy areas surrounding the fire station. The operational checks are performed daily and flushing out residual AFFF from hoses is conducted after a response requiring the use of AFFF. Releases of AFFF at the 800 ramp area would have likely evaporated or drained into the surrounding grassy areas, where AFFF would have infiltrated the ground surface. The Assistant Fire Chief indicated that during operational checks at the fire station, AFFF was released to the grassy areas northwest of the fire station near Building 804, to the grassy areas would likely have infiltrated the ground surface. The Assistant Fire Chief was not aware of the volume of AFFF released during operational checks and the flushing out of hoses. As of 1999/2000, operational checks are no longer performed with the use of AFFF, but instead use only water at Taxiway Juliet (Appendix C, Records of Communication).

In October 2013, SES performed an investigation to identify potential PFC usage areas and to select locations for further evaluation. The fire station (Building 810) was identified as a location for further investigation based on the storage of AFFF near Building 804 and releases of AFFF to the grassy areas south of the fire station during operational checks. According to the site investigation, 5-gallon containers and 55-gallon drums of AFFF concentrated foaming agent were observed stacked on the grass around the storage building west of the fire station. In addition, the grassy area south of the fire station had reportedly been used as a discharge area for fluids potentially containing AFFF residue from cleaning firefighting equipment (SES, 2014).

On April 30, 2014, one surface soil sample, two subsurface soil samples, and two groundwater samples were collected from the grassy area around the storage building. In addition, one surface soil sample, two subsurface soil samples, and two groundwater samples were collected in the grassy area south of the fire station. Groundwater analytical results indicated that PFOA and PFOS were detected in all four groundwater samples. The PFOA and PFOS detections in all four groundwater samples were reported above the corresponding EPA PHA value of $0.4 \mu g/L$ and $0.2 \mu g/L$, respectively. The soil analytical results for the two surface soil samples and four subsurface soil samples indicated that PFOA and PFOS were detected in all of the soil samples. The PFOA and PFOS detections were reported below their

respective EPA RSSLs of 16,000 μ g/kg and 6,000 μ g/kg for all collected soil samples (SES, 2014).

The SES investigation confirmed the presence of PFC in the environmental media surrounding the fire station (Building 810), specifically the grassy area surrounding Building 804 and the grassy area south of the fire station.

3.2.1.3 Pathway and Environmental Hazard Assessment

A complete exposure pathway typically includes the following components: a source of contamination (an environmental medium contaminated at the source or a release mechanism by which chemicals are released from a source medium and transported), an exposure medium by which a receptor comes into contact, and a route of intake for the contaminant into the receptor's body at the exposure point. If any of these elements are missing, the pathway is incomplete. Other release mechanisms resulting in exposure media for receptors may include the uptake of soil contaminants by plants and animals and the emission of soil contaminants into the air in association with dust particles (EPA, 1989).

3.2.1.3.1 Groundwater Pathway

The basewide geologic and hydrogeologic settings are provided in Section 1.3. According to the SES investigation, groundwater in the general area of the fire station was detected at depths ranging from 3 to 4 feet bgs. Groundwater is assumed to follow the basewide shallow groundwater flow westward toward the Banana River, approximately 4,432 feet west. The SES investigation confirmed the presence of PFCs in the shallow groundwater underlying the areas south of the fire station and surrounding Building 804.

Patrick AFB and surrounding off-base communities receive drinking water from the City of Cocoa, which acquires water from groundwater in the Floridan aquifer, ASR wells, and surface water as discussed in Section 1.3.2. The closest PWS well is located 4.8 miles southwest of the fire station. The PWS well is part of the FL3050985 Palm Shores RV Park well system, a non-community public water system that serves a population of 25 residents (EDR, 2015a).

The combined on- and off-base population within a 4-mile radius from the fire station is approximately 16,357 (EDR, 2015b). The closest residential area is an on-base residential campground located approximately 3,791 feet west/southwest of the fire station.

3.2.1.3.2 Surface Water Pathway

There are no natural drainage features at Patrick AFB as discussed in Section 1.3.3. Surface water runoff at Patrick AFB either infiltrates the ground surface or is controlled by a series of drainage channels, manmade ditches, culverts, and canals. Drainage from the fire station flows west/northwest similar to the shallow groundwater in the vicinity of the fire station and infiltrates the surrounding grassy area or drains into storm sewer open drainage areas that discharge to the grassy area south of the fire station.

The fire station is not located within a flood plain and there are no surface water intakes or downstream fisheries adjacent to the surface water migration path downstream of the fire station (EDR, 2015c).

3.2.1.3.3 Soil and Air Exposure Pathways

The SES investigation confirmed the presence of PFCs in the soil surrounding the fire station. The Patrick AFB fire station is an active station located in the northeastern restricted area of the airfield and is accessible to authorized military personnel, the base fire department, and escorted guests. The number of workers at the fire station varies depending on the work shift. Although the potential exists for soil exposure to workers at the fire station, direct contact by workers with soil is not anticipated. Landscape workers who perform mowing at the area south of the fire station could potentially be exposed to soil through the emission of soil contaminants into the air as dust particles while mowing. There are no residents at the fire station. The closest residential area is 3,791 feet west of the fire station. Population details of the residential areas within a 4-mile radius are discussed in Section 3.2.1.3.1.

There are no daycare facilities or schools within a 200-foot radius of the fire station. The closest school is Sea Park Elementary School, located approximately 2.5 miles south of the fire station. The closest daycare is the Child Development Center located on base, approximately 4,714 feet south/southeast of the fire station (EDR, 2015b).

3.3 EMERGENCY RESPONSE

No emergency response or crash locations were identified at Patrick AFB during this PA.

3.4 OTHER SPILLS AND RELEASES

3.4.1 Fire Truck Rollover Area

3.4.1.1 Description and Operational History

According to the Assistant Fire Chief, a fire truck rolled over in 1997 within the Patrick AFB restricted airfield. The fire truck was turning onto Taxiway B from Taxiway E and rolled over releasing AFFF to the taxiway and surrounding grassy areas. The rollover area is bordered to the north, east, and south by taxiways and to the west by a grassy area (Figure 3.5). The approximate geographic coordinates for the incident are

The rollover area was not accessible during the PA visit and no photographs were taken.

3.4.1.2 Waste Characteristics

In 1997, AFFF was released to the taxiway and most likely drained into the surrounding grassy areas. The AFFF would have evaporated on the taxiway or infiltrated the ground surface in the surrounding grassy areas. The Assistant Fire Chief was not aware of the fire truck foam capacity or the amount of AFFF released (Appendix C, Records of

Communication). The presence of PFCs at the grassy areas surrounding the rollover area is likely.

3.4.1.3 Pathway and Environmental Hazard Assessment

A complete exposure pathway typically includes the following components: a source of contamination (an environmental medium contaminated at the source or a release mechanism by which chemicals are released from a source medium and transported), an exposure medium by which a receptor comes into contact, and a route of intake for the contaminant into the receptor's body at the exposure point. If any of these elements are missing, the pathway is incomplete. Other release mechanisms resulting in exposure media for receptors may include the uptake of soil contaminants by plants and animals and the emission of soil contaminants into the air in association with dust particles (EPA, 1989).

3.4.1.3.1 Groundwater Pathway

The basewide geologic and hydrogeologic settings are provided in Section 1.3. Groundwater at the rollover area is assumed to follow the basewide shallow groundwater flow westward toward the Banana River. The Banana River is located approximately 3,071 feet west of the rollover area. The potential presence of PFCs in groundwater exists based on the potential for AFFF to have infiltrated the ground surface surrounding the rollover area.

Patrick AFB and surrounding off-base communities receive drinking water from the City of Cocoa, which acquires water from groundwater in the Floridan aquifer, ASR wells, and surface water as discussed in Section 1.3.2. The closest PWS well is located 4.5 miles southwest of the rollover area. The PWS well is part of the FL3050985 Palm Shores RV Park well system, a non-community public water system that serves a population of 25 residents (EDR, 2015a).

The combined on- and off-base population within a 4-mile radius from the rollover area is approximately 18,099 (EDR, 2015b). The closest residential area is an on-base residential campground located approximately 2,500 feet west/southwest.

3.4.1.3.2 Surface Water Pathway

There are no natural drainage features at Patrick AFB as discussed in Section 1.3.3. Surface water runoff at Patrick AFB either infiltrates the ground surface or is controlled by a series of drainage channels, manmade ditches, culverts, and canals. Drainage from the rollover area flows to the surrounding grassy areas and infiltrates the ground surface or travels over land surface and discharges to the storm sewer open drainage located approximately 210 feet southeast of the rollover area. Drainage that reaches the storm sewer open drainage will likely infiltrate the ground surface.

The rollover area is not located within a flood plain and there are no surface water intakes or downstream fisheries adjacent to the surface water migration path downstream of the rollover area (EDR, 2015c).

3.4.1.3.3 Soil and Air Exposure Pathways

The fire truck rollover occurred on an active taxiway within the restricted airfield. The rollover area is accessible to authorized military personnel, the base fire department, and escorted guests. There are no workers or residents located at the rollover area. The potential exists for soil exposure to burrowing animals. The closest residential area is 2,500 feet west of the rollover area. Population details of the residential areas within a 4-mile radius are discussed in Section 3.4.1.3.1.

There are no daycare facilities or schools within a 200-foot radius of the rollover area. The closest school is Sea Park Elementary School, located approximately 2.3 miles south of the rollover area. The closest daycare is the Child Development Center located on base, approximately 3,645 feet southeast of the rollover area (EDR, 2015b).

3.4.2 Outfall 21 to Banana River

3.4.2.1 Description and Operational History

Outfall 21 is located along the western boundary of Patrick AFB, approximately 440 feet west/northwest of Hangar 630 (Figure 1.1). The outfall is part of the drainage system at Patrick AFB and is potentially influenced by drainage from Hangars 630 and 647. Historical discharges of AFFF from the fire suppression systems at theses hangars flowed out of the hangars doors to the grassy areas and storm sewer inlets north of the hangars. The storm sewer pipelines release to the environment at the drainage canal through Outfall 21 (Figure 3.6). The geographic coordinates for the outfall are

Photographic documentation is provided in Appendix A.

3.4.2.2 Waste Characteristics

Surface water runoff at Patrick AFB either infiltrates the ground surface or is controlled by a series of drainage channels, manmade ditches, culverts, and canals. These structures form a drainage system at Patrick AFB that moves surface water westward to discharge points along the Banana River (SES, 2014). As discussed in Sections 3.1.1 and 3.1.2, AFFF released from the hangar doors of Hangars 630 and 647 drained into the storm sewer inlets north of the hangar and released to the drainage canal through Outfall 21.

In October 2013, SES performed an investigation to identify potential PFC usage areas and to select locations for further evaluation. Hangar 630 was identified as a location for further investigation based on several accidental releases that occurred from 1998 to 2005. The discharges resulted in AFFF running out the hangar doors to the grassy areas north and south of the hangar. Some fluids reportedly ran across the grass to storm drains leading to the drainage canal.

On April 30, 2014, two surface water and three sediment samples were collected from the drainage canal as part of the investigation (SES, 2014). Analytical results indicated that PFOA and PFOS were detected in both of the surface water samples. The PFOA detections were reported below the EPA PHA value of 0.4 μ g/L and the PFOS detections were reported above the EPA PHA value of 0.2 μ g/L.

The sediment analytical results indicated that PFOA was detected in all three sediment samples and PFOS was detected in two of the three sediment samples. The PFOA and PFOS detections were reported below the respective EPA RSSLs of 16,000 μ g/kg and 6,000 μ g/kg. The SES investigation confirmed the presence of PFCs in the environmental media at the drainage canal.

3.4.2.3 Pathway and Environmental Hazard Assessment

A complete exposure pathway typically includes the following components: a source of contamination (an environmental medium contaminated at the source or a release mechanism by which chemicals are released from a source medium and transported), an exposure medium by which a receptor comes into contact, and a route of intake for the contaminant into the receptor's body at the exposure point. If any of these elements are missing, the pathway is incomplete. Other release mechanisms resulting in exposure media for receptors may include the uptake of soil contaminants by plants and animals and the emission of soil contaminants into the air in association with dust particles (EPA, 1989).

3.4.2.3.1 Groundwater Pathway

The basewide geologic and hydrogeologic settings are provided in Section 1.3. Groundwater flow at the shallow aquifer near the outfall is to the west toward Banana River. The Banana River is located approximately 690 feet west of the outfall release point.

Patrick AFB and surrounding off-base communities receive drinking water from the City of Cocoa, which acquires water from groundwater in the Floridan aquifer, ASR wells, and surface water as discussed in Section 1.3.2. The closest PWS well is located 4.8 miles southwest of the outfall. The PWS well is part of the FL3050985 Palm Shores RV Park well system, a non-community public water system that serves a population of 25 residents (EDR, 2015a).

The combined on- and off-base population within a 4-mile radius from Outfall 21 is approximately 15,414 (EDR, 2015c). The closest residential area is an on-base residential campground located approximately 3,835 feet south of the outfall.

3.4.2.3.2 Surface Water Pathway

There are no natural drainage features at Patrick AFB as discussed in Section 1.3.3. Surface water runoff at Patrick AFB either infiltrates the ground surface or is controlled by a series of drainage channels, manmade ditches, culverts, and canals. The outfall is located in the 100-

year flood zone. The SES investigation confirmed the presence of PFCs in the surface water at the drainage canal.

Surface water released to the drainage canal through Outfall 21 flows 530 feet northwest and is directed west for approximately 220 feet before releasing to the Banana River. The Banana River merges with the Indian River approximately 8.7 miles downstream (south) from the discharge point. The Indian River continues to flow in a northern direction over 15 miles (Geofin, 2015).

The Banana River is identified as a wetland and is classified as E1UBL: (E)-Estuarine, (1)-Subtidal, (UB)-Unconsolidated Bottom, (L)-Subtidal. Multiple wetlands are also identified 15 miles downstream of the outfall along the banks of the Banana River and Indian River (EDR, 2015c). Ingestion of surface water by wildlife at these wetlands is a potential pathway for ecological receptors. These wetlands are identified as an ecologically sensitive environment potentially adjacent to the surface water migration pathway. Additionally, the Banana River and Indian River are known to be used for recreational activities including fishing and boating by residents and nearby communities, providing an exposure pathway to humans through dermal contact and ingestion of fish (Banana River, Florida, n.d.).

There are no other surface water intakes or downstream fisheries adjacent to the surface water migration path downstream of the outfall (EDR, 2015c).

3.4.2.3.3 Soil and Air Exposure Pathway

The SES investigation confirmed the presence of PFCs in the sediment of the drainage canal. The outfall is currently active and discharges to a drainage canal. The well-vegetated location would preclude any fugitive dust emissions and potential exposures. The potential exists for soil exposure to burrowing animals along the banks of the drainage canal. There are no residents or workers at the outfall location. The closest residential area is located approximately 3,835 feet south of the outfall. Population details of the residential areas within a 4-mile radius are discussed in Section 3.4.2.3.1.

There are no daycare facilities or schools within a 200-foot radius of the site. The closest school is Sea Park Elementary School, located approximately 3.1 miles south of Outfall 21. The closest daycare is the Child Development Center located on base, approximately 1.6 miles southeast of the outfall (EDR, 2015b).

3.4.3 Northern Sewage Treatment Plant

3.4.3.1 Description and Operational History

The Northern STP is located on the northwestern portion of Patrick AFB. The STP is bordered to the north by Building 312 and grassy areas, to the east by Buildings 653 and 646, to the south by an unnamed access road and a lightly vegetated area, and to the west by the Banana River (Figure 3.7). The geographic coordinates of the Northern STP are

The Northern STP was constructed in 1968 and served as the main STP for Patrick AFB before being decommissioned in February and March 1995. The STP treated domestic wastewater from the housing area, shops, barracks, mess hall, and office buildings. Effluent from the STP was discharged to the Banana River prior to being decommissioned. Since 1995, all wastewater generated at Patrick AFB is pumped to the City of Cocoa for treatment and disposal (Parsons, 1996).

Photographic documentation is provided in Appendix A.

3.4.3.2 Waste Characteristics

According to the Northern STP Lift Station Operator, AFFF was observed at the STP in 1994. The AFFF was from a hangar fire suppression system discharge that drained into the hangar floor drains and was then directed to the STP. The lift station operator was not aware of which hangar the discharge originated from or the amount of AFFF released. During high winds AFFF was observed to have been blown out to the surrounding areas of the STP (Appendix C, Records of Communication). The presence of PFCs in the surrounding grassy areas at the Northern STP exists.

3.4.3.3 Pathway and Environmental Hazard Assessment

A complete exposure pathway typically includes the following components: a source of contamination (an environmental medium contaminated at the source or a release mechanism by which chemicals are released from a source medium and transported), an exposure medium by which a receptor comes into contact, and a route of intake for the contaminant into the receptor's body at the exposure point. If any of these elements are missing, the pathway is incomplete. Other release mechanisms resulting in exposure media for receptors may include the uptake of soil contaminants by plants and animals and the emission of soil contaminants into the air in association with dust particles (EPA, 1989).

3.4.3.3.1 Groundwater Pathway

The basewide geologic and hydrogeologic settings are provided in Section 1.3. According to historical investigations, groundwater in the vicinity of the Northern STP is encountered at approximately 4 to 7 feet bgs. Groundwater flow in the general area of the Northern STP flows west toward Banana River, located approximately 55 feet to the west. The presence of PFCs in groundwater exists based on the release of AFFF to the surrounding area of the Northern STP.

Patrick AFB and surrounding off-base communities receive drinking water from the City of Cocoa, which acquires water from groundwater in the Floridan aquifer, ASR wells, and surface water as discussed in Section 1.3.2. The closest PWS well is located approximately 5.1 miles southwest of the STP. The PWS well is part of the FL3050985 Palm Shores RV Park well system, a non-community public water system that serves a population of 25 residents (EDR, 2015a).

The combined on- and off-base population within a 4-mile radius from the Northern STP is approximately 15,877 (EDR, 2015c). The closest residential area is located approximately 2,949 feet north of the STP.

3.4.3.3.2 Surface Water Pathway

There are no natural drainage features at Patrick AFB as discussed in Section 1.3.3. Surface water runoff at Patrick AFB either infiltrates the ground surface or is controlled by a series of drainage channels, manmade ditches, culverts, and canals. Drainage from the Northern STP drains to the surrounding grassy areas and infiltrates the ground surface or drains over land surface to the Banana River, approximately 55 feet west of the Northern STP.

The Banana River merges with the Indian River approximately 9.1 miles downstream (south) from the discharge point. The Indian River continues to flow in a northern direction over 15 miles (Geofin, 2015).

The Banana River is identified as a wetland and is classified as E1UBL: (E)-Estuarine, (1)-Subtidal, (UB)-Unconsolidated Bottom, (L)-Subtidal. Multiple wetlands are also identified 15 miles downstream of the outfall along the banks of the Banana River and Indian River (EDR, 2015c). Ingestion of surface water by wildlife at these wetlands is a potential pathway for ecological receptors. These wetlands are identified as an ecologically sensitive environment potentially adjacent to the surface water migration pathway. Additionally, the Banana River and Indian River are known to be used for recreational activities including fishing and boating by residents and nearby communities, providing an exposure pathway to humans through dermal contact and ingestion of fish (Banana River, Florida, n.d.). The STP is located in the 100-year flood plain (EDR, 2015c).

There are no surface water intakes or downstream fisheries adjacent to the surface water migration path downstream of the STP (EDR, 2015c).

3.4.3.3.3 Soil and Air Exposure Pathway

The Northern STP is an inactive plant that is accessible to military personnel and civilians. There are no permanent workers at the STP; however, on occasion, lift station operators and landscapers are present at the STP. Although the potential exists for soil exposure to these workers, it is not anticipated that the workers will be in direct contact with the soil. The potential exists for soil exposure to burrowing animals. The closest residential area is located approximately 2,949 feet north of the STP. Population details of the residential areas within a 4-mile radius are discussed in Section 3.4.3.3.1.

There are no daycare facilities or schools within a 200-foot radius of the site. The closest school is Sea Park Elementary School, located approximately 3.4 miles south/southeast of the STP. The closest daycare is the Child Development Center located on base, approximately 1.8 miles southeast of the STP (EDR, 2015b).

3.4.4 Building 705

3.4.4.1 Description and Operational History

Building 705, **Solution** is located in the northwestern portion of Patrick AFB, southwest of the intersection of Delta Road and Redstone Road (Figure 3.2). The pump house is situated directly northeast of Hangar 750 and houses the pump system designed for mixing AFFF concentrate for distribution to the fire suppression systems at Hangars 750 and 751. The pump system includes a 1,200-gallon AST containing 3% AFFF. According to the Assistant Fire Chief, the underground piping from the pump house to Hangar 750 was capped at an unknown date. Currently the pump house only supplies AFFF to Hangar 751 (Appendix C, Records of Communication). The geographic coordinates for the pump house are

Photographic documentation is provided in Appendix A.

3.4.4.2 Waste Characteristics

According to interviews with base personnel, there have been three releases of AFFF associated with the Building 705 Pump House. These releases include:

- August 2007 there was a leak identified in the underground piping leading to Hangar 751. An unknown amount of AFFF released to the subsurface.
- 2008/2009 there was a leak identified in the underground piping leading to Hangar 750. Approximately 800 gallons of AFFF released to the subsurface.
- 2012 there was an accidental release of AFFF concentrate from the pump system in the pump house. Approximately 5 to 10 gallons released to the floor drains. Drainage into the floor drains would have been pumped to the City of Cocoa's Dyal Water Treatment Plant and released to the Banana River.

The interviewees were not aware of the exact locations of the releases in the underground piping leading to Hangars 750 and 751; however, they did indicate that the piping section where AFFF was released was replaced (Appendix C, Records of Communication).

In October 2013, SES performed an investigation to identify potential PFC usage areas and to select locations for further evaluation. Building 705 was identified as a location for further investigation based on a release of AFFF that occurred in 2011. The release occurred during excavation activities in the general area of the pump house where a contractor accidently broke the underground AFFF pipeline between the pump house and Hangar 751. Approximately 800 to 1,000 gallons of AFFF concentrate was released to the environment. The exact location of the pipeline release was never identified and the entire pipeline was replaced.

On April 30, 2014, four groundwater samples and six soil samples (2 surface and 4 subsurface) were collected from the grassy area surrounding the pump house and the suspected area where the AFFF pipeline had broken. Groundwater analytical results detected PFOA and PFOS in all of the collected groundwater samples. The detected concentrations of PFOA and

Preliminary	Assessment	Report

PFOS were reported above the corresponding EPA PHA values of 0.4 μ g/L and 0.2 μ g/L. Soil analytical results indicated that PFOA and PFOS was detected in five of the six soil samples collected. The detected concentration of PFOA and PFOS were reported below their respective EPA RSSLs of 16,000 μ g/kg and 6,000 μ g/kg (SES, 2014).

The SES investigation confirmed the presence of PFCs in the environmental media surrounding Building 705.

3.4.4.3 Pathway and Environmental Hazard Assessment

A complete exposure pathway typically includes the following components: a source of contamination (an environmental medium contaminated at the source or a release mechanism by which chemicals are released from a source medium and transported), an exposure medium by which a receptor comes into contact, and a route of intake for the contaminant into the receptor's body at the exposure point. If any of these elements are missing, the pathway is incomplete. Other release mechanisms resulting in exposure media for receptors may include the uptake of soil contaminants by plants and animals and the emission of soil contaminants into the air in association with dust particles (EPA, 1989).

3.4.4.3.1 Groundwater Pathway

The basewide geologic and hydrogeologic settings are provided in Section 1.3. Groundwater in the area of the pump house was detected at depths ranging from 4 to 5 feet bgs (SES, 2014). Groundwater flow is assumed to follow the basewide shallow groundwater flow westward toward the Banana River, approximately 1,097 feet west. The SES investigation confirmed the presence of PFCs in the shallow groundwater underlying the area surrounding Building 705 and its associated pipelines.

Patrick AFB and surrounding off-base communities receive drinking water from the City of Cocoa, which acquires water from groundwater in the Floridan aquifer, ASR wells, and surface water as discussed in Section 1.3.2. The closest PWS well is located 5.1 miles southwest of the pump house. The PWS well is part of the FL3050985 Palm Shores RV Park well system, a non-community public water system that serves a population of 25 residents (EDR, 2015a).

The combined on- and off-base population within a 4-mile radius from the pump house is approximately 17,124 (EDR, 2015c). The closest residential area is located approximately 3,070 feet north of the pump house.

3.4.4.3.2 Surface Water Pathway

There are no natural drainage features at Patrick AFB as discussed in Section 1.3.3. Surface water runoff at Patrick AFB either infiltrates the ground surface or is controlled by a series of drainage channels, manmade ditches, culverts, and canals. Drainage from the pump house drains into the surrounding grassy areas and infiltrates the ground surface or into storm sewer inlets that release to the Banana River through storm sewer pipelines. However, discharges of

AFFF from Building 705 occurred in the subsurface and would have likely migrated into the shallow groundwater than follow the surface drainage into the grassy areas or storm sewer inlets.

Building 705 is not located within a flood plain and there are no surface water intakes or downstream fisheries adjacent to the surface water migration path downstream of the pump house (EDR, 2015c).

3.4.4.3.3 Soil and Air Exposure Pathway

The SES investigation confirmed the presence of PFCs in soil at the pump house. Building 705 is an **permanent** that is accessible to military personnel and civilians. There are no permanent workers at the pump house but workers do visit for occasional maintenance on the pump system. Although the potential exists for soil exposure to the occasional workers who visit the pump system, direct contact by workers with soil is not anticipated. There are no residents at the pump house. The closest residential area is 3,070 feet north of Building 705. Population details of the residential areas within a 4-mile radius are discussed in Section 3.4.4.3.1.

There are no daycare facilities or schools within a 200-foot radius of the pump house. The closest school is Sea Park Elementary School, located approximately 3.3 miles south/southeast of the pump house. The closest daycare is the Child Development Center located on base, approximately 1.7 miles south/southeast of the pump house (EDR, 2015b).

3.4.5 Building 984

3.4.5.1 Description and Operational History

Building 984 is located in the east central portion of Patrick AFB (Figure 1.1). The building is located directly east of West Tech Road and is surrounded by parking lots (Figure 3.8). The geographic coordinates for Building 984 are

According to the Assistant Fire Chief, approximately twenty-five 55-gallon drums of 3% AFFF are stored at Building 984. The stored AFFF is used to refill fire engines and crash trucks at the fire station **Example 1** There have been no reported or documented releases of AFFF at the storage area of Building 984. Therefore, the environmental media surrounding Building 984 would not be impacted by PFCs.

Photographic documentation is provided in Appendix A.

3.4.5.2 Waste Characteristics

Not Applicable.

3.4.5.3 Pathway and Environmental Hazard Assessment

Not Applicable.

3.4.5.3.1 Groundwater Pathway

Not Applicable.

3.4.5.3.2 Surface Water Pathway

Not Applicable.

3.4.5.3.3 Soil and Air Exposure Pathways

Not Applicable.

3.4.6 Building 676 –

3.4.6.1 Description and Operational History

Building 676 is located in the northwestern portion of Patrick AFB, east of Outfall 21 (Figure 1.1 and Figure 3.9). The building was initially equipped with a wet pipe fire suppression system. In January 2012, the fire suppression system was retrofitted to include an AFFF fire suppression system. A 200-gallon AST located in the mechanical room supplies the building with 3% AFFF. The geographic coordinates for Building 676 are

According to the Fire Systems POC, there have been no reported or document releases of AFFF at the building. Therefore, the presence of PFCs in the environmental media surrounding Building 676 is not likely.

3.4.6.2 Waste Characteristics

Not Applicable.

3.4.6.3 Pathway and Environmental Hazard Assessment

Not Applicable.

3.4.6.3.1 Groundwater Pathway

Not Applicable.

3.4.6.3.2 Surface Water Pathway

Not Applicable.

3.4.6.3.3 Soil and Air Exposure Pathways

Not Applicable.

This page was intentionally left blank.

FIGURES

This page was intentionally left blank.



Figure 3.1 Hangars 630 and 647 Patrick Air Force Base Brevard County, Florida

Legend

- Monitoring Well
- Storm Sewer Inlet
- Storm Sewer Pipeline
 - Storm Sewer Culvert
 - Storm Sewer Open Drainage
- 624 Building Number



Underground Storage Collection Tank

Inferred Location Boundary

\\gst-srv-01\HGLGIS\PA_Sites\Patrick_AFB\PA_Report\ (3-01)Hangar_630_647.mxd 7/31/2015 SS Source: HGL, Patrick AFB NAIP Online Imagery





Figure 3.2 Hangar 750, Hangar 751, and Building 705-Patrick Air Force Base Brevard County, Florida

Legend

- Monitoring Well
- Storm Sewer Inlet
- Storm Sewer Pipeline
 - Storm Sewer Open Drainage
- 708 Building Number
- Underground Storage Collection Tank Inferred Location Boundary

\\gst-srv-01\HGLGIS\PA_Sites\Patrick_AFB\PA_Report\ (3-02)Hangar_750_751_Building_705_PH.mxd 7/31/2015 SS Source: HGL, Patrick AFB Esri World Imagery





Figure 3.3 Hangar 985 Patrick Air Force Base Brevard County, Florida

Legend

Storm Sewer Inlet

Storm Sewer Pipeline

Storm Sewer Open Drainage

Building Number

Inferred Location Boundary

945

Installation Boundary

\\gst-srv-01\HGLGIS\PA_Sites\Patrick_AFB\PA_Report\ (3-03)Hangar_985.mxd 7/31/2015 SS Source: HGL, Patrick AFB ArcGIS Online Imagery





Figure 3.4 Fire Station (Building 810) Patrick Air Force Base Brevard County, Florida

Legend

Storm Sewer Inlet

Storm Sewer Pipeline

Storm Sewer Open Drainage

Building Number

Inferred Location Boundary

804

Installation Boundary

\\gst-srv-01\HGLGIS\PA_Sites\Patrick_AFB\PA_Report\ (3-04)Fire_Station_(Building_810).mxd 7/31/2015 SS Source: HGL, Patrick AFB ArcGIS Online Imagery





Figure 3.5 Fire Truck Rollover Area Patrick Air Force Base Brevard County, Florida

Legend

	Storm Sewer Inlet
→	Storm Sewer Pipeline
	Storm Sewer Culvert
	Storm Sewer Open Drainage
	Inferred Location Boundary

\\gst-srv-01\HGLGIS\PA_Sites\Patrick_AFB\PA_Report\ (3-05)Fire_Truck_Rollover_Area.mxd 7/31/2015 SS Source: HGL, Patrick AFB ArcGIS Online Imagery





Figure 3.6 Outfall 21 to Banana River Patrick Air Force Base Brevard County, Florida

Legend

- Monitoring Well
- Storm Sewer Inlet
- → Storm Sewer Pipeline
 - Storm Sewer Culvert
 - Storm Sewer Open Drainage
- 688 Building Number
 - Inferred Location Boundary
- Installation Boundary

|\gst-srv-01\HGLGIS\PA_Sites\Patrick_AFB\PA_Report\ (3-06)Outfall_21_to_Banana_River.mxd 7/31/2015 SS Source: HGL, Patrick AFB ArcGIS Online Imagery





Figure 3.7 Northern Sewage Treatment Plant Patrick Air Force Base Brevard County, Florida

Legend

- Monitoring Well
- Storm Sewer Inlet
- Storm Sewer Pipeline
 - Storm Sewer Culvert
 - Storm Sewer Open Drainage
- 653 Building Number
- Inferred Location Boundary
 - Installation Boundary

\\gst-srv-01\HGLGIS\PA_Sites\Patrick_AFB\PA_Report\ (3-07)Northern_Sewage_Treatment_Plant.mxd 7/31/2015 SS Source: HGL, Patrick AFB ArcGIS Online Imagery





Figure 3.8 Building 984 Patrick Air Force Base Brevard County, Florida

Legend

- Storm Sewer Inlet
- → Storm Sewer Pipeline
- Storm Sewer Culvert
- ----- Storm Sewer Open Drainage
- Surface Water

- 984 Building Number
 - Inferred Location Boundary
 - Installation Boundary

\\Gst-srv-01\HGLGIS\PA_Sites\Patrick_AFB\PA_Report\ (3-08)Building_984.mxd 9/17/2015 SS Source: HGL, Patrick AFB ArcGIS Online Imagery





Figure 3.9 Building 676

Patrick Air Force Base Brevard County, Florida

Legend

- Storm Sewer Inlet
- ► Storm Sewer Pipeline
- Storm Sewer Culvert
- ----- Storm Sewer Open Drainage
- Surface Water

- 984 Building Number
 - Inferred Location Boundary
 - Installation Boundary

\\Gst-srv-01\HGLGIS\PA_Sites\Patrick_AFB\PA_Report\ (3-09)Building_676.mxd 9/17/2015 SS Source: HGL, Patrick AFB ArcGIS Online Imagery



This page was intentionally left blank.

4.0 SUMMARY AND CONCLUSIONS

4.1 SUMMARY

4.1.1 Fire Training Areas

4.1.1.1 Fire Training Areas Closed Prior to 1970

FTAs that were closed prior to 1970 did not utilize AFFF and could not have been impacted by PFOA or PFOS from AFFF use. FTA 1 operated from 1950 to 1963, prior to the use of AFFF by the Air Force. Therefore, the environmental media surrounding FTA 1 could not have been impacted by PFCs.

4.1.1.2 Fire Training Areas Operational After 1970

Fire training activities were conducted at Former FTA 2 from 1963 to 1985 at an unlined burn pit associated with the FTA. Although the Assistant Fire Chief was not aware of fire training activities performed at the FTA, the potential exists for AFFF being used to extinguish fires between the years of 1970 through 1978.

The potential presence of PFCs in the environmental media at the burn pit is likely.

4.1.2 Non-Fire Training Areas

4.1.2.1 Hangars

Hangars 630, 647, 750, 751, and 985 are hangars that are currently equipped or have been equipped with AFFF fire suppression systems. There have been multiple releases of AFFF from Hangars 630, 647, 750, and 751 to the grassy areas surrounding the hangars, as discussed in their respective sections above. The SES investigation confirmed the presence of PFCs in the environmental media at Hangars 630, 647, and 750.

Although Hangar 751 is equipped with an AFFF fire suppression system and there have been reported releases of AFFF, a 30,000-gallon USCT located near Hangar 750 contained the releases. There are no reported or documented releases of AFFF to the area surrounding Hangar 751. Therefore, it is not likely that the environmental media surrounding Hangar 751 would have been impacted by PFCs.

Hangar 750 was initially equipped with a wet fire sprinkler system and in 1999 was retrofitted with an AFFF fire suppression system. The hangar operated with an AFFF fire suppression system until 2006 when it was retrofitted with an HEF fire suppression system. A 30,000-gallon USCT contained AFFF releases from the hangar; however, in 2001, an AFFF release was not fully captured by the USCT. Surface flow of AFFF on the ground surface over the USCT area was reported. The SES investigation confirmed the presence of PFCs in the environmental media surrounding the USCT at Hangar 750.

Preliminary Assessment Report

Hangar 985 is equipped with an AFFF fire suppression system; however, according to the Assistant Fire Chief, there have been no reported or documented releases of AFFF at the hangar. Therefore, it is not likely that the environmental media surrounding Hangar 985 would have been impacted by PFCs.

Hangar 986 is equipped with a deluge fire suppression system and has never been equipped with an AFFF fire suppression system. There have been no reported or document releases of AFFF at Hangar 986. Therefore, it is not likely that the environmental media surrounding the hangar would have been impacted by PFCs.

4.1.2.2 Fire Stations

Fire Station (Building 810) is the only fire station located at Patrick AFB. The grassy areas surrounding the fire station (Building 810) were historically used to perform operational checks by the Patrick AFB Fire Department. In addition, residual AFFF was flushed out of hoses at these grassy areas. The SES investigation confirmed the presence of PFCs in the soil and groundwater at the grassy areas north and south of the fire station.

4.1.2.3 Emergency Response

No emergency response or crash locations were identified within Patrick AFB during this PA.

4.1.2.4 Other Spills or Releases

As a result of a fire truck rollover, AFFF was released to the taxiway and most likely drained into the surrounding grassy areas. The AFFF would have evaporated on the taxiway and infiltrated the ground surface in the surrounding grassy areas. The potential presence of PFCs exists in the grassy areas surrounding the rollover area.

Outfall 21 is part of the drainage system at Patrick AFB and receives surface water drainage from Hangars 630 and 647. AFFF released from the hangar doors of Hangar 630 and 647 drained into the storm sewer inlets north of the hangars and released to the drainage canal through Outfall 21. The SES investigation confirmed the presence of PFCs in the surface water and sediment of the drainage canal at Outfall 21.

The Northern STP treated domestic wastewater from the housing area, shops, barracks, mess hall, and office buildings at Patrick AFB. In 1994, AFFF originating from a hangar release drained into the hangar floor drains and was directed to the STP. The STP lift station operator indicated that during high winds AFFF was blown from the STP out to the grassy areas surrounding the STP. The potential exists for the presence of PFCs in the grassy areas surrounding the Northern STP.

Building 705 historically supplied AFFF to Hangar 750 and currently supplies AFFF to Hangar 751. There have been multiple releases of AFFF from the underground pipelines leading to the hangars, including the pump system located in Building 705 as

Preliminary Assessment Report
discussed in Section 3.4.4. The SES investigation confirmed the presence of PFCs in the environmental media (soil and groundwater) surrounding Building 705

Building 984 is used for storage of AFFF. According to the Assistant Fire Chief, approximately twenty-five 55-gallon drums of AFFF are stored in the building. There have been no reported or documented releases of AFFF at the storage area of Building 984. Therefore, it is not likely that the environmental media surrounding the building has been impacted by PFCs.

Building 676 was equipped with an AFFF fire suppression system in January 2012. There are no reported or documented releases of AFFF at the building. Therefore, it is not likely that the environmental media surrounding the building have been impacted by PFCs.

4.2 CONCLUSIONS

Table 4.1 summarizes the findings from this PA report and presents possible future management decisions. The identified locations are categorized by "group" in Table 4.1 as follows:

- Group 1 High mass of AFFF released and probability of groundwater contamination.
- Group 2 Unknown mass or medium mass of AFFF released.
- Group 3 Low mass of AFFF released.
- Group 4 No AFFF released.

Based on the "group" designation and rationale for each location, recommendations are provided in Table 4.1. In accordance with the EPA CERCLA Preliminary Assessment and Site Inspections Guidance documents, each of the identified locations are either recommended for implement removal action due to imminent threat; close out of the location due to no release; initiate a Remedial Investigation; or initiate a Site Inspection.

- Removal action, as defined in CERCLA Section 104, are actions taken to eliminate, control, or otherwise mitigate a threat posed to public health or the environment due to a release or threatened release of hazardous substances (EPA, 1991).
- Close out or no further remedial action planned is defined as a disposition decision that further response under the Federal Superfund is not necessary (EPA, 1991).
- Remedial Investigation is defined as a field investigation to characterize the nature and extent of contamination at a release location. The Remedial Investigation supports development, evaluation, and selection of the appropriate response alternative (EPA, 1991).
- Site Inspection is defined as an investigation to collect and analyze waste and environmental samples to support an evaluation of an identified or potential contaminant release location (EPA, 1992).

Preliminary Assessment Report

Preliminary Assessment Report Summary and Findings					
Locations	Group	Rationale	Recommendation		
Former Fire Training Area 1	4	 Operational period from 1950 to 1963. Operational timeframe predates the use of AFFF by the Air Force in 1970. 			
Former Fire Training Area 2	1	 Operational period from 1963 to 1985. From 1970 to 1978, AFFF may have been used to extinguish fires at the burn pit during fire training activities. Unknown amount of AFFF released at burn pit. 	Initiate a Site Inspection		
Hangar 630	3	 Equipped with an AFFF fire suppression system. Supplied AFFF from 800-gallon AST with 3% AFFF. AFFF released outside of the hangar doors and into the grassy areas north and south of the hangar. Unknown amount of AFFF released to the grassy areas of the hangar. The SES Investigation confirmed the presence of PFCs (PFOA and PFOS) in soil and groundwater at the grassy areas north and south of the hangar. 	Initiate a Site Inspection		
Hangar 647	3	 Equipped with an AFFF fire suppression system. Supplied AFFF from 2,000-gallon AST with 3% AFFF. AFFF released out of the hangar doors and into the grassy area Unknown amount of AFFF released to the grassy areas of the hangar. The SES investigation confirmed the presence of PFCs (PFOA and PFOS) in soil and groundwater at the grassy areas north and south of the hangar. 	Initiate a Site Inspection		
Hangar 750	3	 From 1999 to 2006, the fire suppression system was equipped with AFFF fire suppression system. Supplied AFFF from 1,200-gallon AST with 3% AFFF in Building 705. 2001 AFFF released from hangar fire suppression system and mostly contained by 30,000-gallon USCT. AFFF observed in grassy area surrounding USCT. The SES investigation confirmed the presence of PFCs (PFOA and PFOS) in soil and groundwater at the grassy areas surrounding the USCT. 	Initiate a Site Inspection		
Hangar 751	4	 Equipped with an AFFF fire suppression system. Releases of AFFF contained in 30,000-gallon USCT. 1,200-gallon AST charged with 3% AFFF housed at Building 705 No reported or documented release of AFFF to the environment surrounding the hangar. 	Close-out with no additional investigation.		

HGL-Preliminary Assessment Report-Patrick Air Force Base, Brevard County, Florida

 Table 4.1

 Preliminary Assessment Report Summary and Findings

Preliminary Assessment Report Summary and Findings						
Locations	Group	Rationale	Recommendation			
Hangar 985	4	 Currently equipped with AFFF fire suppression system. Supplied AFFF from 800-gallon AST with 3% AFFF located in mechanical room. No reported or documented releases of AFFF at the hangar. 	Close-out with no additional investigation.			
Hangar 986	4	 Currently equipped with a deluge fire suppression system According to Assistant Fire Chief, hangar has never been equipped with an AFFF fire suppression system. According to multiple interviewees, there have never been any reported or documented releases of AFFF. 	Close-out with no additional investigation.			
Fire Station (Building 810)	2	 Operational checks and flushing out hoses with residual AFFF conducted at the grassy areas east, south and west of the Fire Station. Unknown amount of AFFF released during operational checks and flushing out of hoses. The SES investigation confirmed the presence of PFCs (PFOA and PFOS) in soil and groundwater at the grassy areas where operational checks were performed. 	Initiate a Site Inspection			
Fire Truck Rollover	2	 In1997, a fire truck rolled over on Taxiway B at the Patrick AFB airfield. According to Assistant Fire Chief, AFFF released from the fire truck. Unknown amount of AFFF released to taxiway and potentially flowed into the surrounding grassy areas. 	Initiate a Site Inspection			
Outfall 21 to Banana River	3	 Releases of AFFF through the hangar doors of Hangar 630 and 647 drained into the storm sewer inlets and discharged to drainage canal through Outfall 21. The SES investigation confirmed the presence of PFCs (PFOA and PFOS) in surface water and sediment sampled from drainage canal. 	Initiate a Site Inspection			
North Sewage Treatment Plant	3	 In 1994, AFFF released at an unknown hangar drained into hangar floor drains and drainage was directed to the Northern STP. During high winds, AFFF was observed to have been blown onto the surrounding ground surface at the Northern STP. Unknown amount of AFFF released to the ground surface. 	Initiate a Site Inspection			

	Table 4.1 (Continued)		
eliminary	Assessment Report Summary	and	Findings

Locations	Group	Rationale	Recommendation
Building 705	1	 In August 2007, AFFF released from underground pipeline leading to Hangar 751. Unknown amount of AFFF released. In 2008/2009, AFFF released from underground pipeline leading to Hangar 750. Unknown amount of AFFF released. In, 2011 AFFF released from pipeline leading to Hangar 751 that was accidently broken. Approximately 800 to 1,000 gallons released. The SES investigation confirmed the presence of PFCs (PFOA and PFOS) in groundwater and soils at areas surrounding Building 705. 	Initiate a Site Inspection
Building 984	4	 Stores approximately twenty-five 55-gallons drums of 3% AFFF. No reported or documented releases of AFFF at storage area at Building 984. 	Close-out of building with no additional investigation.
Building 676 –	4	 In January 2012, building retrofitted from wet pipe fire suppression system to AFFF fire suppression system. Supplied AFFF from 200-gallon AST with 3% AFFF located in mechanical room. No reported or documented releases of AFFF at storage area at Building 676 	Close-out with no additional investigation.

Table 4.1 (Continued)Preliminary Assessment Report Summary and Findings

5.0 **REFERENCES**

- Banana River, Florida, n.d. About the Banana River Lagoon. Retrieved from <u>http://banana-river/banana-river/</u>.
- Chang, E.T., et al. 2014. A Critical Review of Perfluoroocanoate and Perfluorooctanesulfonate Exposure and Cancer Risk in Humans, Crit Rev Toxicol, 44(S1): 1-81.
- Coates, 1977. A History of USAF Fire Protection Training at Chanute Air Force Base 1964-1976. Coates, C.Y. (Center Historian), Chanute Technical Training Center, Chanute AFB, Illinois, February.
- EDR, 2015a. EDR Geocheck Report 4314523.1s, June.
- EDR, 2015b. EDR Off-Site Receptor Report 4314523.3s, June.
- EDR, 2015c. EDR NEPACheck 4314523.2s, June.
- Environmental Science and Engineering, Inc., 1988. Installation Restoration Program, Phase II-Confirmation/Quantification, Stage I, Final Report. October.
- EPA, 1989. Risk Assessment Guidance for Superfund Volume I: Human Health Evaluation Manual (Part A). Interim Final. USEPA/540/1- 89/002 December. Office of Emergency and Remedial Response U.S. Environmental Protection Agency, Washington, D.C.
- EPA, 1991. EPA Guidance for Performing Preliminary Assessments under CERCLA. September.
- EPA, 1992. EPA Guidance for Performing Site Inspections under CERCLA September. GeoFin, 2015. U.S. Fish and Wildlife Service, Geospatial Fisheries Information System, Retrieved from http://ecos.fws.gov/geofin/.
- GeoFin, 2015. U.S. Fish and Wildlife Service. Geospatial Fisheries Information System. Retrieved from http://ecos.fws.gov/geofin/.
- O'Brien & Grer Engineers, Inc., 1993. Final Report, Installation Restoration Program (IRP), FT-21 (FTA-1), Statement of Basis (Decision Document). July.
- Parsons Engineering Science, Inc., 1995. Final Intrinsic Remediation Treatability Study for Site ST-29 Patrick Air Force Base, Florida. August.
- Parsons Engineering Science, Inc., 1996. Installation Restoration Program Contamination Assessment Report Base Sewage Treatment Plant Facility 650 (ST-11), Patrick Air Force Base, Florida. January.

- Patrick AFB, 2013. 2013 Annual Drinking Water Quality Report for Patrick Air Force Base (AFB), PWS-ID:3054128).
- Porter, 2011. AFCEE/TDV Emerging Issues, Perfluorinated Compounds, Air Force Restoration and Technology Transfer Workshop, March.
- Rak et al., 2009. Increasing Regulation of Perfluorinated Compounds and the Potential Impacts at Air Force Installations, Andrew and Catherine M. Vogel, prepared for the U.S. Air Force, March.
- SES Construction and Fuel Services LLC (SES), 2014. Final Site Investigation Report for Site Investigations of Fire Fighting Foam Usage at Various Air Force Bases in the United States for Patrick Air Force Base, Brevard County, Florida. November.
- URS, 2005. Fact Sheet for: Firefighter Training Area No. 2, SWMU No. 033, Installation Restoration Program FT002, Patrick AFB, Florida. October.

APPENDIX A

PHOTO DOCUMENTATION

This page was intentionally left blank.



Photo 1: View of Building 676

facing west.



Photo 2: View of AFFF piping at the Building 676.



Photo 3: View of 200 gallon AST charge with 3% AFFF.



Photo 4: View of AFFF label on the AST.



Photo 1: View of Building 705.



Photo 2: View of 1,200 gallon AFFF tank at Building 705 that supplies Hangar 751.



Photo 3: View of AFFF label on AFFF AST at Building 705.



Photo 4: View of AFFF piping system at Building 705 that supplies Hangar 750.



Photo 1: View of AFFF stored at Building 984.



Photo 1: View of Current Fire Training Area.



Photo 1: View of Fire Station



Photo 2: View of AFFF trailer located at the Fire Station.



Photo 1: View of former burn pit area at the Former Fire Training Area 2



Photo1: View of Hangar 630.



Photo 2: View of AFFF 800 gallon AST at Hangar 630



Photo 3: View of AFFF label on AST at Hangar 630.



Photo 4: View of AFFF pipelines in Hangar 630



Photo 5: View of AFFF pipeline in Hangar 630.



Photo 1: View of Hangar 647.



Photo 2: View of AFFF pipelines at Hangar 647.



Photo 3: View of 2,000 gallon AFFF AST at Hangar 647.



Photo 4: View of AFFF AST label at Hangar 647.



Photo 1: View of Hangar 750.



Photo 1: View of Hangar 751.



Photo 2: View of AFFF pipelines at Hangar 751.



Photo 1: View of Hangar 985.



Photo 2: View of low level AFFF turret at Hangar 985.



Photo 3: View of low level AFFF turret at Hangar 985.



Photo 4: View of AFFF fire suppression piping at Hangar 985.



Photo 5: View of 800 gallon AFFF AST

Hangar 985.



Photo 6: View of AFFF pump and pump controller at Hangar 985.



Photo 1: View of Hangar 986.



Photo 1: View of Northern Sewage Treatment Plant facing southwest.



Photo 2: View of Northern Sewage Treatment Plant facing west.



Photo 1: View of Outfall 21 facing northeast.

This page was intentionally left blank.

APPENDIX B

FIELD DOCUMENTATION

This page was intentionally left blank.

					Identification		
Potential Haza	ardous Waste	e Site Prel	iminary As	sessment Form	State:	CERCLIS #:	
			•		CERCLIS Discovery D	ate:	
		1. (General Site Info	rmation			
Name: Building 676		Street Address	: NA				
City:		State: FL	Zip Code:32925	County:Brevard	Co. Code:12009	Cong. Dist:8th	
Latitude:	Longitude:	Approximate A	Area of Site:	Status of Site:			
		Acr	es	Active Not Specif	fied		
			Square Ft	Inactive NA (GW pl	ume, etc.)		
Site Name: Building 676							
Site Description: Buildin	g 676 is located in the	e northwestern	portion of Patrick	AFB. The building is a fue	l truck maintenance f	acility that was	
initially equipped with a suppression system and	wet pipe fire suppre a 200 gallon AST loca	ssion system. Ir ated in the mec	h January 2012, the hanical room supp	e fire suppression system lies the building with 3%	was retrofitted to an AFFF.	AFFF fire	
				J. J			
		2. Ow	/ner/Operator In	formation			
Owner: Patrick AFB			Operator: Same a	as "owner"			
Street Address:			Street Address:				
City:		-	City:				
State: FL	Zip Code:32925	Telephone:	State:	Zip Code:	Telephone:		
Type of Ownership:			Type of Ownership:				
Private	County		Private	County			
✓ Federal Agency	Municipal	od	Federal Agency Municipal Name: Not Specified				
State	☐ Not specifi	eu	□ State □ Other				
Indian			Indian				
		3. S	ite Evaluator Info	ormation			
Name of Evaluator: Johr	n Sandoval	Agency/Organi	ization: HydroGeo	Logic, Inc.	Date Prepared:07/06/15		
Street Address:404 E. Ra	amsey Road, Ste. 210		City:San Antonio		State:Texas		
Name of EPA or State Agency Contact:NA			Street Address:		1		
City: State			Telephone:				
		4. Site D	Disposition <i>(for E</i>	PA use only)	1		
Emergency Response/Removal Assessment			CERCLIS Recommendation:		Signature:		
Recommendation:			Li Higner Priority SI		Name (typed):		
					Desition		
Date:			Other:				
			Date:		1		

5. General Site Characteristics - NA						
Predominant Land Use W	ithin 1 Mile of	Site (check all that	Site Setting:		Years of Operation:	
apply):	Agriculture	DOI	🗌 Urbai	ı	Beginning Year	
Commercial Residential Forest/Fields	Mining DOD DOF	Other Federal	Subu	rban	Ending Year	
		Other			Unknown	
Type of Site Operations (c	heck all that a	apply):			Waste Generated:	
Manufacturing (must check subcategory) Lumber and Wood Products Inorganic Chemicals			 Retail Recycling Junk/Salvage Yard Municipal Landfill Other Landfill DOD DOE DOI Other Federal Facility RCRA 		 Onsite Offsite Onsite and Offsite 	
 Plastic and/or Rubber Products Paints, Varnishes Industrial Organic Chemicals Agricultural Chemicals Miscellaneous Chemical Products Primary Metals Metal Coating, Plating, Engraving 					Waste Deposition Authorized By: Present Owner Former Owner Present & Former Owner Unauthorized Unknown 	
 Metal Forging, Stamping Fabricated Structural Metal Products Electronic Equipment Other Manufacturing Mining Metals Coal Oil and Gas Non-metallic Minerals 		 Treatment, Storage, or Disposal Large Quantity Generator Small Quantity Generator Subtitle D Municipal Industrial "Converter" "Protective Filer" "Non-or Late Filer" Note Specified Other 		Waste Accessible to the Public:		
6. Waste Characteristic	cs Informatio	on - NA			(Refer to PA	
			Table 1 for WC So	ore)		
Source Type: (check all that apply)	Source (include u	e Waste Quantity: ^{nit)}	Tier*:	General Type of Waste apply):	(check all that	
Landfill Surface Impoundment Drums Tanks and Non-Dum Contain Chemical Waste Pile Scrap Metal or Junk Pile Tailings Pile Trash Pile (open drum) Land Treatment Contaminated GW/ Plume	ers			Metals Organics Inorganics Solvents Paints/Pigments Laboratory/Hospital Waste Radioactive Waste Construction/Demolition V	Pesticides/Herbicides Acids/Bases Oily Waste Municipal Waste Mining Waste Explosives Other Vaste Acids/Bases Acids/Ba	
(unidentified source)					as Deposited (check all that apply):	
Contaminated Soil Other No Sources *C=Constituent W-V		blume A=Area		Situige Powder Liquid Gas		
	vastestream, v=vt					

7. Ground Water Pathway - NA						
Is Ground Water Used for Drinking Within 4	Is There a Suspected Release to	List Secondary Target Population Served by Ground Water				
Miles:	Ground Water ¹ :	Withdrawn From:				
□ Yes	Ves					
		0 - 1/4 Mile				
If Yes, Distance to nearest Drinking		>1/4 1/2 Mile				
Well:	Have Primary Target Drinking	>1/4 - 1/2 Mile				
Feet	Water Wells Been Identified:	>1/2 - 1 Mile				
Type of Drinking Water Wells Within 4						
Miles		>1 - 2 Mile				
(check all that apply):						
	If Yes, Enter Primary Target	>2 - 3 Mile				
	Population:					
None None	People ³	>3 - 4 Mile				
Depth to Shallowest Aquifer:	Nearest Designated Wellhead					
Feet	Protection Area ⁶	Total Within 4 Miles ⁴				
Karst Terrain/Aquifer Present:						
☐ Yes	\square None Within 4 Miles	*Use population #s for PA Table 2				
		[↑] Note nearest well for #5 on GW Pathway Scoresheet				
	8. Surface Water Patl	hway - NA				
Type of Surface Water Draining Site and 15 N	Ailes Downstream (check all that	Shortest Overland Distance From Any Source to Surface Water:				
		E				
Stream River Por	nd 🗌 Lake	Feel				
Bay Ocean Oth	ner	Miles				
Is There a Suspected Release to Surface Wat	er ¹ :	Site is Located in:				
		🗌 Annual - 10 yr Floodplain				
🗌 Yes		\square >10yr - 100yr Floodplain				
🗌 No		>100yr - 500yr Floodplain				
		>500yr Floodplain				
Drinking Water Intake Located Along the Sur	face Water Migration Path:	List All Secondary Target Drinking Water Intakes:				
Yes						
□ No		Name: Water Body: Flow (cfs): Population Served:				
Have Primary Target Drinking Water Intakes	Been Identified:					
Yes If Vac Distant	ce to Nearest Drinking					
□ No Water Intake	: Miles ⁶					
If Yes, Enter Population Served by Target Inta	ake:					
De estad		Total within 15 Miles ⁴				
People ⁴						
Fisheries Located Along the Surface Water N	ligration Path:	List All Secondary Target Fisheries ¹⁰ :				
□ Yes □ No If Yes, Distanc	e to Nearest Fishery: Miles	<u>Water Body/ Fishery Name</u> : <u>Flow (cfs)</u> :				
Have Primary Target Fisheries Been Identifie	(viies	· · · · · · · · · · · · · · · · · · ·				
∐ Yes ∐ No						

8. Surface Water Pathway (continued)						
Wetlands Located Along the Surface Water M	igration Path:	Other Sensitive Environments Located Along the Surface Water Migration Path:				
☐ Yes ☐ No	 ☐ Yes ☐ Yes ☐ No ☐ Miles 					
Have Primary Target Wetlands Been Identifie	ed:	Have Primary Ta	arget Sensitive	e Environments Been Identified:		
☐ Yes ☐ No	☐ Yes ☐ No					
List All Wetlands:		List All Sensitive Environments ¹¹ :				
Water Body : Flow (cfs): Frontage miles:		<u>Water Body</u> :	<u>Flow (cfs)</u> :	Sensitive Environment Type:		
	9.50	il Exposure Path				
Are People Occupying Residence or	Number of Wo	rkers Onsite ⁴	Have Terres	trial Sensitive Environments Been Identified on or		
Attending School or Daycare on or Within 200 Feet of Area of Known or Suspected Contamination:	e 00 - 1,000 000	Within 200 F Contaminati	Feet of Areas of Known or Suspected fon:			
Ves				□ No		
			If Yes, List Each Terrestrial Sensitive Environm			
If Yes, Enter Total Residential Population: People ²	eople ⁷ *Refer to PA Table 7 for environment types		able 7 for environment types			
		10. Air Pathway	- NA			
Is there a Suspected Release to Air ¹ :		Wetlands Located Within 4 Miles of the Site ⁶ :				
Enter Total Population on or Within:	☐ Yes ☐ No	If Yes, How	Many Acres: Acres			
Onsite		Other Sensitive Environments Located Within 4 Miles of the Site:				
0-1/4 Mile	☐ Yes ☐ No					
>1/4-1/2 Mile	List All Sensitive Environments Within 1/2 Mile of the Site ⁶ :					
>1/2-1 Mile	Distance: Sensitive Environment Type/Wetlands Area (acres):					
>1-2 Miles	Onsite					
>2-3 Miles	0-1/4 Mile					
>3-4 Miles	>1/4-1/2 Mile					
Total Within 4 Miles ³⁻⁵	*Refer to PA Table 10	for calculations or	n air pathway exposures			

¹⁻¹¹ Refers to question number on the PA scoresheet for each particular pathway
					Identification		
Potential Haza	ardous Wast	e Site Pre	liminary Ass	essment Form	State:	CERCLIS #:	
					CERCLIS Discovery	Date:	
		1.	General Site Infor	mation			
Name: Building 705		Street Address	: NA				
City:		State: FL	Zip Code:32925	County:Brevard	Co. Code:12009	Cong. Dist:8th	
Latitude:	Longitude:	Approximate A	rea of Site:	Status of Site:			
		Acres		Active 🗌 Not Specif	ied		
			Square Ft	Inactive NA (GW pl	lume, etc.)		
Site Name: Building 705	Pump House	•					
Site Description: Buildir	ng 705 Pump House i	s located the no	rthwestern portion	of Patrick AFB, southwes	t of intersection of D	elta Road and	
Redstone Road. The pur	np house is situated	directly northea	st of Hangar 750 an	d houses the pump syste	m for mixing AFFF co	ncentrate for	
distribution to the fire s	uppression systems	to Hangar 750 a	nd 751. The pump s	ystem contained a 1,200	gallon poly tank that	supplied 3% AFFF	
to Hangars 750 and 751	. According to the As	sistant Fire Chie	f, the underground	piping from the pump ho	use to Hangar 750 w	as capped off at an	
unknown date. Currentl	y the pump house of	nly supplies AFFI	to Hangar 751.				
		2. 0\	wner/Operator In	ormation			
Owner: Patrick AFB			Operator: Same as	"owner"			
Street Address:			Street Address:				
City:			City:				
State: FL	Zip Code:32925	Telephone:	State:	Zip Code:	Telephone:		
Type of Ownership:			Type of Ownership	:			
Private	County		Private	County			
✓ Federal Agency	Municipal		Federal Agency	Municipal			
Name: <u>DOD</u>	🗌 Not Specif	ied	Name: Not Specified				
	Other		Conter Other				
		3. 9	ite Evaluator Info	rmation			
Name of Evaluator: Johr	n Sandoval	Agency/Organi	ization: HydroGeoLogic, Inc.		Date Prepared:07/06/15		
Street Address:404 E. Ra	amsey Road, Ste. 210)	City:San Antonio		State:Texas		
Name of FPA or State A	gency Contact:NA		Street Address:		I		
	Seriey contact. WY		Street Address.				
City:		State:		Telephone:			
		4. Site	Disposition (for El	PA use only)			
Emergency Response/Re	emoval Assessment		CERCLIS Recomme	ndation:	Signature:		
Recommendation:			Higher Priority S	SI			
	Yes		Lower Priority S	I	Name (typed):		
	🗌 No				Position		
Date:			Other:	_			
			Date:	-			

	5. (General Site Chara	cteristics	
Predominant Land Use Wit	hin 1 Mile of Site (check all that	Site Setting:		Years of Operation:
apply):		_		
		🗌 Urban		Beginning Yearunknown
			an	
Residential	\Box DOD \Box Facility:			Ending Yearpresent
Forest/Fields	DOE			
	Other			✓ Unknown
Type of Site Operations (ch	eck all that apply):			Waste Generated:
Manufacturing (must check su	bcategory)	Retail		✓ Onsite
Lumber and Wood Produ	ucts	Recycling		Offsite
Inorganic Chemicals		🗌 Junk/Salvage Yard		Onsite and Offsite
Plastic and/or Rubber Pr	oducts	Municipal Landfill		
Paints, Varnishes		Other Landfill		Waste Deposition Authorized By:
Industrial Organic Chem	icals			✓ Present Owner
Agricultural Chemicals				Former Owner
Miscellaneous Chemical	Products	Other Federal Facili	tv	Present & Former Owner
Metal Coating Plating E	ngraving			
Metal Forging, Stamping	ingraving	Treatment, Stora	age, or Disposal	
☐ Fabricated Structural Me	tal Products	Large Quantity C	Generator	Waste Accessible to the Public:
Electronic Equipment		Small Quantity G	Senerator	
Other Manufacturing		Subtitle D		√ Yes
		🗌 Municipal		□ No
		Industrial		
		Converter"		Distance to Nearest Dwelling School
		"Protective Filer"	-	or Workplace
Non-metallic Minerals				
				<u>100</u> Feet
6. Waste Characteristics	s Information			(Refer to PA Table 1
Source Type:	Source Waste Quantity:	tor WC Score)	General Type of Waste	(check all that
(shock all that applied			apply:	
(check all that apply)	(Include unit)		appiy).	
			Metals	Pesticides/Herbicides
Surface Impoundment			☐ Organics	Acids/Bases
				Oily Waste
Tanks and Non-Dum Containe	rs		Solvenis Paints/Pigments	
Chemical Waste Pile			Laboratory/Hospital Waste	
Scrap Metal or Junk Pile			Radioactive Waste	I Other AFFF
Tailings Pile			Construction/Demolition V	Vaste
☐ Trash Pile (open drum)				
Land Treatment				
			Physical State of Waste	as Deposited (check all that apply):
(unidentified source)				
(unidentified source)				
Contaminated Soil				
✓ OtherAFFF_				
*C=Constituent, W=W	astestream, V=Volume, A=Area			

	7. Ground Water Pa	athway
Is Ground Water Used for Drinking Within 4	Is There a Suspected Release to	List Secondary Target Population Served by Ground Water
Miles:	Ground Water ¹ :	Withdrawn From:
Voc.		
No		
		0 - 1/4 Mile
If Yes, Distance to nearest Drinking		
Well:	Have Primary Target Drinking	>1/4 - 1/2 Mile
_5.1 miles	Water Wells Reen Identified:	
	Water Wens been dentined.	>1/2 - 1 Mile
Type of Drinking Water Wells Within 4		
Willes	l ✓ Yes	>1 - 2 Mile
(check all that apply):	🗌 No	
✓ Municipal	If Vac Enter Drimon, Target	>2 - 3 Mile
	Deputation:	>2 - 4 Mile
	25 People ³	23 - 4 Wile
		Total Within 4 Miles ⁴ 17 124
Depth to Shallowest Aquifer:	Nearest Designated Wellhead	
_ <u>4-5</u> Feet	Protection Area ⁶ :	
Karst Terrain/Aquifer Present:	Underlies Site	*Use population #s for PA Table 2
	── >0-4 Miles	*Note nearest well for #5 on GW Pathway Scoresheet
☑ Yes	None Within 4 Miles	
No		
	8. Surface Water Pa	athway
Type of Surface Water Draining Site and 15 N	/liles Downstream (check all that	Shortest Overland Distance From Any Source to Surface Water:
apply):		
Stream River Por	nd 🗌 Lake	1,097 Feet
Bay Ocean Oth	er	Miles
	-	
Is There a Suspected Release to Surface Wate	er ¹ :	Site is Located in:
		Annual - 10 yr Floodplain
Yes		>10yr - 100yr Floodplain
		$\square > 100 \text{yr} - 500 \text{yr}$ Floodplain $\square > 500 \text{yr}$ Floodplain
Drinking Water Intake Located Along the Sur	face Water Migration Path:	List All Secondary Target Drinking Water Intakes:
✓ No		Name: Water Body: Flow (cfs): Population Served:
Have Primary Target Drinking Water Intakes	Been Identified:	
Yes If Yes, Distance	ce to Nearest Drinking	
☑ No Water Intake	: Miles ⁶	
If Yes, Enter Population Served by Target Inta	аке:	
People ⁴		Total within 15 Miles ⁴
Fisheries Located Along the Surface Water M	ligration Path:	List All Secondary Target Eicherias ¹⁰
If Vec Distance	e to Nearest Fishery	Vister Body/ Sistem Name
Yes ⊻ No	Miles	water bouy/ Fishery Name : Flow (CIS):
Have Primary Target Fisheries Reen Identified		1
	-	
L Yes 🗹 No		

	8. Surfac	ce Water Pathwa	y (continue	d)	
Wetlands Located Along the Surface Water I	Vigration Path:	Other Sensitive Environments Located Along the Surface Water Migration Path:			
☐ Yes ✓ No		☐ Yes If Yes, Distance to Nearest Sensitive Environment: ✓ No feet			
Have Primary Target Wetlands Been Identif	ied:	Have Primary Tar	get Sensitive	e Environments Been Identified:	
☐ Yes ✓ No		☐ Yes ✓ No			
List All Wetlands:		List All Sensitive I	Environments	5 ¹¹ :	
Water Body : Flow (cfs): Frontage miles:		<u>Water Body</u> :	Flow (cfs):	Sensitive Environment Type:	
	9.	. Soil Exposure Pa	thway		
Are People Occupying Residence or Attending School or Daycare on or Within 200 Feet of Area of Known or Suspected Contamination:Number of Wo □ □ □ □ □ □ 		rkers Onsite ⁴ : 00 - 1,000 000	Have Terre Within 200 Contamina	estrial Sensitive Environments Beer) Feet of Areas of Known or Suspec htion:	ı Identified on or ted
☐ Yes ✓ No		☑ № If Yes, List Each Terrestrial Sensitive Enviror		ment⁵:	
If Yes, Enter Total Residential Population: People ² Population: People ⁷		hin 1 Mile: <u>miles)</u>	*Refer to PA Table 7 for environment types		
		10. Air Pathwa	ay		
Is there a Suspected Release to Air ¹ :		Wetlands Located Within 4 Miles of the Site ⁶ :			
☐ Yes ☑ No Enter Total Population on or Within:		✓ Yes No If Yes, How Many Acres: <u>unknown</u> Acres			5
Onsite		Other Sensitive En	vironments	Located Within 4 Miles of the Site:	
0-1/4 Mile		✓ Yes □ No			
>1/4-1/2 Mile		List All Sensitive E	nvironments	Within 1/2 Mile of the Site ⁶ :	
>1/2-1 Mile		Distance: Sens	itive Environr	nent Type/Wetlands Area (acres):	
>1-2 Miles		Onsite			
>2-3 Miles		0-1/4 Mile <u>Ba</u>	nana River		
>3-4 Miles		>1/4-1/2 Mile			
Total Within 4 Miles ³⁻⁵ <u>17,124</u>		*Refer to PA Table 10 f	or calculations or	n air pathway exposures	

					Identification	
Potential Haz	Potential Hazardous Waste Site Preliminary Assessment Form					CERCLIS #:
					CERCLIS Discovery Da	ate:
		1. (General Site Info	rmation		
Name: Building 984		Street Address	: NA			
City: State		State: FL	Zip Code:32925	County:Brevard	Co. Code:12009	Cong. Dist:8th
Latitude:	titude: Longitude: Approximate Area of Site: Status of Site:		Status of Site:			
		Acr	es	Active 🗌 Not Specifi	ied	
			Square Ft	🗌 Inactive 🛛 🗌 NA (GW pl	ume, etc.)	
Site Name: Building 98	4					
Site Description: Building	ng 984 is located in th	e eastern centra	al portion of Patric	k AFB. approximately 25 S	55-gallon drums of 3%	AFFF are stored
at Building 984. The ex	cess AFFF is used to re	efill fire engines	and crash trucks a	t the fire station (Building	810). There have bee	n no reported or
documented releases of	of AFFF at the storage	area of Building	984.			
		2. Ow	/ner/Operator In	formation		
Owner: Patrick AFB		2.01	Operator: Same a	as "owner"		
Street Address:			Street Address:			
City:			City:			
State: FL	Zip Code:32925	Telephone:	State:	Zip Code:	Telephone:	
Type of Ownership:			Type of Ownersh	ip:		
☐ Private				County		
✓ Federal Agency			Finale Finale	Municipal		
Name: <u>DOD</u>	Not Specif	ied	Name:	Not Specified		
	Other			Other		
		3. Si	ite Evaluator Info	ormation		
Name of Evaluator: Joh	ın Sandoval	Agency/Organi	ization: HydroGeoLogic, Inc.		Date Prepared:07/06/15	
Street Address:404 E. F	Ramsey Road, Ste. 210)	City:San Antonio		State:Texas	
Name of EPA or State A	Agency Contact:NA		Street Address:			
City		Stato		Tolonhono:		
City: State:		State.		relephone.		
		4. Site D	Disposition (for E	PA use only)		
Emergency Response/F	Removal Assessment		CERCLIS Recomm	endation:	Signature:	
Recommendation:			Higher Priority	y SI I SI	Name (typed):	
	Yes			51	ivanie (typeu).	
	∟ No				Position:	
Date:			Date:			

5. General Site Characteristics - NA					
Predominant Land Use W	ithin 1 Mile of	f Site (check all that	Site Setting:		Years of Operation:
apply):	Agriculture	🗌 DOI	🗌 Urbar	ı	Beginning Year
Commercial Residential Forest/Fields	Mining DOD DOF	Other Federal	Subu	rban	Ending Year
		Other			Unknown
Type of Site Operations (c	heck all that a	apply):			Waste Generated:
Manufacturing (must check subcategory) Lumber and Wood Products Inorganic Chemicals Resting and/or Pubber Products			 □ Retail □ Recycling □ Junk/Salvage Yan □ Municipal Landfill 	d	 Onsite Offsite Onsite and Offsite
 Paints, Varnishes Industrial Organic Cher Agricultural Chemicals Miscellaneous Chemica Primary Metals Metal Coating, Plating, 	nicals I Products Engraving		Other Landfill ODD DOE DOI Other Federal Fac	ility	Waste Deposition Authorized By: Present Owner Former Owner Present & Former Owner Unauthorized Unknown
 Metal Forging, Stampin Fabricated Structural M Electronic Equipment Other Manufacturing Mining Metals Coal Oil and Gas Non-metallic Minerals 	ig letal Products		☐ Treatment, Sto ☐ Large Quantity ☐ Small Quantity ☐ Subtitle D ☐ Municipa ☐ Industria ☐ "Converter" ☐ "Protective File ☐ "Non-or Late F ☐ Note Specified ☐ Other	orage, or Disposal generator Generator I I er" Filer"	Waste Accessible to the Public:
6. Waste Characteristic	cs Informatio	on - NA			(Refer to PA
			Table 1 for WC So	core)	
Source Type: (check all that apply)	Source (include u	e Waste Quantity:	Tier*:	General Type of Waste apply):	(check all that
Landfill Surface Impoundment Drums Tanks and Non-Dum Contain Chemical Waste Pile Scrap Metal or Junk Pile Tailings Pile Trash Pile (open drum) Land Treatment Contaminated GW/ Plume	ers			Metals Organics Inorganics Solvents Paints/Pigments Laboratory/Hospital Waste Radioactive Waste Construction/Demolition V	
(unidentified source) Contaminated SW/Sediment (unidentified source)					as Deposited (check all that apply):
Contaminated Soil Other No Sources				Situge Powder Liquid Gas	
*C=Constituent, W=\	vastestream, V=Vo	oiume, A=Area			

7. Ground Water Pathway - NA					
Is Ground Water Used for Drinking Within 4	Is There a Suspected Release to	List Secondary Target Population Served by Ground Water			
Miles:	Ground Water ¹ :	Withdrawn From:			
□ Yes	Ves				
		0 - 1/4 Mile			
If Yes, Distance to nearest Drinking		>1/4 1/2 Mile			
Well:	Have Primary Target Drinking	>1/4 - 1/2 Mile			
Feet	Water Wells Been Identified:	>1/2 - 1 Mile			
Type of Drinking Water Wells Within 4					
Miles		>1 - 2 Mile			
(check all that apply):					
	If Yes, Enter Primary Target	>2 - 3 Mile			
Private	Population:				
None None	People ³	>3 - 4 Mile			
Depth to Shallowest Aquifer:	Nearest Designated Wellhead				
Feet	Protection Area ⁶	Total Within 4 Miles ⁴			
Karst Terrain/Aquifer Present:					
☐ Yes	\square None Within 4 Miles	*Use population #s for PA Table 2			
□ No		[↑] Note nearest well for #5 on GW Pathway Scoresheet			
	8. Surface Water Patl	hway - NA			
Type of Surface Water Draining Site and 15 N	Ailes Downstream (check all that	Shortest Overland Distance From Any Source to Surface Water:			
		E			
Stream River Por	nd 🗌 Lake	Feet			
Bay Ocean Oth	ner	Miles			
Is There a Suspected Release to Surface Wat	er ¹ :	Site is Located in:			
		🗌 Annual - 10 yr Floodplain			
Yes		\square >10yr - 100yr Floodplain			
🗌 No		>100yr - 500yr Floodplain			
		>500yr Floodplain			
Drinking Water Intake Located Along the Sur	face Water Migration Path:	List All Secondary Target Drinking Water Intakes:			
Yes					
□ No		Name: Water Body: Flow (cfs): Population Served:			
Have Primary Target Drinking Water Intakes	Been Identified:				
Yes If Ves Distant	ce to Nearest Drinking				
□ No Water Intake	: Miles ⁶				
If Yes, Enter Population Served by Target Int	ake:				
De este 4		Total within 15 Miles ⁴			
People*					
Fisheries Located Along the Surface Water N	ligration Path:	List All Secondary Target Fisheries ¹⁰ :			
□ Yes □ No If Yes, Distanc	e to Nearest Fishery: Miles	<u>Water Body/ Fishery Name</u> : <u>Flow (cfs)</u> :			
Have Primary Target Fisheries Reen Identifie	d:	<u>۱</u>			
Li Yes Li No					

	8. Surface Water Pathway (continued)					
Wetlands Located Along the Surface Water M	ligration Path:	Other Sensitive E	Environments Located Along the Surface Water Migration Path:			
☐ Yes ☐ No		 ☐ Yes ☐ No If Yes, Distance to Nearest Sensitive Environment: ☐ No Miles 				
Have Primary Target Wetlands Been Identifie	ed:	Have Primary Ta	arget Sensitive Environments Been Identified:			
Yes No		Yes No				
List All Wetlands:		List All Sensitive	e Environments ¹¹ :			
<u>Water Body</u> : <u>Flow (cfs)</u> : <u>Frontage miles:</u>		Water Body :	Flow (cfs): Sensitive Environment Type:			
	0.50	:				
Are People Occupying Residence or	9.50	rkers Orsite ⁴ :	NWay - NA			
Attending School or Daycare on or Within 200 Feet of Area of Known or Suspected Contamination:	Number of WO	rkers Onsite : 9 00 - 1,000 000	Within 200 Feet of Areas of Known or Suspected Contamination:			
☐ Yes			□ No			
□ No			If Yes, List Each Terrestrial Sensitive Environment ⁵ :			
If Yes, Enter Total Residential Population: People ² Population:		People ⁷	*Refer to PA Table 7 for environment types			
		10. Air Pathway	y - NA			
Is there a Suspected Release to Air ¹ :		Wetlands Located Within 4 Miles of the Site ⁶ :				
☐ Yes ☐ No Enter Total Population on or Within:		☐ Yes ☐ No	If Yes, How Many Acres: Acres			
Onsite		Other Sensitive E	Environments Located Within 4 Miles of the Site:			
0-1/4 Mile		☐ Yes ☐ No				
>1/4-1/2 Mile		List All Sensitive Environments Within 1/2 Mile of the Site ⁶ :				
>1/2-1 Mile		Distance: Sen	nsitive Environment Type/Wetlands Area (acres):			
>1-2 Miles		Onsite				
>2-3 Miles		0-1/4 Mile				
>3-4 Miles		>1/4-1/2 Mile				
Total Within 4 Miles ³⁻⁵		*Refer to PA Table 10 for calculations on air pathway exposures				

					Identification	
Potential Haz	ardous Wast	e Site Pre	liminary As	sessment Form	State:	CERCLIS #:
					CERCLIS Discovery Date:	
		1.	General Site Info	mation		
Name: Fire Station		Street Address	5: NA			
City:		State: FL	Zip Code:32925	County:Brevard	Co. Code:12009	Cong. Dist:8th
Latitude:	Longitude:	Approximate Area of Site: Status of Site: Acres Image: Acres in the second s			ied lume, etc.)	
Site Name: Fire Station	Site Name: Fire Station (Building 810)				· ·	
station houses two fire fire engines have a foar respectively, and the fo excess AFFF that is stor	engines (Fire Engine n capacity of 55 gallc am trailer has a foan ed at Building 984.	9 and Fire Engin 9 and Fire Engin ons each, Crash 1 n capacity of 2,0	Funding 810 and is ne 4), two crash truc Fruck 6 and Crash Tr 100 gallons. The cras	ks (Crash Truck 6 and Cra ruck 5 have foam capacitie h trucks and fire engines a	sh Truck 5), and a for es of 210 gallons and are refilled at the fire	am trailer. The two I 500 gallons e station with
		2. 0	wner/Operator In	formation		
Owner: Patrick AFB			Operator: Same as	s "owner"		
Street Address:			Street Address:			
City:			City:			
State: FL	Zip Code:32925	Telephone:	State:	Zip Code:	Telephone:	
Type of Ownership: Private County Federal Agency Municipal Name: DOD Not Specified State Other Indian Other		Type of Ownership: Private County Federal Agency Municipal Name: Other Indian Indian				
		3. 9	Site Evaluator Info	rmation		
Name of Evaluator: Joh	n Sandoval	Agency/Organ	ization: HydroGeoLogic, Inc.		Date Prepared:07/06/15	
Street Address:404 E. R	amsey Road, Ste. 21	0	City:San Antonio		State:Texas	
Name of EPA or State A	gency Contact:NA		Street Address:			
City:		State:	Telephone:			
		4. Site	Disposition (for El	PA use only)	-	
Emergency Response/R Recommendation:	emoval Assessment		CERCLIS Recomme	endation: SI	Signature:	
	Yes No		Lower Priority S	51	Name (typed):	
Date:			Other: Date:	_		

	5. (General Site Chara	cteristics	
Predominant Land Use Wit	hin 1 Mile of Site (check all that	Site Setting:		Years of Operation:
apply):		_		
		🗌 Urban		Beginning Year
			an	
	✓ DOD ✓ Facility:			Ending Year <u>present</u>
Forest/Fields	DOE			
	Other			✓ Unknown
Type of Site Operations (ch	eck all that apply):			Waste Generated:
Manufacturing (must check sub	ocategory)	Retail		✓ Onsite
Lumber and Wood Produ	cts	Recycling		Offsite
Inorganic Chemicals		🗌 Junk/Salvage Yard		Onsite and Offsite
Plastic and/or Rubber Pro	oducts	🔲 Municipal Landfill		
Paints, Varnishes		Other Landfill		Waste Deposition Authorized By:
Industrial Organic Chemi	cals			Present Owner
Agricultural Chemicals				Former Owner
	Products	DUI Other Federal Facili	tv	Present & Former Owner
Primary Metals	paroving			
Metal Coating, Plating, El	ngraving	Treatment, Stora	age, or Disposal	
Eabricated Structural Met	tal Products	Large Quantity C	Generator	Waste Accessible to the Public:
		Small Quantity G	Generator	
Other Manufacturing		Subtitle D		
		🗌 Municipal		⊡ No
		Industrial		
		Converter"		Distance to Nearest Dwelling, School
		"Protective Filer"		Distance to Nearest Dweiling, School,
		□ "Non-or Late File	er"	or workplace:
		Note Specified		
		Other		<u>on-site</u> Feet
6. Waste Characteristics	Information			(Refer to PA Table 1
		for WC Score)		
Source Type:	Source Waste Quantity:	Tier*:	General Type of Waste	(check all that
(check all that apply)	(include unit)		apply):	
			Metals	Pesticides/Herbicides
Surface Impoundment			Organics	Acids/Bases
Drums				Oily Waste
Tanks and Non-Dum Container	s		Solvenis Paints /Pigments	
Chemical Waste Pile			Laboratory/Hospital Waste	
Scrap Metal or Junk Pile				C Other AFFF
Tailings Pile			Construction/Demolition V	Vaste
Trash Pile (open drum)			_	
Land Treatment				
			Physical State of Waste	as Deposited (check all that apply):
(unidentified source)				
(unidentified source)			Solid	
Contaminated Soil			Sludge	
✓ OtherAFFF_			Dev Powder	
No Sources				
*C=Constituent, W=Wa	astestream, V=Volume, A=Area			

	athway			
Is Ground Water Used for Drinking Within 4	Is There a Suspected Release to	List Secondary Target Population Served by Ground Water		
Miles:	Ground Water ¹ :	Withdrawn From:		
لا Ves				
	_	0 - 1/4 Mile		
If Yes, Distance to nearest Drinking				
Well:	Have Primary Target Drinking	>1/4 - 1/2 Mile		
_ <u>4.8</u> miles	Water Wells Been Identified:	>1/2 - 1 Mile		
Type of Drinking Water Wells Within 4		>1/2 - 1 Wille		
Miles	Ves	>1 - 2 Mile		
(check all that apply):				
	L No	>2 - 3 Mile		
✓ Private	If Yes, Enter Primary Target			
□ None	Population:	>3 - 4 Mile		
Depth to Shallowest Aquifer:	Nearest Designated Wellhead	Total Within 4 Miles ⁴ <u>16,537</u>		
A E Foot	Protection Area ⁶			
_ <u>_4-5</u> Feet	Protection Area :			
Karst Terrain/Aquifer Present:	Underlies Site	*Use population #s for PA Table 2		
	□ >0-4 Miles	*Note nearest well for #5 on GW Pathway Scoresheet		
✓ res				
	8. Surface Water Pa	athway		
Type of Surface Water Draining Site and 15 M	Ailes Downstream (check all that	Shortest Overland Distance From Any Source to Surface Water:		
apply):		Shortest overland Distance Hom Any Source to Surface Water.		
		700 5 1		
Stream River Por	nd 🗌 Lake	<u>_/00</u> Feet		
☐ Bay ☐ Ocean ☑ Oth	er_storm sewer open drainage areas_	Miles		
Is There a Suspected Release to Surface Wat	er ¹ :	Site is Located in:		
		🗌 Annual - 10 yr Floodplain		
Yes		>10yr - 100yr Floodplain		
✓ No		>100yr - 500yr Floodplain		
		>500yr Floodplain		
Drinking Water Intake Located Along the Sur	face Water Migration Path:	List All Secondary Target Drinking Water Intakes:		
		<u>Name</u> : <u>Water Body</u> : <u>Flow (cfs)</u> : <u>Population Served</u> :		
Have Primary Target Drinking Water Intakes	Been Identified:			
Yes If Vec Distance	ce to Nearest Drinking			
✓ No Water Intake	· Miles ⁶			
Water Intake	· (Miles			
If Yes, Enter Population Served by Target Inta	ake:			
Decente4		Total within 15 Miles ⁴		
Fisheries Located Along the Surface Water M	ligration Path:	List All Secondary Target Eichorics ¹⁰		
If Vac Dictance	e to Nearest Fishery:			
Yes V No	Miles	<u>water Body/ Fishery Name</u> : <u>FIOW (CTS)</u> :		
Have Primary Target Fisheries Reen Identified		1		
L Yes V No				

	8. Surface Water Pathway (continued)					
Wetlands Located Along the Surface Water I	Vigration Path:	Other Sensitive Environments Located Along the Surface Water Migration Path:				
☐ Yes ✔ No		☐ Yes ✔ No	 ☐ Yes ☐ Yes If Yes, Distance to Nearest Sensitive Environment: ☑ No feet 			
Have Primary Target Wetlands Been Identif	ied:	Have Primary Tar	get Sensitive	e Environments Been Identified:		
☐ Yes ✔ No		☐ Yes ☑ No				
List All Wetlands:		List All Sensitive E	invironments	s ¹¹ :		
<u>Water Body</u> : <u>Flow (cfs)</u> : <u>Frontage miles:</u>		<u>Water Body</u> :	<u>Flow (cfs)</u> :	Sensitive Environment Type:		
	0	Soil Exposure Pa				
Are People Occupying Residence or	S.	rkors Opsito ⁴ :	Have Terre	estrial Sensitive Environments Been Identified on or		
Are People Occupying Residence or Attending School or Daycare on or Within 200 Feet of Area of Known or Suspected Contamination:		e 00 - 1,000 000	Within 200 Contamina	 Feet of Areas of Known or Suspected ation: 		
🗋 Yes				✓ No		
✓ No	Depulation \A/it	If Yes, List Each Terrestrial Sensitive Envir		t Each Terrestrial Sensitive Environment⁵:		
If Yes, Enter Total Residential Population With Population: 16,357 (4		miles)		Table 7 for environment types		
		10. Air Pathwa	ay in the second s			
Is there a Suspected Release to Air^1 :		Wetlands Located	Within 4 Mi	iles of the Site ⁶ :		
Enter Total Population on or Within:		✓ Yes If Yes, How Many Acres: <u>unknown</u> Acres				
Onsite		Other Sensitive En	vironments	Located Within 4 Miles of the Site:		
0-1/4 Mile		✓ Yes □ No				
>1/4-1/2 Mile		List All Sensitive Environments Within 1/2 Mile of the Site ⁶ :				
>1/2-1 Mile		<u>Distance:</u> Sensi	tive Environr	ment Type/Wetlands Area (acres):		
>1-2 Miles		Onsite				
>2-3 Miles		0-1/4 Mile <u>Pac</u>	cific Ocean			
>3-4 Miles		>1/4-1/2 Mile				
Total Within 4 Miles ³⁻⁵ <u>16,357</u>		*Refer to PA Table 10 for calculations on air pathway exposures				

 $^{1\mathchar`-11}$ Refers to question number on the PA scoresheet for each particular pathway

					Identification		
Potential Ha	zardous Wast	e Site Pre	eliminary As	sessment Form	State:	CERCLIS #:	
			-		CERCLIS Discovery D	ate:	
		1.	General Site Info	rmation			
Name: Fire Truck Roll	over Area	Street Addres	s: NA				
City: State: FL		State: FL	Zip Code:32925	County:Brevard	Co. Code:12009	Cong. Dist:8th	
Latitude:	atitude: Longitude: Approximate A		Area of Site:	Status of Site:		_	
		Acres		Active Not Specif	fied		
			Square Ft	☐ Inactive	lume, etc.)		
Site Name: Fire Truck	Rollover Area						
Taxiway E and rolled o	over releasing AFFF to	the taxiway and	I surrounding grassy	areas. The amount of AFI	FF released is unknow	'n.	
		2. 0	wner/Operator In	formation			
Owner: Patrick AFB			Operator: Same as	s "owner"			
Street Address:			Street Address:				
City:			City:				
State: FL	Zip Code:32925	Telephone:	State:	Zip Code:	Telephone:		
Type of Ownership:			Type of Ownership:				
Private	🗌 County		Private	County			
✓ Federal Agency	🗌 Municipal		Federal Agency Municipal				
Name: <u>DOD</u>	🗌 Not Specif	ied	Name: Not Specified				
	Other			U Other			
		3.	Site Evaluator Info	ormation			
Name of Evaluator: Jo	ohn Sandoval	Agency/Orgar	ization: HydroGeoLogic, Inc.		Date Prepared:07/06/15		
Street Address:404 E.	Ramsey Road, Ste. 21)	City:San Antonio		State:Texas		
Name of EPA or State	Agency Contact:NA		Street Address:				
City:		State:		Telenhone			
		4. Site	Disposition (for E	PA use only)			
Emergency Response,	/Removal Assessment		CERCLIS Recomme	endation:	Signature:		
Recommendation:			Lower Priority	SI	Name (typed):		
	□ res □ No						
.			CRA		Position:		
Date: _			Date:				

5. General Site Characteristics						
Predominant Land Use Withi	n 1 Mile of Site (check all that	Site Setting:		Years of Operation:		
apply):		_				
		🗌 Urban		Beginning Year <u>1997</u>		
	Mining Other Federal		an			
Residential	DOD Facility:			Ending Year <u>NA</u>		
Forest/Fields	DOE					
	Other			Unknown		
Type of Site Operations (che	ck all that apply):			Waste Generated:		
Manufacturing (must check subc	ategory)	Retail		 Onsite 		
Lumber and Wood Product	S	Recycling		Offsite		
Inorganic Chemicals		Junk/Salvage Yard		Onsite and Offsite		
Plastic and/or Rubber Prod	ucts	Municipal Landfill				
Paints, Varnishes		Other Landfill		Waste Deposition Authorized By:		
Industrial Organic Chemica	als	✓ DOD		Present Owner		
Agricultural Chemicals				Former Owner		
Miscellaneous Chemical Pro	oducts	DOI	t.,	Present & Former Owner		
Primary Metals			ty			
Metal Coating, Plating, Eng	Iraving	Treatment, Stora	age, or Disposal	Unknown		
	Products	Large Quantity C	Generator	Waste Accessible to the Public:		
	Troducts	Small Quantity G	Generator			
Other Manufacturing		Subtitle D				
		Municipal				
		Industrial				
		Converter"		Distance to Neenest Durelling, Cale al		
		"Protective Filer"		Distance to Nearest Dweiling, School,		
UII and Gas		🗌 "Non-or Late File	er"	or Workplace:		
		Note Specified				
		Other		<u>_814</u> Feet		
6. Waste Characteristics I	nformation			(Refer to PA Table 1		
		for WC Score)				
Source Type:	Source Waste Quantity:	Tier*:	General Type of Waste	(check all that		
(check all that apply)	(include unit)		apply):			
			Metals	Pesticides/Herbicides		
			Organics	Acids/Bases		
☐ Drums				Oily Waste		
Tanks and Non-Dum Containers			Solvents	Municipal Waste		
Chemical Waste Pile						
Scrap Metal or Junk Pile				C Other AFFF		
Tailings Pile			\Box Construction/Demolition W	Vaste		
Trash Pile (open drum)						
Land Treatment						
Contaminated GW Plume			Physical State of Waste	as Deposited (check all that apply):		
(unidentified source)			_			
(unidentified source)			Solid			
Contaminated Soil			Sludge			
✓ OtherAFFF			Powder			
No Sources						
*C=Constituent, W=Wast	estream, V=Volume, A=Area		L Gas			

7. Ground Water Pathway					
Is Ground Water Used for Drinking Within 4	Is There a Suspected Release to	List Secondary Target Population Served by Ground Water			
Miles:	Ground Water ¹ :	Withdrawn From:			
لا Ves					
	_	0 - 1/4 Mile			
If Yes, Distance to nearest Drinking					
Well:	Have Primary Target Drinking	>1/4 - 1/2 Mile			
_ <u>4.5</u> miles	Water Wells Been Identified:	>1/2 - 1 Mile			
Type of Drinking Water Wells Within 4		>1/2 - 1 Wile			
Miles		>1 - 2 Mile			
(check all that apply):					
	L No	>2 - 3 Mile			
✓ Private	If Yes, Enter Primary Target				
□ None	Population:	>3 - 4 Mile			
	<u>25</u> People ³				
Depth to Shallowest Aquifer:	Nearest Designated Wellhead	Total Within 4 Miles ⁴ <u>18,099</u>			
4.5 Foot	Protection Area ⁶				
<u>_4-5</u> Feet	Protection Area :				
Karst Terrain/Aquifer Present:	Underlies Site	*Use population #s for PA Table 2			
[7] Yes	□ >0-4 Miles	*Note nearest well for #5 on GW Pathway Scoresheet			
□ Tes					
	8. Surface Water Pa	athway			
Type of Surface Water Draining Site and 15 M	Ailes Downstream (check all that	Shortest Overland Distance From Any Source to Surface Water:			
apply):					
		2.270 Fact			
Stream River Pon	nd 🗌 Lake	_ <u>_2,270</u> Feet			
	ner _	Miles			
Is There a Suspected Release to Surface Wat	er ¹ :	Site is Located in:			
		🗌 Annual - 10 yr Floodplain			
Yes		☐ >10yr - 100yr Floodplain			
✓ No		>100yr - 500yr Floodplain			
		Solver Floodplain			
Drinking Water Intake Located Along the Sur	face Water Migration Path:	List All Secondary Target Drinking Water Intakes:			
☐ Yes ✓ No		Name: Water Body, Elow (cfc), Deputation Served			
		Name. Water body. riow (cis). Population served.			
Have Primary Target Drinking Water Intakes	Been Identified:				
Yes If Yes Distance	ce to Nearest Drinking				
☑ No Water Intake	: Miles ⁶				
If Yes, Enter Population Served by Target Inta	ake:				
People ⁴		Total within 15 Miles ⁴			
Fisheries Located Along the Surface Water M	ligration Path:	List All Secondary Target Fisheries ¹⁰			
	e to Nearest Fishery	Water Rody/ Eishery Name : Elow (cfr):			
Yes ⊻ No	Miles	water Dugy Fishery Name : FIUW (UIS):			
Have Primary Target Fisheries Been Identified	d:	1			
	-				

8. Surface Water Pathway (continued)						
Wetlands Located Along the Surface Water I	Migration Path:	Other Sensitive Environments Located Along the Surface Water Migration Path:				
☐ Yes ✔ No		☐ Yes ✔ No	☐ Yes If Yes, Distance to Nearest Sensitive Environment: ✓ No feet			
Have Primary Target Wetlands Been Identif	Have Primary Tar	get Sensitive	e Environments Been Identified:			
☐ Yes ✓ No		☐ Yes ☑ No				
List All Wetlands:		List All Sensitive I	Environments	5 ¹¹ :		
Water Body : Flow (cfs): Frontage miles:		<u>Water Body</u> :	<u>Flow (cfs)</u> :	Sensitive Environment Type:		
	9.	. Soil Exposure Pa	thway			
Are People Occupying Residence or Attending School or Daycare on or Within 200 Feet of Area of Known or Suspected Contamination:		rkers Onsite ⁴ : 00 - 1,000 000	Have Terre Within 200 Contamina	Peet of Areas of Known or Suspected The top of Areas of Known or Suspected Peet of Areas of Known or Suspected Pees		
☐ Yes ✓ No		If Yes. List Each Terrestrial Sensitive Environmen		☑ No t Each Terrestrial Sensitive Environment ⁵ :		
If Yes, Enter Total Residential Population: People ² Population:		hin 1 Mile: <u>miles)</u>	*Refer to PA	Table 7 for environment types		
		10. Air Pathwa	ay			
Is there a Suspected Release to Air ¹ :		Wetlands Located	Within 4 Mi	les of the Site ⁶ :		
☐ Yes ☑ No Enter Total Population on or Within:		✓ Yes If Yes, How Many Acres: <u>unknown</u> Acres				
Onsite		Other Sensitive Environments Located Within 4 Miles of the Site:				
0-1/4 Mile		✓ Yes □ No				
>1/4-1/2 Mile		List All Sensitive Environments Within 1/2 Mile of the Site ⁶ :				
>1/2-1 Mile		Distance: Sens	itive Environr	nent Type/Wetlands Area (acres):		
>1-2 Miles		Onsite				
>2-3 Miles		0-1/4 Mile <u>Ba</u>	nana River			
>3-4 Miles		>1/4-1/2 Mile				
Total Within 4 Miles ³⁻⁵ <u>18,099</u>		*Refer to PA Table 10 fo	or calculations or	n air pathway exposures		

 $^{1\mathchar`-11}$ Refers to question number on the PA scoresheet for each particular pathway

					Identification		
Potential Haza	rdous Waste	e Site Prel	iminary As	sessment Form	State:	CERCLIS #:	
			-		CERCLIS Discovery Date:		
		1. (General Site Info	rmation			
Name: Former Fire Train	ning Area 1	Street Address	: NA				
City:		State: FL	Zip Code:32925	County:Brevard	Co. Code:12009	Cong. Dist:8th	
Latitude:	Longitude:	Approximate A 0.25	Approximate Area of Site: Status of Site: 0.25 Acres Active Not Specific Square Ft Inactive NA (GW plue)			ïed lume, etc.)	
Site Name: Former Fire	Training Area 1						
Site Description: Former Fire Training Area (FTA) 1 (Site FT-21 [Solid Waste Mangement Unit #032]) was located along the eastern central portion of Patrick AFB, southeast of Building 820. The Former FTA was in operation from 1950 to 1963 and was used for burning waste fuels (Aviation gasoline, motor gasoline, and diesel), waste oils, halogenated and non-halogenated solvents during firefighting training exercises. The site consisted of approximately 0.25 acres with a shallow unlined depression in sandy soils into which combustible material were placed and ignited during fire training activities. The operational timeframe of the Former FTA was prior to the use of AFFF by the Air Force; therefore, impact to the environmental media surrounding the Former FTA is not possible.							
	2 Owner/Operator Information						
Owner: Patrick AFB		2.01	Operator: Same as "owner"				
Street Address:			Street Address:				
City:			City:				
State: FL	Zip Code:32925	Telephone:	State:	Zip Code:	Telephone:		
Type of Ownership:		•	Type of Ownersh	ip:			
 Private Federal Agency Name: <u>DOD</u> State Indian 	□ Private □ County □ Private □ Municipal □ Name: _DOD □ Not Specified □ State □ Other □ Indian □ Other		 Private County Federal Agency Municipal Name: Not Specified State Other Indian 				
		3. Si	ite Evaluator Info	ormation			
Name of Evaluator: John	n Sandoval	Agency/Organi	ization: HydroGeol	Logic, Inc.	Date Prepared:07/0	6/15	
Street Address:404 E. Ra	amsey Road, Ste. 210		City:San Antonio		State:Texas		
Name of EPA or State Ag	gency Contact:NA		Street Address:				
City: State:		Telephone:					
		4. Site D	Disposition (for E	PA use only)			
Emergency Response/Re	emoval Assessment			endation:	Signature:		
Recommendation:	Yes			SI	Name (typed):		
Date:			RCRA Other: Date:		Position:		

5. General Site Characteristics - NA						
Predominant Land Use W	ithin 1 Mile of	f Site (check all that	Site Setting:		Years of Operation:	
apply):	Agriculture	🗌 DOI	🗌 Urbar	ı	Beginning Year	
Commercial Residential Forest/Fields	Mining DOD DOF	Other Federal	Subu	rban	Ending Year	
		Other			Unknown	
Type of Site Operations (c	heck all that a	apply):			Waste Generated:	
Manufacturing (must check subcategory) Lumber and Wood Products Inorganic Chemicals Resting and/or Pubber Products			 Retail Recycling Junk/Salvage Yard Municipal Landfill Other Landfill DOD DOE DOI Other Federal Facility ACRA 		 Onsite Offsite Onsite and Offsite 	
 Plastic and/or Rubber Products Paints, Varnishes Industrial Organic Chemicals Agricultural Chemicals Miscellaneous Chemical Products Primary Metals Metal Coating, Plating, Engraving 		Waste Deposition Authorized By: Present Owner Former Owner Present & Former Owner Unauthorized Unknown 				
 Metal Forging, Stampin Fabricated Structural M Electronic Equipment Other Manufacturing Mining Metals Coal Oil and Gas Non-metallic Minerals 	ig letal Products		☐ Treatment, Sto ☐ Large Quantity ☐ Small Quantity ☐ Subtitle D ☐ Municipa ☐ Industria ☐ "Converter" ☐ "Protective File ☐ "Non-or Late F ☐ Note Specified ☐ Other	orage, or Disposal generator Generator I I er" Filer"	Waste Accessible to the Public:	
6. Waste Characteristic	cs Informatio	on - NA			(Refer to PA	
			Table 1 for WC So	core)		
Source Type: (check all that apply)	Source (include u	e Waste Quantity:	Tier*:	General Type of Waste apply):	(check all that	
Landfill Surface Impoundment Drums Tanks and Non-Dum Contain Chemical Waste Pile Scrap Metal or Junk Pile Tailings Pile Trash Pile (open drum) Land Treatment Contaminated GW/ Plume	ers			Metals Organics Inorganics Solvents Paints/Pigments Laboratory/Hospital Waste Radioactive Waste Construction/Demolition V		
(unidentified source) Contaminated SW/Sediment (unidentified source)					as Deposited (check all that apply):	
Contaminated Soil Other No Sources				Situge Powder Liquid Gas		
*C=Constituent, W=\	vastestream, V=Vo	oiume, A=Area				

7. Ground Water Pathway - NA					
Is Ground Water Used for Drinking Within 4	Is There a Suspected Release to	List Secondary Target Population Served by Ground Water			
Miles:	Ground Water ¹ :	Withdrawn From:			
□ Yes	Ves				
		0 - 1/4 Mile			
If Yes, Distance to nearest Drinking		>1/4 1/2 Mile			
Well:	Have Primary Target Drinking	>1/4 - 1/2 Mile			
Feet	Water Wells Been Identified:	>1/2 - 1 Mile			
Type of Drinking Water Wells Within 4					
Miles		>1 - 2 Mile			
(check all that apply):					
	If Yes, Enter Primary Target	>2 - 3 Mile			
Private	Population:				
None None	People ³	>3 - 4 Mile			
Depth to Shallowest Aquifer:	Nearest Designated Wellhead				
Feet	Protection Area ⁶	Total Within 4 Miles ⁴			
Karst Terrain/Aquifer Present:					
☐ Yes	\square None Within 4 Miles	*Use population #s for PA Table 2			
		[↑] Note nearest well for #5 on GW Pathway Scoresheet			
	8. Surface Water Patl	hway - NA			
Type of Surface Water Draining Site and 15 N	Ailes Downstream (check all that	Shortest Overland Distance From Any Source to Surface Water:			
		E			
Stream River Por	nd 🗌 Lake	Feet			
Bay Ocean Oth	ner	Miles			
Is There a Suspected Release to Surface Wat	er ¹ :	Site is Located in:			
		🗌 Annual - 10 yr Floodplain			
Yes		\square >10yr - 100yr Floodplain			
🗌 No		>100yr - 500yr Floodplain			
		>500yr Floodplain			
Drinking Water Intake Located Along the Sur	face Water Migration Path:	List All Secondary Target Drinking Water Intakes:			
Yes					
□ No		Name: Water Body: Flow (cfs): Population Served:			
Have Primary Target Drinking Water Intakes	Been Identified:				
Yes If Ves Distant	ce to Nearest Drinking				
□ No Water Intake	: Miles ⁶				
If Yes, Enter Population Served by Target Int	ake:				
De este 4		Total within 15 Miles ⁴			
People*					
Fisheries Located Along the Surface Water N	ligration Path:	List All Secondary Target Fisheries ¹⁰ :			
□ Yes □ No If Yes, Distanc	e to Nearest Fishery: Miles	<u>Water Body/ Fishery Name</u> : <u>Flow (cfs)</u> :			
Have Primary Target Fisheries Reen Identifie	d:	<u>۱</u>			
Li Yes Li No					

8. Surface Water Pathway (continued)						
Wetlands Located Along the Surface Water M	igration Path:	Other Sensitive Environments Located Along the Surface Water Migration Path:				
☐ Yes ☐ No		☐ Yes ☐ No	If Yes, [Distance to Nearest Sensitive Environment: _ Miles		
Have Primary Target Wetlands Been Identifie	ed:	Have Primary Ta	arget Sensitive	e Environments Been Identified:		
☐ Yes ☐ No	□ Yes □ No					
List All Wetlands:		List All Sensitive	Environments	¹¹ :		
<u>Water Body</u> : <u>Flow (cfs)</u> : <u>Frontage miles:</u>		<u>Water Body</u> :	<u>Flow (cfs)</u> :	Sensitive Environment Type:		
	9.50	il Exposure Path				
Are People Occupying Residence or	Number of Wo	rkers Onsite ⁴	Have Terres	trial Sensitive Environments Been Identified on or		
Attending School or Daycare on or Within 200 Feet of Area of Known or Suspected Contamination:		e 00 - 1,000 000	Within 200 F Contaminati	Feet of Areas of Known or Suspected fon:		
				□ No		
			If Yes, List Each Terrestrial Sensitive Environme			
If Yes, Enter Total Residential Population Wit Population: People ²		People ⁷	*Refer to PA T	able 7 for environment types		
		10. Air Pathway	- NA			
Is there a Suspected Release to Air ¹ :		Wetlands Locate	d Within 4 Mi	les of the Site ⁶ :		
Enter Total Population on or Within:		☐ Yes ☐ No	If Yes, How	Many Acres: Acres		
Onsite		Other Sensitive Environments Located Within 4 Miles of the Site:				
0-1/4 Mile			☐ Yes ☐ No			
>1/4-1/2 Mile		List All Sensitive Environments Within 1/2 Mile of the Site ⁶ :		Within 1/2 Mile of the Site ⁶ :		
>1/2-1 Mile		<u>Distance:</u> <u>Sen</u>	nsitive Environr	nent Type/Wetlands Area (acres):		
>1-2 Miles		Onsite				
>2-3 Miles		0-1/4 Mile				
>3-4 Miles		>1/4-1/2 Mile				
Total Within 4 Miles ³⁻⁵		*Refer to PA Table 10	for calculations or	n air pathway exposures		

					Identification	_
Potential Haza	ndous Waste	Sito Prol	iminary As	sessment Form	State:	CERCLIS #:
		Site riei		sessment i onni	CERCLIS Discovery Date:	
		1 (Conoral Sita Info	rmation		
Name: Former Fire Train	ning Area 2	L. C		rmation		
Name. Former File fram		Street Address	. NA			
City:		State: FL	Zip Code:32925	County:Brevard	Co. Code:12009	Cong. Dist:8th
Latitude:	Longitude:	Approximate A	rea of Site:	Status of Site:		•
			Acres	Active Not Specif	ied	
			Square Ft	✓ Inactive □ NA (GW pl	ume, etc.)	
Site Name: Former Fire Training Area 2 Site Description:Former FTA 2 (Site FT-22) consisted of two burn pits (Burn Pit 1 and Burn Pit 2) located on northwestern portion of Patrick AFB. Burn pit 1 was an unlined circular burn pit approximately 2 feet in depth and 150 feet in diameter. Combustible wastes were ignited in the burn pit from 1963 to 1978 and only uncontaminated fuels were used after 1978. The Assistant Fire Chief was not aware of fire training activities conducted at burn pit 1 (Appendix C, Records of Communication; Environmental Science and Engineering, Inc., 1988). Based on the use of AFFF by the Air Force in 1970, the potential exists for AFFF being used to extinguish fires during the operational period from 1970 through 1978. Burn pit 2 is a lined pit with a concrete pad that was constructed and has been in operations since 1985. The new training facility was equipped with a water collection and treatment system. According to the Assistant Fire Chief, AFFF was used when fire training activities were performed and occasionally was accidentally over sprayed outside of the lined pit. Additionally, during high winds AFFF was blown to the ground surface surrounding the pit. The Assistant Fire Chief was not aware of the amounts of AFFF accidentally released to surrounding environment at burn pit 2. Fire training activities at the burn pit ceased around 2000/2001 (Appendix C, Records of Communication; Environmental Science and Engineering, Inc., 1988). The potential for PFC Contamination to the environmental media surrounding burn pit 2 exists based on accidental releases/overspray outside of the lined pit.						
		2. Ow	/ner/Operator In	formation		
Owner: Patrick AFB			Operator: Same a	as "owner"		
Street Address:			Street Address:			
City:			City:			
State: Fl	Zin Code:32925	Telephone:	State:	Zin Code:	Telephone:	
	210 0000.32323	relephone.	State.		relephone.	
Type of Ownership:			Type of Ownersh	ip:		
 Private Federal Agency Name: DOD State Indian 	 County Municipal Not Specifi Other 	ed	Private County Federal Agency Municipal Name: Not Specified State Other Indian Other			
		3. Si	ite Evaluator Info	ormation		
Name of Evaluator: Johr	n Sandoval	Agency/Organi	ization: HydroGeoLogic, Inc.		Date Prepared:07/06/15	
Street Address:404 E. Ra	amsey Road, Ste. 210		City:San Antonio		State:Texas	
Name of EPA or State Ag	gency Contact:NA		Street Address:			
City:		State:	1	Telephone:		
		4. Site D	Disposition (for E	PA use only)		
Emergency Response/Re	emoval Assessment		CERCLIS Recomm	endation:	Signature:	
Recommendation:			Higher Priority	/ SI		
	Yes		Lower Priority	SI	Name (typed):	
	🗌 No				Position:	
Date:			Date:			
			I		I	

5. General Site Characteristics						
Predominant Land Use Withir	n 1 Mile of Site (check all that	Site Setting:		Years of Operation:		
apply):	Agriculture 🔲 DOI	🗌 Urban		Beginning Year <u>1963</u>		
Commercial	Mining Other Federal DOD Facility:	☐ Suburt ☑ Rural	ban	Ending Year		
	Other			Unknown		
Type of Site Operations (chec	k all that apply):			Waste Generated:		
Manufacturing (must check subca Lumber and Wood Products Inorganic Chemicals Plastic and/or Rubber Produ	itegory) ; icts	 □ Retail □ Recycling □ Junk/Salvage Yard □ Municipal Landfill 		 ✓ Onsite ☐ Offsite ☐ Onsite and Offsite 		
 Paints, Varnishes Industrial Organic Chemical Agricultural Chemicals Miscellaneous Chemical Pro Primary Metals Metal Coating, Plating, Engr 	s ducts raving	 Other Landfill DOD DOE DOI Other Federal Faci RCRA 	ility	Waste Deposition Authorized By: Present Owner Former Owner Present & Former Owner Unauthorized Unknown		
 Metal Forging, Stamping Fabricated Structural Metal Electronic Equipment Other Manufacturing Mining Metals Coal Oil and Gas Non-metallic Minerals 	Products	Treatment, Store Large Quantity Small Quantity Subtitle D Municipal Industrial "Converter" "Protective Filee "Non-or Late Filee Note Specified	rage, or Disposal Generator Generator r" iler"	Waste Accessible to the Public: Yes No Distance to Nearest Dwelling, School, or Workplace:		
		Other		<u>_290</u> Feet		
6. Waste Characteristics In	nformation	1 for WC Score)	(Refer to PA Table		
Source Type:	Source Waste Quantity:	Tier*:	General Type of Waste	(check all that		
 Landfill Surface Impoundment Drums Tanks and Non-Dum Containers Chemical Waste Pile Scrap Metal or Junk Pile Tailings Pile Trash Pile (open drum) Land Treatment Contaminated GW Plume (unidentified source) Contaminated SW/Sediment 	(include unit)		Appiy). Metals Organics Inorganics Solvents Paints/Pigments Laboratory/Hospital Waste Radioactive Waste Construction/Demolition W Physical State of Waste	 ☐ Pesticides/Herbicides ☐ Acids/Bases ☐ Oily Waste ☐ Municipal Waste ☐ Mining Waste ☐ Explosives ☑ OtherAFFF 		
(unidentified source) ☐ Contaminated Soil ✓ Other <u>AFFF_</u> ☐ No Sources *C=Constituent, W=Waster	estream, V=Volume, A=Area		☐ Solid ☐ Sludge ☐ Powder ✔ Liquid ☐ Gas			

	7. Ground Water Pathway - NA					
Is Ground Water Used for Drinking Within 4	Is There a Suspected Release to	List Secondary Target Population Served by Ground Water				
Miles:	Ground Water ¹ :	Withdrawn From:				
J Yes	Ves					
	_	0 - 1/4 Mile				
If Yes, Distance to nearest Drinking						
Well:	Have Primary Target Drinking	>1/4 - 1/2 Mile				
_ <u>4.56</u> Feet	Water Wells Been Identified:	>1/2 - 1 Mile				
Type of Drinking Water Wells, Within 4		>1/2 - 1 Mile				
Miles	Ves	>1 - 2 Mile				
(check all that apply):						
	L] No	>2 - 3 Mile				
✓ Private	If Yes, Enter Primary Target					
	Population:	>3 - 4 Mile				
	_25 People ³					
Depth to Shallowest Aquifer:	Nearest Designated Wellhead	Total Within 4 Miles ⁴ <u>20,379</u>				
A-5 Foot	Protoction Aroa ⁶					
<u>_4-5</u>	Protection Area .					
Karst Terrain/Aquifer Present:		*Use population #s for PA Table 2				
[] Vos	□ >0-4 Miles	*Note nearest well for #5 on GW Pathway Scoresheet				
□ Tes						
	8. Surface Water P	athway				
Type of Surface Water Draining Site and 15 M	Ailes Downstream (check all that	Shortest Overland Distance From Any Source to Surface Water:				
apply):						
		100 East				
☐ Stream	nd 🗌 Lake	<u> 100 </u>				
Bay Ocean Oth	ier	Miles				
Is There a Suspected Release to Surface Wat	er ¹ :	Site is Located in:				
		🗌 Annual - 10 vr Floodplain				
✓ Yes		☐ >10yr - 100yr Floodplain				
□ No		✓ >100yr - 500yr Floodplain				
		└ >500yr Floodplain				
Drinking Water Intake Located Along the Sur	face Water Migration Path:	List All Secondary Target Drinking Water Intakes:				
Yes						
		Name: Water Body: Flow (Cts): Population Served:				
Have Primary Target Drinking Water Intakes	Been Identified:					
Yes If Yes. Distance	ce to Nearest Drinking					
✓ No Water Intake	: Miles ⁶					
If Yes, Enter Population Served by Target Inta	аке:					
Decelet		Total within 15 Miles ⁴				
Peopler						
People*						
Fisheries Located Along the Surface Water N	ligration Path:	List All Secondary Target Fisheries ¹⁰ :				
Fisheries Located Along the Surface Water M	ligration Path: e to Nearest Fisherv:	List All Secondary Target Fisheries ¹⁰ : Water Body/ Fishery Name : Flow (cfs):				
Fisheries Located Along the Surface Water N Yes Vo If Yes, Distance	ligration Path: e to Nearest Fishery: Miles	List All Secondary Target Fisheries ¹⁰ : <u>Water Body/ Fishery Name</u> : <u>Flow (cfs)</u> :				
Fisheries Located Along the Surface Water M Yes Volume No Have Primary Target Fisheries Been Identifie	ligration Path: e to Nearest Fishery: Miles d:	List All Secondary Target Fisheries ¹⁰ : <u>Water Body/ Fishery Name</u> : <u>Flow (cfs)</u> :				
Fisheries Located Along the Surface Water M Yes No If Yes, Distance Have Primary Target Fisheries Been Identifie	ligration Path: e to Nearest Fishery: Miles d:	List All Secondary Target Fisheries ¹⁰ : <u>Water Body/ Fishery Name</u> : <u>Flow (cfs)</u> :				

8. Surface Water Pathway (continued)						
Wetlands Located Along the Surface Water N	Aigration Path:	Other Sensitive E	nvironments Located Along the Surface Water Migration Path:			
✓ Yes □ No		☐ Yes ☑ No	If Yes, Distance to Nearest Sensitive Environment: Miles			
Have Primary Target Wetlands Been Identif	ied:	Have Primary Ta	rget Sensitive Environments Been Identified:			
✓ Yes □ No			✓ Yes □ No			
List All Wetlands:		List All Sensitive	Environments ¹¹ :			
Water Body : Flow (cfs): Frontage miles:		<u>Water Body</u> :	Flow (cfs): Sensitive Environment Type:			
Baznan River		multiple wetlands				
	9.	Soil Exposure Pa	nthway			
Are People Occupying Residence or Attending School or Daycare on or Within 200 Feet of Area of Known or Suspected Contamination:		rkers Onsite ⁴ : 00 - 1,000 000	Have Terrestrial Sensitive Environments Been Identified on or Within 200 Feet of Areas of Known or Suspected Contamination:			
Yes			✓ No			
✓ No	Deputation \A/it	hin 1 NAilar	If Yes, List Each Terrestrial Sensitive Environment ⁵ :			
If Yes, Enter Total Residential Population Wit Population: 20,379 People ² 20,379		_People ⁷	*Refer to PA Table 7 for environment types			
		10. Air Pathw	ау			
Is there a Suspected Release to Air ¹ :		Wetlands Located	d Within 4 Miles of the Site ⁶ :			
□ Yes ☑ No Enter Total Population on or Within:		✓ Yes □ No	If Yes, How Many Acres: <u>unknown</u> Acres			
Onsite		Other Sensitive E	nvironments Located Within 4 Miles of the Site:			
0-1/4 Mile		✓ Yes □ No				
>1/4-1/2 Mile		List All Sensitive Environments Within 1/2 Mile of the Site ⁶ :				
>1/2-1 Mile		Distance: Sen	sitive Environment Type/Wetlands Area (acres):			
>1-2 Miles		Onsite				
>2-3 Miles		0-1/4 Mile	anana River			
>3-4 Miles		>1/4-1/2 Mile	multieple wetlands			
Total Within 4 Miles ³⁻⁵ <u>20,379</u>		*Refer to PA Table 10	for calculations on air pathway exposures			

					Identification	
Potential Haza	ardous Wast	e Site Pre	liminary As	sessment Form	State:	CERCLIS #:
			-		CERCLIS Discovery Da	ate:
		1. (General Site Info	rmation		
Name: Hangar 630		Street Address	s: NA			
City: State: FL		State: FL	Zip Code:32925	County:Brevard	Co. Code:12009	Cong. Dist:8th
Latitude:	Longitude:	Approximate /	Area of Site:	Status of Site:		_
			Acres	Active 🗌 Not Specif	ied	
			Square Ft	🗌 Inactive 🛛 🗌 NA (GW pl	lume, etc.)	
Site Name: Hangar 630		J				
Site Description: Hanga	r 630 is located in the	e northwestern	portion of Patrick	AFB. Hangar 630 was cons	structed in 1964 and is	s currently
equipped with an AFFF	fire suppression syst	em. The hangar	is supplied with 39	% AFFF from an 800 gallor	n AST located in the ha	angar mechanical
room.		-				-
		2. Ov	vner/Operator In	formation		
Owner: Patrick AFB			Operator: Same a	as "owner"		
Street Address:			Street Address:			
City:			City:			
State: FL	Zip Code:32925	Telephone:	State:	Zip Code:	Telephone:	
Type of Ownership:			Type of Ownersh	ip:		
☐ Private						
✓ Federal Agency	Municipal		Federal Agency Municipal			
Name: <u>DOD</u>	Not Specif	ied	Name: Not Specified			
	Other			Other		
		3. S	ite Evaluator Info	ormation		
Name of Evaluator: Joh	n Sandoval	Agency/Organ	ization: HydroGeoLogic, Inc.		Date Prepared:07/06/15	
			, , ,			
Street Address:404 E. R	amsey Road, Ste. 210))	City:San Antonio		State:Texas	
	,					
Name of EPA or State A	gency Contact:NA		Street Address:			
	-					
City:		State:		Telephone:		
		4. Site [Disposition (for E	PA use only)		
Emergency Response/R	emoval Assessment		CERCLIS Recomm	nendation:	Signature:	
Recommendation:			Higher Priority	y SI		
	Yes		Lower Priority	r SI	Name (typed):	
	🗌 No				Desition:	
Date			Other:			
Date:			Date:			

	5. General Site Characteristics						
Predominant Land Use Within	1 Mile of Site (check all that	Site Setting:		Years of Operation:			
apply):	griculture 🔲 DOI	🗌 Urban		Beginning Year <u>1964</u>			
Commercial M Residential J Forest/Fields	Ining Other Federal OD Facility:	☐ Suburt ✓ Rural	ban	Ending Yearpresent			
	Other			Unknown			
Type of Site Operations (check	all that apply):			Waste Generated:			
Manufacturing (must check subcategory) Lumber and Wood Products Inorganic Chemicals		 □ Retail □ Recycling □ Junk/Salvage Yard □ Municipal Landfill 		 Onsite Offsite Onsite and Offsite 			
 Paints, Varnishes Industrial Organic Chemicals Agricultural Chemicals Miscellaneous Chemical Produce Primary Metals Metal Coating, Plating, Engradimeter 	ucts	 Other Landfill DOD DOE DOI Other Federal Faci RCRA 	lity	Waste Deposition Authorized By: Present Owner Former Owner Present & Former Owner Unauthorized Unknown			
 Metal Forging, Stamping Fabricated Structural Metal P Electronic Equipment Other Manufacturing Mining Metals Coal Oil and Gas Non-metallic Minerals 	roducts		rage, or Disposal Generator Generator r" ler"	Waste Accessible to the Public: Yes No Distance to Nearest Dwelling, School, or Workplace:			
		Other		<u>on-site</u> Feet			
6. Waste Characteristics In	formation	1 for WC Score)	(Refer to PA Table			
Source Type:	Source Waste Quantity:	Tier*:	General Type of Waste	(check all that			
 (check all that apply) Landfill Surface Impoundment Drums Tanks and Non-Dum Containers Chemical Waste Pile Scrap Metal or Junk Pile Tailings Pile Trash Pile (open drum) Land Treatment Contaminated GW Plume (unidentified source) 	(include unit)		Appiy): Metals Organics Inorganics Solvents Paints/Pigments Laboratory/Hospital Waste Radioactive Waste Construction/Demolition W Physical State of Waste a	 Pesticides/Herbicides Acids/Bases Oily Waste Municipal Waste Mining Waste Explosives OtherAFFF 			
Contaminated SW/Sediment (unidentified source) Contaminated Soil OtherAFEF_ No Sources *C=Constituent, W=Wastes	tream, V=Volume, A=Area		☐ Solid ☐ Sludge ☐ Powder ✔ Liquid ☐ Gas				

	7. Ground Water Pat	hway - NA
Is Ground Water Used for Drinking Within 4	Is There a Suspected Release to	List Secondary Target Population Served by Ground Water
Miles:	Ground Water ¹ :	Withdrawn From:
ন Yes	Ves.	
		0 - 1/4 Mile
If Yes, Distance to nearest Drinking		>1/4 1/2 Mile
Well:	Have Primary Target Drinking	>1/4 - 1/2 Mile
<u>4.8</u> miles	Water Wells Been Identified:	>1/2 - 1 Mile
Type of Drinking Water Wells Within 4		
Miles	√ Yes	>1 - 2 Mile
(check all that apply):		
		>2 - 3 Mile
✓ Private	If Yes, Enter Primary Target	
None None	Population:	>3 - 4 Mile
	_25 People ³	
Depth to Shallowest Aquifer:	Nearest Designated Wellhead	Total Within 4 Miles ⁴ <u>16,416</u>
4-5 Feet	Protection Area ⁶ :	
		*
Karst Terrain/Aquifer Present:		*Use population #s for PA Table 2
✓ Yes	None Within 4 Miles	Note hearest well for #5 on GW Pathway Scoresheet
□ No	_	
	8. Surface Water Pa	athway
Type of Surface Water Draining Site and 15 M	Ailes Downstream (check all that	Shortest Overland Distance From Any Source to Surface Water:
apply):		
Stream River Der	od Distant	1.042 Feet
		Miles
Is There a Suspected Release to Surface Wat	er ¹ :	Site is Located in:
_		🗌 Annual - 10 yr Floodplain
✓ Yes		>10yr - 100yr Floodplain
		✓ >100yr - 500yr Floodplain
Drinking Water Intake Located Along the Sur	face Water Migration Path:	List All Secondary Target Drinking Water Intakes:
✓ No		Name: Water Body: Elow (cfs): Population Served:
		wane. water body. now (crs). robulation served.
Have Primary Target Drinking Water Intakes	Been Identified:	
Yes If Yes, Distant	ce to Nearest Drinking	
✓ No Water Intake	: Miles ⁶	
If Yes, Enter Population Served by Target Inte	ake:	
Decele4		Total within 15 Miles ⁴
Fisherics Loosted Along the Surface Water N	ligration Dath.	
Fisheries Located Along the Surface Water N	ingration Path:	List All Secondary Target Fisheries [™] :
□ Yes ☑ No If Yes, Distanc	e to nearest Fishery:	Water Body/ Fishery Name : Flow (cfs):
Have Drimany Target Fisheries Deen Identifie	ivilies	
	u.	
🗌 Yes 🔽 No		

	8. Surfac	e Water Pathwa	y (continue	d)	
Wetlands Located Along the Surface Water	Migration Path:	Other Sensitive Environments Located Along the Surface Water Migration Path:			
✓ Yes □ No		✓ Yes☐ No	lf Yes, I _ <u>1,042</u>	Distance to Nearest Sensitive Environment: feet	
Have Primary Target Wetlands Been Identif	ied:	Have Primary Ta	rget Sensitive	e Environments Been Identified:	
✓ Yes □ No			✓ Yes □ No		
List All Wetlands:		List All Sensitive	Environment	S ¹¹ :	
Water Body : Flow (cfs): Frontage miles:		Water Body :	<u>Flow (cfs)</u> :	Sensitive Environment Type:	
Baznan River		multiple wetlands			
	9.	Soil Exposure Pa	thway		
Are People Occupying Residence or Attending School or Daycare on or Within 200 Feet of Area of Known or Suspected Contamination:		rkers Onsite ⁴ : 00 - 1,000 000	Have Terres Within 200 I Contaminati	trial Sensitive Environments Been Identified on or Feet of Areas of Known or Suspected ion:	
Tes Yes				☑ No	
√ No			If Yes, List	Each Terrestrial Sensitive Environment ⁵ :	
If Yes, Enter Total Residential Population With Population:		<u>niles)</u>	*Refer to PA T	- able 7 for environment types	
		10. Air Pathw	ау		
Is there a Suspected Release to Air ¹ : Yes No Enter Total Population on or Within:		Wetlands Located Ves No	d Within 4 Mi If Yes, How	iles of the Site ⁶ : Many Acres: <u>unknown</u> Acres	
Onsite		Other Sensitive E	nvironments	Located Within 4 Miles of the Site:	
0-1/4 Mile		✓ Yes □ No			
>1/4-1/2 Mile		List All Sensitive Environments Within 1/2 Mile of the Site ⁶ :		Within 1/2 Mile of the Site ⁶ :	
>1/2-1 Mile		<u>Distance:</u> <u>Sen</u>	sitive Environr	ment Type/Wetlands Area (acres):	
>1-2 Miles		Onsite			
>2-3 Miles		0-1/4 Mile <u>B</u>	anana River		
>3-4 Miles		>1/4-1/2 Mile	multieple weth	ands	
Total Within 4 Miles ³⁻⁵ <u>16,416</u>		*Refer to PA Table 10 for calculations on air pathway exposures			

 $^{1\mathchar`-11}$ Refers to question number on the PA scoresheet for each particular pathway

					Identification		
Potential Haz	ardous Wast	e Site Pre	eliminary Ass	sessment Form	State:	CERCLIS #:	
			-		CERCLIS Discovery D	ate:	
		1.	General Site Infor	mation			
Name: Hangar 647		Street Addres	s: NA				
City: State: FL		State: FL	Zip Code:32925	County:Brevard	Co. Code:12009	Cong. Dist:8th	
Latitude:	Longitude:	Approximate	Area of Site:	Status of Site:			
		Acres		Active 🗌 Not Specif	ied		
			Square Ft	Inactive INA (GW p	lume, etc.)		
Site Name: Hangar 647		•					
Site Description: Hanga	ar 647 is located in the	e northwestern	portion of Patrick A	FB. Hangar 647 is a fuel ce	ell maintenance hanga	ir that was	
constructed in 1970 an	d is currently equippe	ed with an AFFF	fire suppression sys	tem. The hangar is supplie	ed with 3% AFFF from	a 2,000 gallon	
AST located in the hang	gar mechanical room.						
		2. 0	wner/Operator In	formation			
Owner: Patrick AFB			Operator: Same as	s "owner"			
Street Address:			Street Address:				
City:			City:				
State: FL	Zip Code:32925	Telephone:	State:	Zip Code:	Telephone:		
Type of Ownership:			Type of Ownership):			
Private	County		Private	County			
✓ Federal Agency	Municipal		Federal Agency	Municipal			
Name: <u>DOD</u>	Not Specif	ied	Name:	Not Specified			
	Other			Other			
		3.	Site Evaluator Info	rmation			
Name of Evaluator: Joh	ın Sandoval	Agency/Orgar	ization: HydroGeoLogic, Inc.		Date Prepared:07/06/15		
Street Address:404 E. F	amsey Road, Ste. 21	0	City:San Antonio		State:Texas		
Name of EPA or State A	Agency Contact:NA		Street Address:		I		
City:		State:	Telephone:				
		4. Site	Disposition (for El	A use only	Circulture		
Emergency Response/F	Removal Assessment		CERCLIS Recomme	endation:	Signature:		
Recommendation:			Lower Priority	51 51	Name (typed):		
			□ NFRAP				
					Position:		
Date:			Date:	_			

5. General Site Characteristics					
Predominant Land Use Wit	hin 1 Mile of Site (check all that	Site Setting:		Years of Operation:	
apply):		_			
		🗌 Urban		Beginning Year <u>1970</u>	
			an		
Residential	✓ DOD ✓ Facility:			Ending Yearpresent	
Forest/Fields	DOE				
	Other			Unknown	
Type of Site Operations (ch	eck all that apply):			Waste Generated:	
Manufacturing (must check sul	ocategory)	Retail		✓ Onsite	
Lumber and Wood Produ	cts	Recycling		Offsite	
Inorganic Chemicals		Junk/Salvage Yard		Onsite and Offsite	
Plastic and/or Rubber Pro	oducts	🔲 Municipal Landfill			
Paints, Varnishes		Other Landfill		Waste Deposition Authorized By:	
Industrial Organic Chemi	cals			Present Owner	
Agricultural Chemicals				Former Owner	
	Products	Other Federal Facili	ty	Present & Former Owner	
Metal Coating Plating Fi	paraving				
	igraving	Treatment, Stora	age, or Disposal		
Fabricated Structural Met	tal Products	Large Quantity C	Generator	Waste Accessible to the Public:	
Electronic Equipment		Small Quantity G	Generator		
Other Manufacturing		Subtitle D		□ Yes	
Mining		🔲 Municipal		☑ No	
		Industrial		_	
		Converter"		Distance to Nearest Dwelling School	
		"Protective Filer"		Distance to Nearest Dweining, School,	
		□ "Non-or Late File	∋r"	or workplace:	
		Note Specified			
		U Other		<u>on-site</u> Feet	
6. Waste Characteristics	Information			(Refer to PA Table 1	
		for WC Score)			
Source Type:	Source Waste Quantity:	Tier*:	General Type of Waste	(check all that	
(check all that apply)	(include unit)		apply):		
			Metals	Pesticides/Herbicides	
			Organics	Acids/Bases	
				Oily Waste	
Tanks and Non-Dum Container	s		Solvents	Municipal Waste	
Chemical Waste Pile					
Scrap Metal or Junk Pile				C Other AFFF	
Tailings Pile				Vaste	
Trash Pile (open drum)					
Land Treatment					
Contaminated GW Plume			Physical State of Waste	as Deposited (check all that apply):	
(unidentified source)					
(unidentified source)			Solid		
Contaminated Soil			Sludge		
OtherAFFF			∐ Powder		
No Sources					
*C=Constituent, W=Wa	astestream, V=Volume, A=Area				

	7. Ground Water Pa	athway
Is Ground Water Used for Drinking Within 4	Is There a Suspected Release to	List Secondary Target Population Served by Ground Water
Miles:	Ground Water ¹ :	Withdrawn From:
لع Yes	لا Yes	
□ No	□ No	
		0 - 1/4 Mile
If Yes, Distance to nearest Drinking		>1/4 - 1/2 Mile
Well:	Have Primary Target Drinking	
<u>4.8</u> miles	Water Wells Been Identified:	>1/2 - 1 Mile
Type of Drinking Water Wells Within 4		
Miles	✓ Yes	>1 - 2 Mile
(check all that apply):	🗌 No	
Municipal	If Vee, Fater Driver w. Towart	>2 - 3 Mile
	Ropulation:	>3 - 4 Mile
	25 People ³	23 - 4 Mile
		Total Within 4 Miles ⁴ 16,416
Depth to Shallowest Aquifer:	Nearest Designated Wellnead	
_ <u>4-5</u> Feet	Protection Area [°] :	
Karst Terrain/Aquifer Present:	Underlies Site	*Use population #s for PA Table 2
	>0-4 Miles	*Note nearest well for #5 on GW Pathway Scoresheet
✓ res		
	8. Surface Water Pa	ithway
Type of Surface Water Draining Site and 15 M	Ailes Downstream (check all that	Shortest Overland Distance From Any Source to Surface Water:
apply):		,,
Stream V River		685 Eest
Is There a Suspected Release to Surface Wat	er ¹ :	Site is Located in:
		Annual - 10 yr Floodplain
✓ Yes		>10yr - 100yr Floodplain
		$\square > 500yr$ Floodplain
Dein bin e Michael Intelle Leaster d'Alene the Com	for a Materia Missistica Data	List All Course dams Target Deinking Mister Intelses
Drinking water intake located Along the Sur	face water Migration Path:	List All Secondary Target Drinking Water Intakes:
🗌 Yes		
✓ No		Name: Water Body: Flow (cfs): Population Served:
Have Primary Target Drinking Water Intakes	Been Identified:	
☐ Yes If Yes, Distant	ce to Nearest Drinking	
water intake	: Miles*	
If Yes, Enter Population Served by Target Inte	ake:	
		Total within 15 Miles ⁴
People ⁴		
Fisheries Located Along the Surface Mater M	ligration Dath:	
risheries Located Along the Surface Water M	ngration Patri:	List All Secondary Target Fisheries*:
□ Yes ☑ No IT Yes, Distance	e to nearest Fishery: Miles	Water Body/ Fishery Name : Flow (cfs):
Have Primary Target Fisheries Reen Identifie	Miles	· ، ، ، ، ، ، ، ، ، ، ، ، ، ، ، ،
	<u>.</u>	
Li Yes V No		

	8. Surfac	ce Water Pathway	y (continue	ed)	
Wetlands Located Along the Surface Water I	Vigration Path:	Other Sensitive Environments Located Along the Surface Water Migration Path:			
✓ Yes □ No		✓ Yes If Yes, Distance to Nearest Sensitive Environment: No <u>1,280</u> feet			invironment:
Have Primary Target Wetlands Been Identif	Have Primary Tar	get Sensitive	e Environments Been Identified:		
✓ Yes □ No			✓ Yes □ No		
List All Wetlands:		List All Sensitive E	Invironment	S ¹¹ :	
Water Body : Flow (cfs): Frontage miles:		<u>Water Body</u> :	Flow (cfs):	Sensitive Environment Type:	
Baznan River		multiple wetlands			
	9.	I . Soil Exposure Pa	thway		
Are People Occupying Residence or Attending School or Daycare on or Within 200 Feet of Area of Known or Suspected Contamination:	Number of Wo □ None ☑ 1 - 1 □ 101 - □ > 1,0	rkers Onsite ⁴ : 00 - 1,000 000	Have Terre Within 200 Contamina	estrial Sensitive Environments B D Feet of Areas of Known or Susp ation:	een Identified on or pected
☐ Yes ✓ No	Population Wit	hin 1 Mile	If Yes, Lis	☑ № t Each Terrestrial Sensitive Envir	ronment ⁵ :
If Yes, Enter Total Residential Population: People ² Population Wit Population Wit <u>16,416 (4 r</u> People ⁷		niles) *Refer to PA Table 7 for environment types			
		10. Air Pathwa	ay		
Is there a Suspected Release to Air^1 :		Wetlands Located	Within 4 M	iles of the Site ⁶ :	
Enter Total Population on or Within:		✓ Yes If Yes, How Many Acres: <u>unknown</u> Acres			25
Onsite		Other Sensitive En	vironments	Located Within 4 Miles of the S	ite:
0-1/4 Mile		✓ Yes □ No			
>1/4-1/2 Mile		List All Sensitive Environments Within 1/2 Mile of the Site ⁶ :			
>1/2-1 Mile		Distance: Sens	itive Environ	ment Type/Wetlands Area (acres):	
>1-2 Miles		Onsite			_
>2-3 Miles		0-1/4 Mile <u>Ba</u>	nana River		
>3-4 Miles		>1/4-1/2 Mile	ultieple wetl	ands	-
Total Within 4 Miles ³⁻⁵ <u>16,416</u>		*Refer to PA Table 10 fo	or calculations o	n air pathway exposures	

					Identification	
Potential Hazardous Waste Site Preliminary Assessment Form					State:	CERCLIS #:
			-		CERCLIS Discovery I	Date:
		1.	General Site Infor	mation		
Name: Hangar 750		Street Address	5: NA			
City:		State: FL	Zip Code:32925	County:Brevard	Co. Code:12009	Cong. Dist:8th
Latitude:	Longitude:	Approximate A	Area of Site:	Status of Site:		
		Acres		Active 🗌 Not Specif	īed	
			Square Ft	🗌 Inactive 🛛 🗌 NA (GW p	lume, etc.)	
Site Name: Hangar 750						
Site Description: Hanga	r 750 is located in the	e northern porti	on of Patrick AFB, so	outheast of the intersection	on of Rescue Road ar	nd Redstone Road.
Hangar 750 was constru	ucted in 1943 and wa	s initially equipp	bed with a wet fire s	prinkler system. In 2000,	Hangar 750 was retr	ofitted with an
AFFF fire suppression sy	/stem and operated ι	until 2007. In 20	07, the fire suppres	sion system was retrofitte	ed to a HEF system. V	When the hangar
was equipped with an A	FFF fire suppression	system. a 1.200	gallon poly AST cha	arged with 3% AFFF suppli	ied the hangar via un	derground piping.
		o youchin (a 2)200	Serier perf / er ene			
		2. 0	wner/Operator In	formation		
Owner: Patrick AFB			Operator: Same as	s "owner"		
Street Address:			Street Address:			
City:			City:			
State: FL	Zip Code:32925	Telephone:	State:	Zip Code:	Telephone:	
Type of Ownership:	•		Type of Ownership):		
Private	🗌 County		Private	County		
Federal Agency	🗌 Municipal		Federal Agency	Municipal		
Name: <u>DOD</u>	Not Specif	ied	Name:	Not Specified		
	Other		☐ State ☐ Other			
		3. 9	Site Evaluator Info	ormation		
Name of Evaluator: Joh	n Sandoval	Agency/Organ	ization: HydroGeoLo	ogic, Inc.	Date Prepared:07/0	06/15
Street Address:404 E. R	amsey Road, Ste. 210)	City:San Antonio		State:Texas	
Name of EPA or State A	gency Contact:NA		Street Address:			
City: State:		Telephone:				
		4. Site	Disposition (for El	PA use only)		
Emergency Response/R	emoval Assessment		CERCLIS Recomme	endation:	Signature:	
Recommendation:			Higher Priority	SI	Name (typed):	
	Yes			01	Name (typed):	
	🗌 No				Position:	
Date:			Other:	_		
			Date:			

5. General Site Characteristics					
Predominant Land Use Wit	thin 1 Mile of Site (check all that	Site Setting:		Years of Operation:	
apply):		_			
		🗌 Urban		Beginning Year <u>1943</u>	
			an		
Residential	DOD Excility:			Ending Yearpresent	
Forest/Fields	DOE				
	U Other			Unknown	
Type of Site Operations (ch	neck all that apply):			Waste Generated:	
Manufacturing (must check su	ubcategory)	Retail		✓ Onsite	
Lumber and Wood Prod	ucts	Recycling		Offsite	
Inorganic Chemicals		Junk/Salvage Yard		Onsite and Offsite	
Plastic and/or Rubber Pl	roducts	Municipal Landfill			
Paints, Varnishes		Other Landfill		Waste Deposition Authorized By:	
Industrial Organic Chem	nicals	✓ DOD		Present Owner	
Agricultural Chemicals				Former Owner	
Miscellaneous Chemical	Products	DOI	*	Present & Former Owner	
Primary Metals			ty	Unauthorized	
Metal Coating, Plating, E		Treatment Stor	age or Disposal	Unknown	
Eabricated Structural M) Nal Products	Large Quantity 0	Generator	Waste Accessible to the Public:	
		Small Quantity G	Senerator		
Other Manufacturing		Subtitle D			
		 Municipal			
		Industrial			
Metals		Converter"			
		"Protective Filer"		Distance to Nearest Dwelling, School,	
Oil and Gas		🔲 "Non-or Late File	er"	or Workplace:	
Non-metallic Minerals		Note Specified			
		Other		_on-site Feet	
6. Waste Characteristic	s Information	for WC Score)		(Refer to PA Table 1	
Source Type:	Source Waste Quantity:	Tier*:	General Type of Waste	(check all that	
(check all that apply)	(include unit)		apply):		
			Metals	Pesticides/Herbicides	
			Organics	Acids/Bases	
			Inorganics	Oily Waste	
			Solvents	Municipal Waste	
			Paints/Pigments	Mining Waste	
Scrap Metal or Junk Pile			Laboratory/Hospital Waste		
Tailings Pile			Radioactive Waste	Ucata Ucata Ucata	
Trash Pile (open drum)				vaste	
Land Treatment					
Contaminated GW Plume			Physical State of Waste	as Deposited (check all that apply):	
(unidentified source)					
Contaminated SW/Sediment			Solid		
(unidentified source)					
			Powder		
			 ✓ Liquid		
*C=Constituent \\/_\	astestream V-Volume A-Aroa		Gas		
"C=Constituent, W=W	astestream, v=volume, A=Area				

	7. Ground Water Pa	athway
Is Ground Water Used for Drinking Within 4	Is There a Suspected Release to	List Secondary Target Population Served by Ground Water
Miles:	Ground Water ¹ :	Withdrawn From:
Voc.		
l v res	I vo	
		0 - 1/4 Mile
If Yes, Distance to nearest Drinking		
Well:	Have Primary Target Drinking	>1/4 - 1/2 Mile
<u>5.06</u> miles	Water Wells Been Identified:	
	Water Wens been dentined.	>1/2 - 1 Mile
Type of Drinking Water Wells Within 4		
IVIIIes	l ✓ Yes	>1 - 2 Mile
(check all that apply):	🗌 No	
✓ Municipal	If Vac Enter Drimony Target	>2 - 3 Mile
	n Yes, Enter Primary Target	>2 4 Mile
	Population:	>3 - 4 Mile
		Total Within 4 Miles ⁴ 17 124
Depth to Shallowest Aquifer:	Nearest Designated Wellhead	
_4-5 Feet	Protection Area ⁶ :	
Karst Terrain/Aquifer Present:	Underlies Site	*Lise population #s for PA Table 2
Karse Ferrain, Aquirer Fresenti	\square >0-4 Miles	*Note nearest well for #5 on GW Pathway Scoresheet
✓ Yes	None Within 4 Miles	
□ No		
	8. Surface Water Pa	nthway
Type of Surface Water Draining Site and 15 N	/iles Downstream (check all that	Shortest Overland Distance From Any Source to Surface Water:
apply):		
		1.042 Feet
		_ <u></u> reet Miles
	er	
Is There a Suspected Release to Surface Wat	er ¹ :	Site is Located in:
		Annual - 10 yr Floodplain
Yes		>10yr - 100yr Floodplain
✓ No		>100yr - 500yr Floodplain
		└ >500yr Floodplain
Drinking Water Intake Located Along the Sur	face Water Migration Path:	List All Secondary Target Drinking Water Intakes:
_		
Yes		
I NO		<u>Name</u> : <u>Water Body</u> : <u>Flow (cfs)</u> : <u>Population Served</u> :
Have Primary Target Drinking Water Intakes	Been Identified:	
	to Neorost Driving	
If Yes, Distance		
water intake		
If Yes, Enter Population Served by Target Inta	ake:	
		Total within 15 Miles ⁴
People ⁴		
Fisheries Located Along the Surface Water M	ligration Path:	List All Secondary Target Fisheries ¹⁰ :
If Yes, Distance	e to Nearest Fishery:	Water Body/ Fishery Name : Flow (cfs):
	Miles	
Have Primary Target Fisheries Been Identifie	d:]

	8. Surfac	ce Water Pathway	ay (continued)			
Wetlands Located Along the Surface Water N	Vigration Path:	Other Sensitive En	nvironments Located Along the Surface Water Migration Path:			
☐ Yes ✔ No		☐ Yes ✔ No	 ☐ Yes ☐ Yes If Yes, Distance to Nearest Sensitive Environment: ☑ No feet 			
Have Primary Target Wetlands Been Identif	ied:	Have Primary Tar	arget Sensitive Environments Been Identified:			
☐ Yes ☑ No			☐ Yes ✓ No			
List All Wetlands:		List All Sensitive E	Environments ¹¹ :			
<u>Water Body</u> : <u>Flow (cfs)</u> : <u>Frontage miles:</u>		Water Body :	Flow (cfs): Sensitive Environment Type:			
9. Soil Exposure Pathway						
Are People Occupying Residence or Attending School or Daycare on or Within 200 Feet of Area of Known or Suspected Contamination:		rkers Onsite ⁴ : 00 - 1,000 000	Have Terrestrial Sensitive Environments Been Identified on or Within 200 Feet of Areas of Known or Suspected Contamination:			
Yes			✓ No			
✓ No	Deputation \A/it	hin 1 Nailas	If Yes, List Each Terrestrial Sensitive Environment ⁵ :			
If Yes, Enter Total Residential Population: People ² Population Wit		<u>niles)</u>	*Refer to PA Table 7 for environment types			
		10. Air Pathwa	vay			
Is there a Suspected Release to Air ¹ : Yes No Enter Total Population on or Within:		Wetlands Located	d Within 4 Miles of the Site ⁶ : If Yes, How Many Acres: <u>unknown</u> Acres			
Onsite		Other Sensitive Environments Located Within 4 Miles of the Site:				
0-1/4 Mile		✓ Yes □ No				
>1/4-1/2 Mile		List All Sensitive Environments Within 1/2 Mile of the Site ⁶ :				
>1/2-1 Mile		Distance: Sensi	sitive Environment Type/Wetlands Area (acres):			
>1-2 Miles		Onsite				
>2-3 Miles		0-1/4 Mile <u>Bar</u>	anana River			
>3-4 Miles		>1/4-1/2 Mile	multieple wetlands			
Total Within 4 Miles ³⁻⁵ <u>17,124</u>		*Refer to PA Table 10 fo	for calculations on air pathway exposures			
					Identification	
--	--	---	---	---	------------------------	----------------------------------
Potential Haz	ardous Wast	e Site Pre	liminary Ass	sessment Form	State:	CERCLIS #:
			•		CERCLIS Discovery D	ate:
		1.	General Site Infor	rmation		
Name: Hangar 751		Street Address	:: NA			
City:		State: FL	Zip Code:32925	County:Brevard	Co. Code:12009	Cong. Dist:8th
Latitude:	Longitude:	Approximate A Acres	Area of Site: Square Ft	Status of Site: Active Not Specif Inactive NA (GW p	ïed lume, etc.)	
Site Name: Hangar 751						
Hangar 751 was constru 1,200 gallon poly AST th the environment from t	acted in 1945 and is c at is located in the B he containment syste	urrently equipp uilding 705 Pum em at Hangar 75	ed with an AFFF fire op House. There was 51.	s no available documenta	tion or evidence of an	% AFFF from a AFFF release to
		2. 0\	wner/Operator In	formation		
Owner: Patrick AFB		Operator: Same as "owner"				
Street Address:		Street Address:				
City:		City:				
State: FL	Zip Code:32925	Telephone:	State:	Zip Code:	Telephone:	
Type of Ownership:			Type of Ownership):		
 Private Federal Agency Name: <u>DOD</u> State Indian 	County Municipal Not Specif Other	ied	 Private Federal Agency Name: State Indian 	County Municipal Not Specified Other		
		3. 9	I Site Evaluator Info	ormation		
Name of Evaluator: Joh	n Sandoval	Agency/Organi	ization: HydroGeoLo	ogic, Inc.	Date Prepared:07/0	6/15
Street Address:404 E. Ra	amsey Road, Ste. 210)	City:San Antonio		State:Texas	
Name of EPA or State A	gency Contact:NA		Street Address:			
City:		State:		Telephone:		
		4. Site	Disposition (for El	PA use only)		
Emergency Response/R	emoval Assessment		CERCLIS Recomme	endation:	Signature:	
Recommendation:	Yes		Higher Priority	SI SI	Name (typed):	
Date:	No		RCRA Other: Date:	-	Position:	

5. General Site Characteristics - NA				
Predominant Land Use Wit	thin 1 Mile of Site (check all that	Site Setting:		Years of Operation:
apply):				
		🗌 Urban		Beginning Year
	Mining Other Federal	□ Suburba	an	
Residential	DOD Facility:			Ending Year
Forest/Fields				
	Other			Unknown
Type of Site Operations (ch	neck all that apply):			Waste Generated:
Manufacturing (must check su	ubcategory)	Retail		Onsite
Lumber and Wood Prod	ucts	Recycling		Offsite
Inorganic Chemicals		🗌 Junk/Salvage Yard		Onsite and Offsite
Plastic and/or Rubber P	roducts	Municipal Landfill		
Paints, Varnishes		Other Landfill		Waste Deposition Authorized By:
Industrial Organic Chem	nicals			Present Owner
Agricultural Chemicals	Deciderate			Former Owner
	Products	Other Federal Facili	tv	Present & Former Owner
Metal Coating Plating	Ingraving		.,	
Metal Forging, Stamping	1	Treatment, Stora	age, or Disposal	
Fabricated Structural Me	etal Products	Large Quantity C	Generator	Waste Accessible to the Public:
Electronic Equipment		Small Quantity G	Generator	
Other Manufacturing		Subtitle D		Yes
		Municipal		🗌 No
Metals				
		U "Converter"		Distance to Nearest Dwelling, School,
Oil and Gas			البر	or Workplace
Non-metallic Minerals				
				E.u.
				Feet
6. Waste Characteristi	cs Information - NA			(Refer to PA
		Table 1 for WC Sco	ore)	
Source Type:	Source Waste Quantity:	Tier*:	General Type of Waste	(check all that
(check all that apply)	(include unit)		apply):	_
			Metals	Pesticides/Herbicides
Surface Impoundment				Acids/Bases
Drums				
Tanks and Non-Dum Containe	rs		Paints/Pigments	
Chemical Waste Pile			Laboratory/Hospital Waste	
Scrap Metal or Junk Pile			Radioactive Waste	☐ Other
Tailings Pile			Construction/Demolition W	Vaste
Irash Pile (open drum)				
			Dhusical State of Misste	
			Physical State of Waste	as Deposited (check all that apply):
Contaminated SW/Sediment				
(unidentified source)				
Contaminated Soil				
			Gas	
*C=Constituent, W=W	/astestream, V=Volume, A=Area			

7. Ground Water Pathway - NA			
Is Ground Water Used for Drinking Within 4	Is There a Suspected Release to	List Secondary Target Population Served by Ground Water	
Miles:	Ground Water ¹	Withdrawn From:	
		0 - 1/4 Mile	
If Yes, Distance to nearest Drinking			
Well:	Have Primary Target Drinking	- >1/4 - 1/2 Mile	
miles	Water Wells Been Identified:	>1/2 1 Mile	
Type of Drinking Water Wells Within 4		>1/2 - 1 Mile	
Miles		>1 - 2 Mile	
(check all that apply):			
	L No	>2 - 3 Mile	
	If Yes, Enter Primary Target		
□ None	Population:	>3 - 4 Mile	
	People ³		
Depth to Shallowest Aquifer:	Nearest Designated Wellhead	Total Within 4 Miles ⁴	
Feet	Protection Area ⁶		
Karst Terrain/Aquifer Present:		*Use population #s for PA Table 2	
	□ >0-4 Miles	[↑] Note nearest well for #5 on GW Pathway Scoresheet	
	8. Surface Water Path	nway - NA	
Type of Surface Water Draining Site and 15 M	Ailes Downstream (check all that	Shortest Overland Distance From Any Source to Surface Water:	
apply):		,,,,,	
		Foot	
	nd 🗌 Lake	Feet	
	er		
Is There a Suspected Release to Surface Wat	er ¹ :	Site is Located in:	
		🗌 Annual - 10 yr Floodplain	
Yes		>10yr - 100yr Floodplain	
L No		>100yr - 500yr Floodplain	
		Solver Floodplain	
Drinking Water Intake Located Along the Sur	face Water Migration Path:	List All Secondary Target Drinking Water Intakes:	
□ Yes			
		Name: Water Body: Flow (cfs): Population Served:	
Have Primary Target Drinking Water Intakes	Been Identified:		
□ Yes If Yes, Distant	ce to Nearest Drinking		
□ No Water Intake	: Miles ⁶		
If Yes, Enter Population Served by Target Int	ake:		
in res, Enter ropulation served by ranget inte	are.		
People ⁴		Total within 15 Miles ⁴	
Fisheries Located Along the Surface Water N	ligration Path:	List All Secondary Target Fisheries ¹⁰ :	
If Yes. Distance	e to Nearest Fisherv:	Water Body/ Fishery Name : Flow (cfs):	
	Miles		
Have Primary Target Fisheries Been Identifie	d:		

	8. Surface	Water Pathway	y (continued) - NA	
Wetlands Located Along the Surface Water N	Vigration Path:	Other Sensitive E	Environments Located Along the Surface Water Migration Path:	:
☐ Yes ☐ No		☐ Yes ☐ No	If Yes, Distance to Nearest Sensitive Environment: feet	
Have Primary Target Wetlands Been Identif	ied:	Have Primary Ta	Target Sensitive Environments Been Identified:	
Yes No			☐ Yes ☐ No	
List All Wetlands:		List All Sensitive	e Environments ¹¹ :	
Water Body : Flow (cfs): Frontage miles:		<u>Water Body</u> :	Flow (cfs): Sensitive Environment Type:	
	0.6			
Are Beenle Occupying Residence or	9.50	bil Exposure Pat	thway - NA	or
Attending School or Daycare on or Within 200 Feet of Area of Known or Suspected Contamination:	Number of Wo	rkers Onsite*: 900 - 1,000 000	Within 200 Feet of Areas of Known or Suspected Contamination:	Ur
Yes				
No			If Yes, List Each Terrestrial Sensitive Environment ⁵ :	
If Yes, Enter Total Residential Population: People ²	Population Within 1 Mile: People ⁷		*Refer to PA Table 7 for environment types	-
		10. Air Pathw	way	
Is there a Suspected Release to Air ¹ :		Wetlands Locate	ed Within 4 Miles of the Site ⁶ :	
Enter Total Population on or Within:		☐ Yes ☐ No	If Yes, How Many Acres: Acres	
Onsite		Other Sensitive E	Environments Located Within 4 Miles of the Site:	
0-1/4 Mile			☐ Yes ☐ No	
>1/4-1/2 Mile		List All Sensitive	Environments Within 1/2 Mile of the Site ⁶ :	
>1/2-1 Mile		Distance: Ser	ensitive Environment Type/Wetlands Area (acres):	
>1-2 Miles		Onsite		
>2-3 Miles		0-1/4 Mile		
>3-4 Miles		>1/4-1/2 Mile		
Total Within 4 Miles ³⁻⁵		*Refer to PA Table 10	L0 for calculations on air pathway exposures	

¹⁻¹¹ Refers to question number on the PA scoresheet for each particular pathway

					Identification	
Potential Haz	ardous Wast	e Site Pre	liminary Ass	essment Form	State:	CERCLIS #:
			-		CERCLIS Discovery Da	ate:
		1.	General Site Infor	mation		
Name: Hangar 985		Street Address	5: NA			
City:		State: FL	Zip Code:32925	County:Brevard	Co. Code:12009	Cong. Dist:8th
Latitude:	Longitude:	Approximate A	Area of Site:	Status of Site:		
		Acres		Active Not Specif	ied	
			Square Ft	Inactive NA (GW p	lume, etc.)	
Site Name: Hangar 985						
Site Description: Hanga	r 985 is located in the	e eastern centra	I portion of Patrick A	AFB. Hangar 985 was cons	structed in 1953 and is	scurrently
equipped with an AFFF	fire suppression syste	em and four low	/ level turrets. The h	angar is supplied 3% AFF	F from an 800 gallon A	ST located in the
hangar mechanical roor	m. There is no contair	nment system a	ssociated with Hang	ar 985. There have been	no reported or docum	ented releases of
AFFF at Hangar 985 .						
		2. 0	wner/Operator Inf	ormation		
Owner: Patrick AFB		Operator: Same as	"owner"			
Street Address:		Street Address:				
City:		City:				
State: FL	Zip Code:32925	Telephone:	State:	Zip Code:	Telephone:	
Type of Ownership:	-		Type of Ownership	:		
Private	🗌 County		Private	County		
✓ Federal Agency	🗌 Municipal		Federal Agency	Municipal		
Name: <u>DOD</u>	Not Specifi	ed	Name: Not Specified			
	Other		☐ Indian			
		3. 9	Site Evaluator Info	rmation		
Name of Evaluator: Joh	n Sandoval	Agency/Organ	ization: HydroGeoLo	ogic, Inc.	Date Prepared:07/06	/15
Street Address:404 E. R	amsey Road, Ste. 210)	City:San Antonio		State:Texas	
Name of EPA or State A	gency Contact:NA		Street Address:			
City:		State:	ļ	Telephone:		
		4. Site	Disposition (for EP	PA use only)	-	
Emergency Response/R	emoval Assessment		CERCLIS Recomme	ndation:	Signature:	
Recommendation:	_		Higher Priority S	51 	Name (typed):	
	Yes		□ NFRAP			
					Position:	
Date:			Date:	_		

5. General Site Characteristics - NA				
Predominant Land Use Wit	hin 1 Mile of Site (check all that	Site Setting:		Years of Operation:
apply):				
		🗌 Urban		Beginning Year
	Mining Other Federal	Suburba	an	
Residential	DOD Facility:			Ending Year
Forest/Fields [DOE			
	Other			Unknown
Type of Site Operations (ch	eck all that apply):	-		Waste Generated:
Manufacturing (must check su	bcategory)	Retail		Onsite
Lumber and Wood Produ	icts	Recycling		Offsite
Inorganic Chemicals		🗌 Junk/Salvage Yard		Onsite and Offsite
Plastic and/or Rubber Press	oducts	Municipal Landfill		
Paints, Varnishes		Other Landfill		Waste Deposition Authorized By:
Industrial Organic Chemi	icals			Present Owner
Agricultural Chemicals				Former Owner
	Products	Other Federal Facili	tv	Present & Former Owner
Metal Coating, Plating, F	ngraving		·)	
Metal Forging, Stamping		Treatment, Stora	age, or Disposal	
Fabricated Structural Me	tal Products	Large Quantity C	Generator	Waste Accessible to the Public:
Electronic Equipment		Small Quantity G	Generator	
Other Manufacturing		Subtitle D		Yes
Mining		Municipal		🗌 No
Metals		Industrial		
		U "Converter"		Distance to Nearest Dwelling, School,
\Box Oil and Gas			البر	or Workplace
Non-metallic Minerals				
		Other		E.u.
				Feet
6. Waste Characteristic	s Information - NA			(Refer to PA
		Table 1 for WC Sco	ore)	
Source Type:	Source Waste Quantity:	Tier*:	General Type of Waste	(check all that
(check all that apply)	(include unit)		apply):	_
			Metals	Pesticides/Herbicides
Surface Impoundment			Organics	Acids/Bases
Drums				Oily Waste
Tanks and Non-Dum Container	·s		Solvenis Paints/Pigments	
Chemical Waste Pile			Laboratory/Hospital Waste	
Scrap Metal or Junk Pile			Radioactive Waste	Other
Tailings Pile			Construction/Demolition W	Vaste
Trash Pile (open drum)				
			Physical State of Waste	as Deposited (check all that apply):
(unidentified source)				
(unidentified source)				
Contaminated Soil				
Other				
□ No Sources				
*C=Constituent, W=Wa	astestream, V=Volume, A=Area			

7. Ground Water Pathway - NA			
Is Ground Water Used for Drinking Within 4	Is There a Suspected Release to	List Secondary Target Population Served by Ground Water	
Miles:	Ground Water ¹	Withdrawn From:	
		0 - 1/4 Mile	
If Yes, Distance to nearest Drinking			
Well:	Have Primary Target Drinking	- >1/4 - 1/2 Mile	
miles	Water Wells Been Identified:	>1/2 1 Mile	
Type of Drinking Water Wells Within 4		>1/2 - 1 Mile	
Miles		>1 - 2 Mile	
(check all that apply):			
	L No	>2 - 3 Mile	
	If Yes, Enter Primary Target		
□ None	Population:	>3 - 4 Mile	
	People ³		
Depth to Shallowest Aquifer:	Nearest Designated Wellhead	Total Within 4 Miles ⁴	
Feet	Protection Area ⁶		
Karst Terrain/Aquifer Present:		*Use population #s for PA Table 2	
	□ >0-4 Miles	[↑] Note nearest well for #5 on GW Pathway Scoresheet	
	8. Surface Water Path	nway - NA	
Type of Surface Water Draining Site and 15 M	Ailes Downstream (check all that	Shortest Overland Distance From Any Source to Surface Water:	
apply):		,,,,,	
		Foot	
	nd 🗌 Lake	Feet	
	er		
Is There a Suspected Release to Surface Wat	er ¹ :	Site is Located in:	
		🗌 Annual - 10 yr Floodplain	
Yes		>10yr - 100yr Floodplain	
L No		>100yr - 500yr Floodplain	
		Solver Floodplain	
Drinking Water Intake Located Along the Sur	face Water Migration Path:	List All Secondary Target Drinking Water Intakes:	
□ Yes			
		Name: Water Body: Flow (cfs): Population Served:	
Have Primary Target Drinking Water Intakes	Been Identified:		
□ Yes If Yes, Distant	ce to Nearest Drinking		
□ No Water Intake	: Miles ⁶		
If Yes, Enter Population Served by Target Int	ake:		
in res, Enter ropulation served by ranget into	are.		
People ⁴		Total within 15 Miles ⁴	
Fisheries Located Along the Surface Water N	ligration Path:	List All Secondary Target Fisheries ¹⁰ :	
If Yes. Distance	e to Nearest Fisherv:	Water Body/ Fishery Name : Flow (cfs):	
	Miles		
Have Primary Target Fisheries Been Identifie	d:		

	8. Surface	Water Pathway	y (continued) - NA	
Wetlands Located Along the Surface Water N	Aigration Path:	Other Sensitive E	Environments Located Along the Surface Water Migration Path	:
☐ Yes ☐ No		☐ Yes ☐ No	If Yes, Distance to Nearest Sensitive Environment: feet	
Have Primary Target Wetlands Been Identif	ied:	Have Primary Ta	Farget Sensitive Environments Been Identified:	
☐ Yes ☐ No			Yes No	
List All Wetlands:		List All Sensitive	e Environments ¹¹ :	
Water Body : Flow (cfs): Frontage miles:		Water Body :	Flow (cfs): Sensitive Environment Type:	
	0.6	Deti		
Are Beenle Occupying Residence or	9.50	bil Exposure Pati	thway - NA	or
Attending School or Daycare on or Within 200 Feet of Area of Known or Suspected Contamination:	Number of Wo	rkers Onsite*: 00 - 1,000 000	Within 200 Feet of Areas of Known or Suspected Contamination:	or
Tes Yes				
No			If Yes, List Each Terrestrial Sensitive Environment ⁵ :	
If Yes, Enter Total Residential Population: People ²	Population Within 1 Mile: People ⁷		*Refer to PA Table 7 for environment types	-
		10. Air Pathw	way	
Is there a Suspected Release to Air ¹ :		Wetlands Located	ed Within 4 Miles of the Site ⁶ :	
Enter Total Population on or Within:		☐ Yes ☐ No	If Yes, How Many Acres: Acres	
Onsite		Other Sensitive E	Environments Located Within 4 Miles of the Site:	
0-1/4 Mile			☐ Yes ☐ No	
>1/4-1/2 Mile		List All Sensitive I	Environments Within 1/2 Mile of the Site ⁶ :	
>1/2-1 Mile		Distance: Sen	ensitive Environment Type/Wetlands Area (acres):	
>1-2 Miles		Onsite		
>2-3 Miles		0-1/4 Mile		
>3-4 Miles		>1/4-1/2 Mile		
Total Within 4 Miles ³⁻⁵		*Refer to PA Table 10	L0 for calculations on air pathway exposures	

¹⁻¹¹ Refers to question number on the PA scoresheet for each particular pathway

					Identification	
Potential Haz	ardous Wast	e Site Pre	eliminary Ass	sessment Form	State:	CERCLIS #:
					CERCLIS Discovery D	ate:
		1.	General Site Infor	rmation		
Name: Hangar 986		Street Address	s: NA			
City:		State: FL	Zip Code:32925	County:Brevard	Co. Code:12009	Cong. Dist:8th
Latitude:	Longitude:	Approximate /	Area of Site:	Status of Site:		•
		Acres		Active Not Specif	ied	
			Square Ft	🗌 Inactive 🛛 🗌 NA (GW p	lume, etc.)	
Site Name: Hangar 986	1					
Site Description: Hanga	r 986 is located in the	eastern centra	al portion of Patrick	AFB. Hangar 986 was cons	structed in 1953 and is	s currently
equipped with a deluge	e fire suppression syst	em. The hanga	r has always been eo	quipped with a deluge fire	suppression system a	ind has never
been equipped with an	AFFF fire suppression	n system.	·			
		2. 0	wner/Operator In	formation		
Owner: Patrick AFB			Operator: Same as	s "owner"		
Street Address:		Street Address:				
City:			City:			
State: FL	Zip Code:32925	Telephone:	State:	Zip Code:	Telephone:	
Type of Ownership:	-1		Type of Ownership	D:		
Private	🗌 County		Private	County		
Federal Agency	🗌 Municipal		Federal Agency	Municipal		
Name: <u>DOD</u>	🗌 Not Specif	ied	Name: Not Specified			
	Other		☐ Indian			
		3. :	Site Evaluator Info	ormation		
Name of Evaluator: Joh	ın Sandoval	Agency/Organ	ization: HydroGeoLo	ogic, Inc.	Date Prepared:07/06	5/15
Street Address:404 E. R	amsey Road, Ste. 210)	City:San Antonio		State:Texas	
Name of EPA or State A	Agency Contact:NA		Street Address:			
City:		State:		Telephone:		
		4. Site	Disposition (for El	PA use only)	-	
Emergency Response/F	Removal Assessment		CERCLIS Recomme	endation:	Signature:	
Recommendation:			Higher Priority	SI	Name (typed):	
	Yes			51	ivame (typed):	
	🗌 No				Position:	
Date:			Other:			
			Date:			

5. General Site Characteristics - NA				
Predominant Land Use Wit	hin 1 Mile of Site (check all that	Site Setting:		Years of Operation:
apply):				
		🗌 Urban		Beginning Year
	Mining Other Federal	Suburba	an	
Residential	DOD Facility:			Ending Year
Forest/Fields [DOE			
	Other			Unknown
Type of Site Operations (ch	eck all that apply):	-		Waste Generated:
Manufacturing (must check su	bcategory)	Retail		Onsite
Lumber and Wood Produ	icts	Recycling		Offsite
Inorganic Chemicals		🗌 Junk/Salvage Yard		Onsite and Offsite
Plastic and/or Rubber Press	oducts	Municipal Landfill		
Paints, Varnishes		Other Landfill		Waste Deposition Authorized By:
Industrial Organic Chemi	icals			Present Owner
Agricultural Chemicals				Former Owner
	Products	Other Federal Facili	tv	Present & Former Owner
Metal Coating, Plating, F	ngraving		·)	
Metal Forging, Stamping		Treatment, Stora	age, or Disposal	
Fabricated Structural Me	tal Products	Large Quantity C	Generator	Waste Accessible to the Public:
Electronic Equipment		Small Quantity G	Generator	
Other Manufacturing		Subtitle D		Yes
Mining		Municipal		🗌 No
Metals		Industrial		
		U "Converter"		Distance to Nearest Dwelling, School,
\Box Oil and Gas			البر	or Workplace
Non-metallic Minerals				
		Other		E.u.
				Feet
6. Waste Characteristic	s Information - NA			(Refer to PA
		Table 1 for WC Sco	ore)	
Source Type:	Source Waste Quantity:	Tier*:	General Type of Waste	(check all that
(check all that apply)	(include unit)		apply):	_
			Metals	Pesticides/Herbicides
Surface Impoundment				Acids/Bases
Drums				Oily Waste
Tanks and Non-Dum Container	·s		Solvenis Paints/Pigments	
Chemical Waste Pile			Laboratory/Hospital Waste	
Scrap Metal or Junk Pile			Radioactive Waste	Other
Tailings Pile			Construction/Demolition W	Vaste
Trash Pile (open drum)				
			Physical State of Waste	as Deposited (check all that apply):
(unidentified source)				
(unidentified source)				
Contaminated Soil				
Other				
□ No Sources				
*C=Constituent, W=Wa	astestream, V=Volume, A=Area			

7. Ground Water Pathway - NA			
Is Ground Water Used for Drinking Within 4	Is There a Suspected Release to	List Secondary Target Population Served by Ground Water	
Miles:	Ground Water ¹	Withdrawn From:	
		0 - 1/4 Mile	
If Yes, Distance to nearest Drinking			
Well:	Have Primary Target Drinking	- >1/4 - 1/2 Mile	
miles	Water Wells Been Identified:	>1/2 1 Mile	
Type of Drinking Water Wells Within 4		>1/2 - 1 Mile	
Miles		>1 - 2 Mile	
(check all that apply):			
	L No	>2 - 3 Mile	
	If Yes, Enter Primary Target		
□ None	Population:	>3 - 4 Mile	
	People ³		
Depth to Shallowest Aquifer:	Nearest Designated Wellhead	Total Within 4 Miles ⁴	
Feet	Protection Area ⁶		
Karst Terrain/Aquifer Present:		*Use population #s for PA Table 2	
	□ >0-4 Miles	[↑] Note nearest well for #5 on GW Pathway Scoresheet	
	8. Surface Water Path	nway - NA	
Type of Surface Water Draining Site and 15 M	Ailes Downstream (check all that	Shortest Overland Distance From Any Source to Surface Water:	
apply):		,,,,,	
		Foot	
	nd 🗌 Lake	Feet	
	er		
Is There a Suspected Release to Surface Wat	er ¹ :	Site is Located in:	
		🗌 Annual - 10 yr Floodplain	
Yes		>10yr - 100yr Floodplain	
L No		>100yr - 500yr Floodplain	
		Solver Floodplain	
Drinking Water Intake Located Along the Sur	face Water Migration Path:	List All Secondary Target Drinking Water Intakes:	
□ Yes			
		Name: Water Body: Flow (cfs): Population Served:	
Have Primary Target Drinking Water Intakes	Been Identified:		
□ Yes If Yes, Distant	ce to Nearest Drinking		
□ No Water Intake	: Miles ⁶		
If Yes, Enter Population Served by Target Int	ake:		
in res, Enter ropulation served by ranget into	are.		
People ⁴		Total within 15 Miles ⁴	
Fisheries Located Along the Surface Water N	ligration Path:	List All Secondary Target Fisheries ¹⁰ :	
If Yes. Distance	e to Nearest Fisherv:	Water Body/ Fishery Name : Flow (cfs):	
	Miles		
Have Primary Target Fisheries Been Identifie	d:		

	8. Surface	Water Pathway	y (continued) - NA	
Wetlands Located Along the Surface Water N	Aigration Path:	Other Sensitive E	Environments Located Along the Surface Water Migration Path	:
☐ Yes ☐ No		☐ Yes ☐ No	If Yes, Distance to Nearest Sensitive Environment: feet	
Have Primary Target Wetlands Been Identif	ied:	Have Primary Ta	Farget Sensitive Environments Been Identified:	
☐ Yes ☐ No			Yes No	
List All Wetlands:		List All Sensitive	e Environments ¹¹ :	
Water Body : Flow (cfs): Frontage miles:		Water Body :	Flow (cfs): Sensitive Environment Type:	
	0.6	Deti		
Are Beenle Occupying Residence or	9.50	bil Exposure Pati	thway - NA	or
Attending School or Daycare on or Within 200 Feet of Area of Known or Suspected Contamination:	Number of Wo	rkers Onsite*: 00 - 1,000 000	Within 200 Feet of Areas of Known or Suspected Contamination:	or
Tes Yes				
No			If Yes, List Each Terrestrial Sensitive Environment ⁵ :	
If Yes, Enter Total Residential Population: People ²	Population Within 1 Mile: People ⁷		*Refer to PA Table 7 for environment types	-
		10. Air Pathw	way	
Is there a Suspected Release to Air ¹ :		Wetlands Located	ed Within 4 Miles of the Site ⁶ :	
Enter Total Population on or Within:		☐ Yes ☐ No	If Yes, How Many Acres: Acres	
Onsite		Other Sensitive E	Environments Located Within 4 Miles of the Site:	
0-1/4 Mile			☐ Yes ☐ No	
>1/4-1/2 Mile		List All Sensitive I	Environments Within 1/2 Mile of the Site ⁶ :	
>1/2-1 Mile		Distance: Sen	ensitive Environment Type/Wetlands Area (acres):	
>1-2 Miles		Onsite		
>2-3 Miles		0-1/4 Mile		
>3-4 Miles		>1/4-1/2 Mile		
Total Within 4 Miles ³⁻⁵		*Refer to PA Table 10	L0 for calculations on air pathway exposures	

¹⁻¹¹ Refers to question number on the PA scoresheet for each particular pathway

					Identification	
Potential Haza	ardous Wast	e Site Pre	liminary Ass	essment Form	State:	CERCLIS #:
				CERCLIS Discovery Date:		
		1.	General Site Infor	mation		
Name: Northern Sewage	e Treatment Plant	Street Address	: NA			
City:		State: FL	Zip Code:32925	County:Brevard	Co. Code:12009	Cong. Dist:8th
Latitude:	Longitude:	Approximate A Acres	area of Site:	Status of Site:	ied	
		·	Square Ft	✓ Inactive □ NA (GW pl	lume, etc.)	
Site Name: Northern Se	wage Treatment Plar	nt				
served as the main STP from the housing area, s decommissioned. Begin	for Patrick AFB befor shops, barracks, mes ning in 1995 all wast	e being decomr s hall, and office ewater generate	nissioned during Fel e buildings. Effluent ed at Patrick AFB is j	from the STP was dischar pumped to the City of Coc	The STP treated dome rged to the Banana Ri coa for treatment and	estic wastewater ver prior to being d disposal.
		2. 0	wner/Operator In	ormation		
Owner: Patrick AFB			Operator: Same as	"owner"		
Street Address:			Street Address:			
City:			City:			
State: FL	Zip Code:32925	Telephone:	State:	Zip Code:	Telephone:	
Type of Ownership:			Type of Ownership	:		
 Private Federal Agency Name: <u>DOD</u> State Indian 	County County Municipal Not Specif Other	ied	 Private Federal Agency Name: State Indian 	County County Municipal Not Specified Other		
		3. 9	ite Evaluator Info	rmation		
Name of Evaluator: Johr	n Sandoval	Agency/Organ	ization: HydroGeoLogic, Inc.		Date Prepared:07/0	6/15
Street Address:404 E. Ra	amsey Road, Ste. 210)	City:San Antonio		State:Texas	
Name of EPA or State Agency Contact:NA			Street Address:			
City: State:		Telephone:				
		4. Site	Disposition (for El	PA use only)		
Emergency Response/Re	emoval Assessment		CERCLIS Recommendation:		Signature:	
	Yes		Lower Priority SI		Name (typed):	
Date:			RCRA Other: Date:		Position:	

	5. (General Site Chara	cteristics	
Predominant Land Use Within	1 Mile of Site (check all that	Site Setting:		Years of Operation:
apply):		_		
		🗌 Urban		Beginning Year <u>1968</u>
	Aining Other Federal		an	
Residential	OOD Facility:			Ending Year <u>1995</u>
Forest/Fields	DOE			
	Other			
Type of Site Operations (check	all that apply):	-		Waste Generated:
Manufacturing (must check subcat	egory)	Retail		Onsite
Lumber and Wood Products		Recycling		Offsite
Inorganic Chemicals		Junk/Salvage Yard		Onsite and Offsite
Plastic and/or Rubber Produce	cts	Municipal Landfill		
Paints, Varnishes		Other Landfill		Waste Deposition Authorized By:
Industrial Organic Chemicals				Present Owner
Agricultural Chemicals				Former Owner
Miscellaneous Chemical Prod	lucts	Other Federal Facilit	tv	Present & Former Owner
Metal Coating Plating Engra	aving		·	
Metal Forging Stamping	iving	Treatment, Stora	age, or Disposal	
Fabricated Structural Metal F	Products	Large Quantity C	Generator	Waste Accessible to the Public:
Electronic Equipment		Small Quantity G	Senerator	
Other Manufacturing		Subtitle D		□ Yes
		🗌 Municipal		✓ No
		Industrial		
		Converter"		Distance to Nearest Dwelling School
		"Protective Filer"		Distance to Nearest Dweining, School,
		□ "Non-or Late File	er"	or workplace:
		Note Specified		
		Other		<u>on site</u> Feet
6. Waste Characteristics In	formation			(Refer to PA Table 1
		for WC Score)		
Source Type:	Source Waste Quantity:	Tier*:	General Type of Waste	(check all that
(check all that apply)	(include unit)		apply):	
			Metals	Pesticides/Herbicides
Surface Impoundment			Organics	Acids/Bases
Drums				Oily Waste
Tanks and Non-Dum Containers			Solvents Paints /Pigments	
Chemical Waste Pile			Laboratory/Hospital Waste	
Scrap Metal or Junk Pile				
Tailings Pile			Construction/Demolition W	Vaste
Trash Pile (open drum)				
Land Treatment				
Contaminated GW Plume			Physical State of Waste	as Deposited (check all that apply):
(unidentified source)				
(unidentified source)			Solid	
Contaminated Soil			Sludge	
✓ OtherAFFF			Powder	
No Sources				
*C=Constituent, W=Wastes	stream, V=Volume, A=Area		L Gas	

	7. Ground Water Pa	athway		
Is Ground Water Used for Drinking Within 4	Is There a Suspected Release to	List Secondary Target Population Served by Ground Water		
Miles:	Ground Water ¹ :	Withdrawn From:		
لا الم	Ves			
	□ No			
	_	0 - 1/4 Mile		
If Yes, Distance to nearest Drinking				
Well:	Have Primary Target Drinking	>1/4 - 1/2 Mile		
_ <u>5.08</u> miles	Water Wells Been Identified:	>1/2 - 1 Mile		
Type of Drinking Water Wells Within 4		>1/2 - 1 Mile		
Miles	IJ Yes	>1 - 2 Mile		
(check all that apply):				
	L No	>2 - 3 Mile		
 ✓ Private 	If Yes, Enter Primary Target			
None None	Population:	>3 - 4 Mile		
	<u>25</u> People ³			
Depth to Shallowest Aquifer:	Nearest Designated Wellhead	Total Within 4 Miles ⁴ 15,877		
4.5 Foot	Protection Area ⁶			
_ <u></u>	Protection Area :			
Karst Terrain/Aquifer Present:	Underlies Site	*Use population #s for PA Table 2		
	□ >0-4 Miles	*Note nearest well for #5 on GW Pathway Scoresheet		
□ No				
	8. Surface Water Pa	athway		
Type of Surface Water Draining Site and 15 M	Viles Downstream (check all that	Shortest Overland Distance From Any Source to Surface Water:		
annly):	mes Downstream (check an that	Shortest overland Distance from Any Source to Surface Water.		
Stream River Por	nd 🗌 Lake	<u>55</u> Feet		
Bay Ocean Oth	ner _	Miles		
Is There a Suspected Release to Surface Wat	er ¹ :	Site is Located in:		
		Annual - 10 yr Floodplain		
✓ Yes		>10yr - 100yr Floodplain		
🗌 No		✓ >100yr - 500yr Floodplain		
		>500yr Floodplain		
Drinking Water Intake Located Along the Sur	face Water Migration Path:	List All Secondary Target Drinking Water Intakes:		
		<u>Name:</u> <u>Water Body</u> : <u>Flow (cfs)</u> : <u>Population Served</u> :		
Have Primary Target Drinking Water Intakes	Been Identified:			
Yes If Vec Distance	ce to Nearest Drinking			
✓ No Water Intake	· Miles ⁶			
Water make	· Miles			
If Yes, Enter Population Served by Target Inta	ake:			
People ⁴		Total within 15 Miles ⁴		
Fisheries Located Along the Surface Water M	ligration Path:	List All Secondary Target Eisberies ¹⁰		
	e to Nearest Fisherv	Water Body/ Eishen Name		
∐ Yes ⊻ No	Miles	water bouy/ Fishery Name : Flow (CIS):		
Have Primary Target Eisheries Been Identified:				

8. Surface Water Pathway (continued)				
Wetlands Located Along the Surface Water I	Other Sensitive Environments Located Along the Surface Water Migration Path:			
✓ Yes □ No		✓ Yes If Yes, Distance to Nearest Sensitive Environment: Nofeet		
Have Primary Target Wetlands Been Identif	ied:	Have Primary Tar	get Sensitive	e Environments Been Identified:
✓ Yes □ No			☐ Yes ✓ No	
List All Wetlands:		List All Sensitive E	Invironment	5 ¹¹ :
Water Body : Flow (cfs): Frontage miles:		<u>Water Body</u> :	Flow (cfs):	Sensitive Environment Type:
Banana River		multiple wetlands		
	9.	I . Soil Exposure Pa	thway	
Are People Occupying Residence or Attending School or Daycare on or Within 200 Feet of Area of Known or Suspected Contamination:		rkers Onsite ⁴ : 00 - 1,000 000	Have Terre Within 200 Contamina	estrial Sensitive Environments Been Identified on or) Feet of Areas of Known or Suspected ation:
☐ Yes ☑ No		hip 1 Milo:	If Yes, Lis	☑ No t Each Terrestrial Sensitive Environment ⁵ :
If Yes, Enter Total Residential Population: People ²	<u>15,877 (4-i</u> People ⁷	miles)	*Refer to PA	Table 7 for environment types
		10. Air Pathwa	ay	
Is there a Suspected Release to Air ¹ : Yes No Enter Total Population on or Within:		Wetlands Located	Within 4 Mi If Yes, Ho	les of the Site ⁶ : w Many Acres: <u>unknown</u> Acres
Onsite		Other Sensitive Environments Located Within 4 Miles of the Site:		
0-1/4 Mile		✓ Yes □ No		
>1/4-1/2 Mile		List All Sensitive Environments Within 1/2 Mile of the Site ⁶ :		
>1/2-1 Mile		Distance: Sens	itive Environr	nent Type/Wetlands Area (acres):
>1-2 Miles		Onsite		
>2-3 Miles		0-1/4 Mile <u>Bar</u>	nana River	
>3-4 Miles		>1/4-1/2 Mile		
Total Within 4 Miles ³⁻⁵ <u>15,877</u>		*Refer to PA Table 10 fo	or calculations or	n air pathway exposures

 $^{1\mathchar`-11}$ Refers to question number on the PA scoresheet for each particular pathway

					Identification	
Potential Haz	ardous Wast	e Site Pre	liminary Ass	essment Form	State:	CERCLIS #:
			-		CERCLIS Discovery Da	ate:
		1.	General Site Infor	mation		
Name: Outfall 21 to Bar	nana River	Street Address	: NA			
City:		State: FL	Zip Code:32925	County:Brevard	Co. Code:12009	Cong. Dist:8th
Latitude:	Longitude:	Approximate A Acres	rea of Site:	Status of Site:	ied	1
			Square Ft	Inactive NA (GW pl	lume, etc.)	
Site Name: Outfall 21 to	Banana River					
outfall is part of the dra AFFF from the fire supp the hangars. These store	inage system at Patri ression systems at th m sewer inlets releas	ck AFB and is po eses hangars flo e to the environ	otentially influenced owed out of the han ment at the drainag	by drainage from Hangar gars doors to the grassy a ge canal through Outfall 2	rs 630 and 647. Histor areas and storm sewer 1.	ical discharge of inlets north of
		2. Ov	wner/Operator Inf	ormation		
Owner: Patrick AFB			Operator: Same as	"owner"		
Street Address:			Street Address:			
City:			City:			
State: FL	Zip Code:32925	Telephone:	e: State: Zip Code:		Telephone:	
Type of Ownership:			Type of Ownership:			
□ Private □ County ☑ Federal Agency □ Municipal Name: □ Not Specified □ State □ Other □ Indian □ Other		Private County Federal Agency Municipal Name: Not Specified State Other Indian Indian				
		3. S	ite Evaluator Info	rmation		
Name of Evaluator: Joh	n Sandoval	Agency/Organi	ization: HydroGeoLogic, Inc.		Date Prepared:07/06/15	
Street Address:404 E. Ra	amsey Road, Ste. 210)	City:San Antonio		State:Texas	
Name of EPA or State Agency Contact:NA		Street Address:				
City: State:		Telephone:				
		4. Site I	Disposition (for EP	A use only)		
Emergency Response/R	emoval Assessment		CERCLIS Recomme	ndation:	Signature:	
	Yes		Lower Priority SI		Name (typed):	
Date:	No		RCRA Other: Date:		Position:	

	5. (General Site Chara	cteristics	
Predominant Land Use Wit	hin 1 Mile of Site (check all that	Site Setting:		Years of Operation:
apply):		_		
		🗌 Urban		Beginning Yearunknown
			an	
Residential	✓ DOD		an	Ending Yearpresent
Forest/Fields	DOE			
	Other			Unknown
Type of Site Operations (ch	eck all that apply):			Waste Generated:
Manufacturing (must check su	bcategory)	Retail		✓ Onsite
Lumber and Wood Produ	ucts	Recycling		Offsite
Inorganic Chemicals		Junk/Salvage Yard		Onsite and Offsite
Plastic and/or Rubber Pr	oducts	Municipal Landfill		
Paints, Varnishes		Other Landfill		Waste Deposition Authorized By:
Industrial Organic Chem	icals	✓ DOD		Present Owner
Agricultural Chemicals				Former Owner
Miscellaneous Chemical	Products	DOI	+.,	Present & Former Owner
Primary Metals			ty	
Metal Coating, Plating, E	ngraving	Treatment, Stora	age, or Disposal	Unknown
	tal Products	Large Quantity C	Generator	Waste Accessible to the Public:
		Small Quantity G	Generator	
Other Manufacturing		Subtitle D		
		🗌 Municipal		
		Industrial		
		Converter"		Distance to Necret Dwelling, Colored
		"Protective Filer"		Distance to Nearest Dweiling, School,
		🗌 "Non-or Late File	er"	or Workplace:
		Note Specified		
		Other		<u>80</u> Feet
6. Waste Characteristics	Information			(Refer to PA Table 1
		for WC Score)		
Source Type:	Source Waste Quantity:	lier*:	General Type of Waste	(check all that
(check all that apply)	(include unit)		apply):	
			Metals	Pesticides/Herbicides
Surface Impoundment			Organics	Acids/Bases
Drums				Oily Waste
Tanks and Non-Dum Container	rs		Solvents Depints / Pigmonts	
Chemical Waste Pile			Laboratory/Hospital Waste	
Scrap Metal or Junk Pile			Radioactive Waste	C Other AFFF
Tailings Pile			Construction/Demolition V	Vaste
Trash Pile (open drum)				
Land Treatment				
Contaminated GW Plume			Physical State of Waste	as Deposited (check all that apply):
(unidentified source)				
(unidentified source)			Solid	
Contaminated Soil			Sludge	
OtherAFFF				
No Sources				
*C=Constituent, W=Wa	astestream, V=Volume, A=Area			

	7. Ground Water Pa	athway
Is Ground Water Used for Drinking Within 4	Is There a Suspected Release to	List Secondary Target Population Served by Ground Water
Miles:	Ground Water ¹ :	Withdrawn From:
لع Yes	لا Yes	
□ No	□ No	
		0 - 1/4 Mile
If Yes, Distance to nearest Drinking		>1/4 - 1/2 Mile
Well:	Have Primary Target Drinking	
	Water Wells Been Identified:	>1/2 - 1 Mile
Type of Drinking Water Wells Within 4		
Miles	✓ Yes	>1 - 2 Mile
(check all that apply):	□ No	
Municipal		>2 - 3 Mile
✓ Private	If Yes, Enter Primary Target	
∟ None	Population:	>3 - 4 Mile
	People ³	Total Within 4 Milos ⁴ 15 414
Depth to Shallowest Aquifer:	Nearest Designated Wellhead	10tal Within 4 Miles 15,414
_ <u>4-5</u> Feet	Protection Area ⁶ :	
Karst Terrain/Aquifer Present:	Underlies Site	*Use population #s for PA Table 2
	>0-4 Miles	*Note nearest well for #5 on GW Pathway Scoresheet
✓ Yes	None Within 4 Miles	
No		
	8. Surface Water Pa	athway
Type of Surface Water Draining Site and 15 N	Ailes Downstream (check all that	Shortest Overland Distance From Any Source to Surface Water:
apply):		
Stream River Por	nd 🗌 Lake	<u>on-site</u> Feet
🗌 Bay 🗌 Ocean 🗔 Oth	ner _drainage canal	Miles
	1	Site is Legated in:
Is There a Suspected Release to Surface Wat	er*:	
لا Yes		$\square \text{ Annual - 10 yr Floodplain}$
		\checkmark >100yr - 100yr Hoodplain
		>500yr Floodplain
Drinking Water Intake Located Along the Sur	face Water Migration Path	List All Secondary Target Drinking Water Intakes
Drinking water intake Located Along the Su		List All Secondary Target Drinking Water Intakes.
Yes		
✓ No		Name: Water Body: Flow (cfs): Population Served:
Have Primary Target Drinking Water Intakes	Been Identified:	
nuver minury runger brinking water makes	been identified.	
☐ Yes If Yes, Distant	ce to Nearest Drinking	
₩ Water Intake	: Miles ⁶	
If Yes, Enter Population Served by Target Inta	ake:	
People ⁴		Total within 15 Miles ⁴
Fisheries Located Along the Surface Water N	ligration Path:	List All Secondary Target Fisheries ¹⁰ :
If Yes, Distance	e to Nearest Fishery:	Water Body/ Fishery Name : Flow (cfs):
	Miles	
Have Primary Target Fisheries Been Identifie	d:	1

8. Surface Water Pathway (continued)				
Wetlands Located Along the Surface Water N	Other Sensitive Environments Located Along the Surface Water Migration Path:			
✓ Yes □ No		✓ Yes If Yes, Distance to Nearest Sensitive Environment: Nofeet		
Have Primary Target Wetlands Been Identif	ied:	Have Primary Tar	get Sensitive	e Environments Been Identified:
✓ Yes □ No			☐ Yes ✓ No	
List All Wetlands:		List All Sensitive E	invironment	s ¹¹ :
Water Body : Flow (cfs): Frontage miles:		<u>Water Body</u> :	Flow (cfs):	Sensitive Environment Type:
Banana River				
	9.	. Soil Exposure Pa	thway	
Are People Occupying Residence or Attending School or Daycare on or Within 200 Feet of Area of Known or Suspected Contamination:		rkers Onsite ⁴ :	Have Terre Within 200 Contamina	estrial Sensitive Environments Been Identified on or O Feet of Areas of Known or Suspected ation:
☐ Yes ✓ No		hip 1 Mile:	If Yes, Lis	☑ No t Each Terrestrial Sensitive Environment ⁵ :
If Yes, Enter Total Residential Population: People ²	<u>15,414 (4-1</u> People ⁷	miles)	*Refer to PA	Table 7 for environment types
		10. Air Pathwa	ay	
Is there a Suspected Release to Air ¹ : Yes No Enter Total Population on or Within:		Wetlands Located	Within 4 Mi If Yes, Ho	iles of the Site ⁶ : w Many Acres: <u>unknown</u> Acres
Onsite		Other Sensitive En	vironments	Located Within 4 Miles of the Site:
0-1/4 Mile		✓ Yes □ No		
>1/4-1/2 Mile		List All Sensitive Environments Within 1/2 Mile of the Site ⁶ :		
>1/2-1 Mile		Distance: Sensi	tive Environr	ment Type/Wetlands Area (acres):
>1-2 Miles		Onsite		
>2-3 Miles		0-1/4 Mile <u>Bar</u>	nana River	
>3-4 Miles		>1/4-1/2 Mile		
Total Within 4 Miles ³⁻⁵ <u>15,414</u>		*Refer to PA Table 10 fo	or calculations of	n air pathway exposures

 $^{1\mathchar`-11}$ Refers to question number on the PA scoresheet for each particular pathway

APPENDIX C

RECORDS OF COMMUNICATION

This page was intentionally left blank.



COMMUNICATION RECORD

y, Florida Phone: 210-348-8778 Email: <u>jsandoval@hgl.com</u>
Phone: 210-348-8778 Email: jsandoval@hgl.com
Phone: 210-348-8778 Email: jsandoval@hgl.com
Email: jsandoval@hgl.com
Phone:
Email:
sitions?
om a Hangar and discharged to the floor drains. /hat hangar the release occurred at. y of Cocoa Beach Dyal Water Treatment Plant, where it



•

Constructed in 1953.

COMMUNICATION RECORD

Name of Base, State: Patrick AFB, Brevard County, Florida				
Interviewer: John Sandoval				
Organization: HGL Phone: 210-348-8778				
Position/role on this project: Field Team Lead Email: <u>jsandoval@hgl.com</u>				
Interviewee: Dallas More				
Organization: Patrick AFB Fire Department Phone:				
Position/Job Title: Assistant Fire Chief Email:				
How Long in this Position? March 1998				
How long at this Base in current and previous positions?				
Have you held similar positions at other bases?				
Which bases?				
How long?				
Discussion:				
Building 984:				
AFFE Storage				
Hangar 750				
 Constructed in 1943; Unable to access during PA visit. 				
 Initially equipped with AFFF fire suppression system and retrofitted to HEF in 2006. 				
 1999 AFFF release at hangar. System tripped and 3 feet of foam filled the hangar. Unknown 				
amount of AFFF released.				
• 3% AFFF				
Time and Distance Testing and Flushing out hoses:				
Known as Ops Check				
Conducted at 800 ramp and fire station				
• Discharged at South, west, and north areas of fire station.				
 1999/2000 Ops Check was no longer conducted with AFFF 				
Water is now used for OPS checks at Taxiway Juliet				
Fire Station				
Fire Engine 9 Crash Truck 6				
 Holds 55-gallons of AFFF 3% 210 gallons of AFFF 3% 				
Fire Engine 4 Crash Truck 5				
 Holds 55-gallons of AFFF 3% 500 gallons of AFFF 3% 				
• 2,000 gallon AFFF Trailer				

Discussion:

- No containment system
- Equipped with an AFFF system (3% AFFF) including four low level turrets in 2003.
- Has an 800 gallon AFFF AST
- No reported or documented releases of AFFF.

Hangar 986

- Constructed in 1953
- Equipped with a Deluge System.
- Hangar has never been equipped with an AFFF fire suppression system.
- No reported or documented releases of AFFF.

Building 705

- Constructed in 1999.
- formerly supplied Hangar 750
- Supplies AFFF to Hangar 751
- Contains 1,200 gallon AFFF poly tank
- AFFF pipelines that supplied Hangar 750 were capped off in 2006.

Former Fire Training Area 2.

- Was not aware of former burn pit
- FTA no longer used; FTA activities ceased around 2000/2001.

Hangar 630

- Constructed in 1964.
- Equipped with an AFFF fire suppression system in 1999.
- In 2003, the AFFF fire suppression system was upgraded.
- 800 gallon AFFF (3%) AST in MEC room
- AFFF suppression system tripped and hangar was filled with 3 feet of AFFF.
- Timeframe and amount of AFFF released unknown.

Hangar 647

- Constructed in 1970.
- •
- Equipped with an AFFF fire suppression system in 1999.
 - contains 2,000 gallon AFFF AST.

Fire Truck Rollover

- Rollover occurred in 1997.
- AFFF spilled out of the fire truck onto Taxiway Echo and Taxiway Bravo.
- Unknown amount of AFFF released.

Excess AFFF is stored at Building 984. Approximately 25 55-gallon drums stored at Building 984. Hangar systems are tested annually and the foam is sampled.

Every two years the systems are dumped. Only water is dumped not AFFF.

Discussion: Hangar 751 Currently equipped with an AFFF fire suppression system. • 1,200 gallon AFFF poly AST supplies hangar with AFFF. • .

- 1,200 gallon AST located in Building 705
- AFFF system installed in 1999.

V	HGL HydroGeoLogic, Inc

COMMUNICATION RECORD

Name of Base, State: Patrick AFB, Brevard County, Florida				
Interviewer: John Sandoval				
Organization: HGL	Phone: 210-348-8778			
Position/role on this project: Field Team Lead	Email: jsandoval@hgl.com			
Interviewee: Darren Schubert, Charlie Roy, George	2 Robbins			
Organization:	Phone:			
Position/Job Title:	Email:			
How Long in this Position?				
How long at this Base in current and previous posi	tions?			
Have you held similar positions at other bases?				
Which bases?				
How long?				
now long:				
Discussion:				
Hangar 750:				
 Initially equipped with wet system prior to 2000 				
 Retrofitted to 3% AFFF in 2000 and operated 	d until 2006.			
 In 2006, fire suppression system was retrofit 	ted to HEF.			
• Fire suppression system is currently charged with HEF.				
Building 705				
Building 705 historical	y pumped AFFF to Hangars 750 and 751 via			
underground piping.				
 AFFF was accidently released in 2012 at Building 705 to the floor drains. Approximately 5 to 10 gallons of AFFF was released. 				
• There was a leak in the underground piping to Hangar 751 in 2008/2009. Approximately 800 gallons of AFEE was released to the subsurface				
 Pipelines from Building 705 to Hangar 750 w 	vere capped off in 2006.			
 August 2007 there was a leak in the underground nining to Hangar 751. Unknown amount of AF 				
released. Leak occurred where piping was al	pout to go into Hangar 751.			
Hangar 751:				
 Equipped with an AFFE fire suppression system 	em.			
 1 200 gallon AFEE poly tank is located in pun 	nn house			
 Building 705 still pumps AFFF to Hangar 751 				
Every two years Hangars 750 and 751 fire suppression	on systems were dumped to the hangar floor drains			

Every two years Hangars 750 and 751 fire suppression systems were dumped to the hangar floor drains and discharged to a 30,000 UST located south of Hangar 750. Contents of the 30,000 UST have been pumped out to the ground surface surrounding the UST. Alternately, the contents are pumped out by a subcontractor using a Vacuum truck and disposed off-base.

Discussion:

Hangar 630:

• 2004 there was a release of AFFF at the hangar due to a natural disaster (Hurricane) that tripped the system. Unknown amount of AFFF released out of the hangar doors.

Hangar 647:

• Accidental AFFF Release in 2009. Approximately 440 gallons released to the hangar floor and out to the parking apron in front of the Hangar.

Underground containment tank installed in June/July 2014 to collect discharge from Hangar 630 and 647. Underground tank is pumped out by subcontractors and disposed off-base.

Hangar 985 and 986

• No reported or documented releases of AFFF.

Building 676

- A Facility that was initially equipped with a wet pipe fire suppression system. In January 2012, the system was retrofitted with an AFFF fire suppression system.
- No reported or documented release of AFFF.



COMMUNICATION RECORD

Name of Base, State: Patrick AFB, Brevard County, Florida	
Interviewer: John Sandoval	
Organization: HGL	Phone: 210-348-8778
Position/role on this project: Field Team Lead	Email: jsandoval@hgl.com
Interviewee: John Langett	
Organization: AFCEC	Phone:
Position/Job Title: Patrick AFB RPM	Email:
How Long in this Position?	
How long at this Base in current and previous positions?	
Have you held similar positions at other bases?	
Which bases?	
How long?	
Discussion:	
Reviewed sites from previous investigations.	
• Mr. Langett confirmed that all information provided in the previous PFC Investigation is correct.	
Groundwater is not used as drinking water.	
Drinking water for Patrick AFB is purchased from the City of Cocoa.	
 Indicated that there was a FTA PFC Investigation conducted in 2011. 	

- There are no current Fire Training Areas.
- AFFF releases from hangars drain to the hangar floor drains and releases to the Y-Drainage Ditch. The drainage ditch releases to Banana River.

This page was intentionally left blank.